Rowan University Rowan Digital Works

Theses and Dissertations

4-2-2020

Qualitative study investigating Western student affairs practices in Qatar

Amjad Omar Abdo Rowan University

Follow this and additional works at: https://rdw.rowan.edu/etd

Part of the Higher Education Commons

Recommended Citation

Abdo, Amjad Omar, "Qualitative study investigating Western student affairs practices in Qatar" (2020). *Theses and Dissertations*. 2768. https://rdw.rowan.edu/etd/2768

This Dissertation is brought to you for free and open access by Rowan Digital Works. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of Rowan Digital Works. For more information, please contact graduateresearch@rowan.edu.

QUALITATIVE STUDY INVESTIGATING WESTERN STUDENT AFFAIRS PRACTICES IN QATAR

by

Amjad O. Abdo

A Dissertation

Submitted to the Department of Educational Services and Leadership College of Education In partial fulfillment of the requirement For the degree of Doctor of Education at Rowan University March 23, 2020

Dissertation Chair: Dr. MaryBeth Walpole, Ph.D.

Dedication

To Qatar Foundation for establishing such unique global higher education opportunities of which I have been privileged to be part for the past 8 years.

To all of the student affairs practitioners, faculty members, and administrators who choose to travel across the world to support the advancement of higher education and contribute to educating students from across the globe.

Acknowledgments

I would like to acknowledge and thank everyone who guided and supported me throughout the dissertation process, beginning with my committee, Dr. MaryBeth Walpole, Dr. Katherine Harris, and Dr. Dianna Dale. I could not have made it without their guidance, commitment, and support.

To Dr. MaryBeth Walpole, my dissertation committee chair: Thank you for challenging and inspiring me to produce this work. I appreciate your patience and understanding with the challenges that I faced during my dissertation journey.

To Dr. Katherine Harris, committee member and mentor: Thank you for your unending faith in my research and in me. Thank you for encouraging me to explore this research topic after taking my first class in the program.

To Dr. Mohamud Verjee, Assistant Dean for Medical Student Affairs at Weill Cornell Medicine-Qatar: Thank you for serving as my local PI and assisting me in securing the local IRB approval in Qatar.

Finally, and most important, I wish to acknowledge my wife, Kholoud; my children, Sara and Yousef; my parents; my extended family; and my friends for their continuing support and understanding of my absence during my work on this research and dissertation. I could not have done it with your support and believing in me. Thank you!

Abstract

Amjad O. Abdo QUALITATIVE STUDY INVESTIGATING WESTERN STUDENT AFFAIRS PRACTICES IN QATAR 2019-202 MaryBeth Walpole, Ph.D. Doctor of Education

The internationalization movement in higher education expanded quickly in the past two decades. While efforts varied among institutions, the common and preferred approach has been creation of International Branch Campuses (IBC). Currently, there are 283 IBCs, 30% of which are located in the Middle East region.

This study explored the role of local culture context in shaping Western student affairs practices implemented at the American IBCs in the State of Qatar. A qualitative single-case study included interviews with 14 Western expatriates who were recited from the United States to work in the Student Affairs departments in Education City, Qatar. The intercultural competence framework was utilized to guide the study.

Findings revealed the influence of the local cultural context in shaping student affairs practitioners' programs and events. Participants noted that they work in a collective society, with strong families ties and influence, with significant differences between Eastern and Western cultures. They described modifications to their events and programs to make them more relevant and to meet the needs of their local students and made recommendations for transition of new student affairs staff in the Middle East.

Research findings suggest the significant role of the local Arab culture in shaping student affairs programs and initiatives differently from those in the United States.

V

Table of (Contents
------------	----------

Abstract
Chapter 1: Introduction1
The Rising Global Demand for Higher Education
The Internationalization Movement
International Branch Campuses
IBCs in the State of Qatar
Challenges Facing IBCs5
Student Affairs Practitioners
Problem Statement
Purpose of the Study
Research Questions
Theoretical Framework
Limitations and Delimitations of the Study12
Significance of the Research
Definition of Key Terms14
Organization of the Dissertation15
Chapter 2: Review of Literature17
Internationalization of Higher Education17
Internationalization Versus Globalization18
History of the Internationalization Movement
Benefits of Internationalization
Middle Eastern Trends in Higher Education24

Development of IBCs
Education City in Qatar27
Student Affairs Services
Student Affairs Practitioners
Student Affairs in the Middle East
Western Student Affairs Practice Models
The Extracurricular Model
The Academic-Student Affairs Collaboration Model
Student Affairs Models Outside the United States
Developments in Student Affairs and Services
Developments in Europe
Developments in the Middle East
Future Global Challenges for Student Affairs
The Concept and Role of Culture
Intercultural Communication
The Arab World
The Arab Culture
Eastern Versus Western Cultures
Communication Challenges
Cultural Challenges for IBCs
Cultural Challenges in Education City
Theoretical Framework

Chapter 3: Research Design and Methodology	62
The Research Site	62
The Research Design	63
The Constructivist Worldview	64
Positionality Statement	66
Qualitative Methodology	68
Qualitative Case Study Approach	71
Applying the Case Study Approach	75
Research Questions	76
Research Participants	77
Data Collection Procedures	79
Data Analysis	80
Trustworthiness and Validity Threats	81
Ethical Considerations	82
Chapter 4: Results and Findings	83
Research Participants	83
Background of the Participants	84
Lucas	84
William	85
Sophia	86
Olivia	86
James	

L	ogan	87
Ja	ack	88
Ľ	David	88
C	Dliver	89
A	Amelia	90
S	amantha	90
K	Kathy	91
L	ina	91
E	Emma	92
S	Summary	92
Then	ne 1: The Role of the Local Culture	92
C	Conservative Cultural Context	93
R	Respecting the Local Culture and Host Country	95
Iı	nvolvement and Engagement Outside the Classroom	96
Then	ne 2: Student Affairs Professional Practices	98
R	Replicating Western Practices Is Not Effective	98
Ľ	Differences Between Eastern and Western Cultures	100
C	Gender Interactions and Mixing	103
Then	ne 3: Best Practices for IBCs in the Middle East	104
A	Adopted Successful Student Affairs Practices	104
В	Best Practices for New Incoming Staff	108
S	Summary of the Results	110

Chapter 5: Discussion	12
Results for Research Questions1	12
Research Question 1: The Role of the Local Culture1	12
Research Question 2: Student Affairs Professional Practices1	15
Research Question 3: Best Practices for IBCs in the Middle East1	17
Discussion and Analysis1	19
The Role of the Local Culture1	19
Student Affairs Practitioners1	22
Student Affairs Practices at IBCs1	23
Deardorff Intercultural Competence Theoretical Framework1	23
Implications for Practice1	25
Implications for Research1	26
Implications for Leadership and Leadership Theories1	28
Implications for Leadership1	28
Implications for Leadership Theories1	29
Conclusion1	30
References1	32
Appendix1	49

Chapter 1

Introduction

Higher education has been increasingly engaged in efforts to internationalize (Altbach, Resiberg, & Rumbley, 2009). In the United States, colleges and universities are forging global connections in order to introduce educational programs in many forms. These collaborations are created with the following goals: diversifying the study-abroad destinations, increasing international student enrollment, developing joint degree programs with international universities, and establishing International Branch Campuses (IBCs) around the world (Altbach & Knight, 2007; Lane, 2011). These initiatives generate revenue, provide prestige, and enhance academic programs and research opportunities (Knight, 2012a). As these efforts grow rapidly in the United States, higher education institutions face new challenges related to globalization and efforts to replicate their educational services around the globe (Altbach, 2011; Calderon, 2015).

The Rising Global Demand for Higher Education

The global demand for access to higher education has been spurred by the rapid growth of the student population worldwide, which has doubled in size over the span of 20 years (van der Wende, 2003; World Bank, 2000). As of 2015, there were more than 150 million students, with projections of 660 million by 2040 (Calderon, 2015). This student population size represented 10% of the overall worldwide population in 2015, compared to 4% in 2012 (Calderon, 2015). These trends provide a clear indication that the current student population will continue to grow, leading to an increase in demand for access to higher education and a subsequent increase in student mobility worldwide (Calderon, 2015; Connelly & Olsen, 2013; United Nations Educational, Scientific and Cultural Organization [UNESCO], 2009; UNESCO Institute for Statistics, 2013).

In the United States, the Institute of International Education (IIE) reported a 51% increase in international student enrollment from 2002 to 2014 (IIE, 2014). These data reflect the increase in student mobility and new global trends that higher education institutions will face in meeting high demands globally (Connelly & Olsen, 2013; IIE, 2014).

The Internationalization Movement

Internationalization plays a significant role in shaping the experiences of students in higher education institutions throughout the world (Knight, 2004, 2006, 2008; Altbach, 2011). *Internationalization* can be defined as the process of creating a global focus and culture in higher education institutions through establishing and delivering international initiatives and programs (Knight, 2003, 2014; Soilemetzidis, 2011). Internationalization efforts are uniquely structured to meet the interest and needs of each institution (Knight, 2014).

Internationalization can be classified in two ways: (a) internationalization inside the home country, and (b) internationalization abroad (Knight, 2004). Internationalization at home consists of strategies adopted by institutions to enhance international presence on home campuses. This may include recruiting international students, faculty, and researchers. Internationalization abroad requires institutions to venture into the world through study-abroad opportunities, IBCs, long-distance education programs, and international partnerships (Altbach, 2011; Knight, 2004; UNESCO, 2009). The establishment of IBCs is one of the most common and preferred methods for internationalization (Cross-Border Education Research Team [C-BERT], 2017: Lane, 2011).

International Branch Campuses

The IBC is an educational facility established by a home institution in a foreign country (Altbach, 2004; Wilkins & Huisman, 2012). IBCs share structures with their home campuses by engaging in face-to-face teaching and offering the same basic services to students, faculty, and staff (Altbach, 2004; Marijk, 2003; Naidoo, 2008; Wilkins & Huisman, 2012). IBCs generally provide full academic programs that lead to degrees recognized by the home institution (Farrugia & Lane, 2013; Lane, 2011). IBCs can be fully subsidized by the host country or established as a joint venture with a local partner (Naidoo, 2008).

Since 2006, the number of IBCs around the world has increased by 43% (Becker, 2009). At the time of this writing, there were 284 IBCs, of which more than 85% were based in the United States (C-BERT, 2017). American colleges and universities are still widely considered to be best in providing standards of excellence in higher education (American Council on Education [ACE], 2011; Green, Kinser, & Eckel, 2008). As a result, the United States continues to take the lead in establishing IBCs, followed by Australia, United Kingdom, France, and India (Becker, 2009; C-BERT, 2017). American IBCs can play a significant role in meeting rising demands in the global market, especially in developing countries (Becker, 2009; C-BERT, 2017).

Several countries in the Middle East have started to offer incentives to attract IBCs to invest in building a knowledge-based economy (Romani, 2009). Some of these countries have recruited many IBCs and thereby serve as international higher education hubs for their regions. The United Arab Emirates (UAE) serve as the main educational hub in the Middle East region; they have established Dubai International Academic City, which serves as the home of 15 IBCs (Becker, 2009; Romani, 2009). Similar to the UAE, the State of Qatar established Education City to serve as the home of six American IBCs (Vora, 2014).

Qatar's ability to serve as an educational hub was acknowledged in data generated by the UNESCO Institute of Statistics (Clark, 2013; UNESCO Institute of Statistics, 2013). For example, the State of Qatar increased its international student enrollment by 60% from 2008 to 2010, from 3,393 students to 5,387 students. These international students have contributed significantly to development of human capital in the region (Clark, 2013; UNESCO Institute of Statistics, 2013).

IBCs in the State of Qatar

The State of Qatar is a nation of 2.2 million people, located in the Arabian Peninsula; it shares a border with Saudi Arabia (Anderson, 2015). Qatar has grown from one of the poorest countries in the area to one of the wealthiest, achieving the highestlevel per capita income in the world (Clark, 2013). The State of Qatar, similar to many carbon-based countries, is proactively seeking to change its dependency on oil and gas to a knowledge-based economy requiring a skilled and educated labor force (Clark, 2013; Qatar Foundation for Education, Science, and Community Development, herein Qatar Foundation [QF], 2016).

Qatar Foundation (QF) is a private nonprofit organization that aims to serve the Qatari people by promoting development in three key areas: education, science and research, and community development (QF, 2016). The Foundation "aspires to develop sustainable human capacity, social, and economic prosperity for a knowledge-based economy" (QF, 2016, para. 1). Education City, a QF educational initiative, was officially established in 1995 and is the home of six American IBCs in Doha, the capital city of the State of Qatar (Clark, 2013; Vora, 2014). These six American branch campuses (Texas A&M, Northwestern, Georgetown, Carnegie Mellon, Cornell, and Virginia Commonwealth) are considered some of the world's best institutions and are the main higher education providers in Qatar (Vora, 2014).

The American IBCs in Education City contribute to meeting the high demand for education in Qatar by accommodating more than 2,000 local and international students (Anderson, 2015). These IBCs provide opportunities for students who seek an American degree without the need to leave their home country or region (Anderson, 2015; Lawton & Katsomitros, 2012). This incentive is important to many students because the majority of IBCs grant degrees identical to those of their home institutions, with no branch campus distinction (Hughes, 2011). Support for this arrangement is strong among students. Ahmed Al-Qahtani, a Weill Cornell Medical student, stated, "The idea of having an Ivy League education in the comfort of your own home is great. . . . All I can say is, thank you, your highness, for everything" (Anderson, 2015, p. 1). In addition to the six American IBCs in Education City, Qatar hosts five IBCs from Canada, the United Kingdom, The Netherlands, and France in locations outside of Education City (C-BERT, 2017).

Challenges Facing IBCs

There are many concerns on the part of faculty, staff, and administrators in terms of maintaining reputations and institutional brands at these IBCs (Lawton & Katsomitros, 2012). For example, the president of Yale stated, "You can't replicate a 300-year-old university in a remote location" (Lawton & Katsomitros, 2012, p. 27). In addition, Altbach (2001) warned that American higher education values, such as institutional independence and academic freedom, are not shared around the world and may be restricted by host government laws and policies.

On the financial side, many consider the IBC initiative to be a risky project (Chalmers, 2011; Croom, 2012). Multiple accounts of struggling IBCs have emerged, beginning with American IBC closures in Japan in 1991 and including recent closures such as those by the University of New South Wales in Singapore and George Mason University in the UAE (Becker, 2009; Lewin, 2009; Ruby, 2010; Staley, 2013). Many of these campuses fail to meet enrollment targets or to cover their own operating expenses, sometimes related to economic recessions (Becker, 2009; Diconsiglio, 2009). The IBCs in the State of Qatar receive generous financial support from QF to maintain operations in Qatar (Anderson, 2016). Everette Dennis, Dean of Northwestern University in Qatar, stated, "It was a good deal . . . with no financial risks. The only risk was reputational" (Anderson, 2015, p. 1).

IBCs can face major challenges when operating in host counties that are very different in terms of cultural values and beliefs (Diconsiglio, 2009; Lane, 2011; Vora, 2014). To be successful in their internationalization efforts, IBCs must reach a deep level of cultural understanding of and respect for their host country's culture, values, and traditions as they carry out daily operations (Altbach et al., 2009; Diconsiglio, 2009; Knight, 2004; Shams & Huisman, 2012; Stryker, Witt, & Konecny, 2016).

6

Learning about other cultures should focus on a genuine understanding of these cultural values, norms, and traditions (Magolda, 2003; Meyer & Rhoades, 2006). It should focus not on learning the differences but on promoting understanding and acceptance of those cultures (Magolda, 2003; Meyer & Rhoades, 2006). Diversity and inclusion issues challenge student affairs professionals who serve an increasingly diverse student population (College Student Educators International [ACPA], 2016; Student Affairs Administrators in Higher Education [NASPA], 2016a, 2016b, 2016c). Student affairs professionals can significantly influence the promotion of diversity, equity, inclusion, and social justice through direct work with students (ACPA, 2016; NASPA, 2016a, 2016b, 2016c).

Student Affairs Practitioners

Student affairs practitioners play a significant role in supporting the mission and vision of their higher education institutions through services that they provide to college students (Ciobanu, 2013; Manning, Kinzie, & Schuh, 2014; UNESCO, 2002). The student services concept originated in ancient Athena's educational system and the modern model continues to be developed by student affairs educators in the United States (Ciobanu, 2013). However, because this concept is still a new phenomenon outside the United States, many non-U.S. institutions use the term *student services* instead of *student affairs* (Ciobanu, 2013; UNESCO, 2002, 2009). Student affairs or student services and support to students in higher education to ensure their growth and development as they complete academic coursework in college (Ciobanu, 2013; NASPA, 2016a, 2016b, 2016c). Student affairs professionals play an important role in enhancing the overall

quality of the educational mission through services and support systems that they establish for students (Ciobanu, 2013).

Problem Statement

As the number of IBCs continues to increase globally, demands for student affairs professionals who can efficiently support and sustain operations and services of IBCs around the world will increase. The main challenge for the student affairs professional is to function effectively with students from cultural backgrounds with which the professional may not be familiar (Otten, 2003; UNESCO, 2002). Understanding the local cultural context would help the student affairs professional to provide adequate services to students (NASPA, 2016c; UNESCO, 2002). Deficiencies in cultural understanding can impair the ability of student affairs staff to provide services and programs that meet student needs (NASPA, 2016c; UNESCO, 2002).

IBCs, as an international venture, require leaders who can navigate and operate in multiple-culture environments (Lane, 2011). Internationalization efforts such as IBCs do not lead directly to developing cross-cultural learning and understanding (Deardorff, 2016a; Otten, 2003). Therefore, student affairs professionals must take proactive measures to understand their students and the local cultural context to advocate for student needs (Ciobanu, 2013; Deardorff, 2016a; UNESCO, 2002).

The World Declaration on Higher Education adopted in the *UNESCO World Conference* highlighted the importance of developing student services globally (Blake, 2007; Ciobanu, 2013; UNESCO, 2002). In this declaration, an important call was made for higher education institutions worldwide to provide appropriate student services and programs that would enhance the quality of student life, meet the needs of students, and support their learning, success, and achievement (Ciobanu, 2013; UNESCO, 2002). Student affairs professionals who oversee student life can play a significant role in accomplishing this task (Blake, 2007; Ciobanu, 2013; Evans, Forney, & Guido-DiBrito, 1998; Magolda, 2003; Manning et al., 2014; Schuh, Jones, Harper, & Associates, 2011; UNESCO, 2002).

Purpose of the Study

The State of Qatar, specifically Education City, was selected as the site for this study. Education City is the home of six prestigious American institutions that were selected because of their academic excellence in particular fields. Virginia Commonwealth University was the first to join, in 1998, to offer programs in fine arts, fashion design, graphic design, interior design, and painting and printmaking (QF, 2017). Next, Weill Cornell Medical Colleges joined in 2002 to offer premedical and medical studies. Georgetown and Northwestern University were invited by QF to offer academic programs in foreign service and media and opened in 2005 and 2008, respectively (Anderson, 2015; Clark, 2013). Texas A&M University was invited in 2003 to offer specific engineering programs to enhance Qatar's national resources. Carnegie Mellon University joined in 2004 to offer programs in business administration, computer science, and information systems (Carnegie Mellon University in Qatar, 2015).

The American IBCs operating in Education City share common interests and have similar agreements with QF, the official agency hosting these IBCs in the State of Qatar (Clark, 2013; Vora, 2014). The American IBCs have been operating in the State of Qatar for more than 10 years. However, it is not clear how the local cultural context influences the operations of these Western IBCs, especially in services delivered by student affairs practitioners.

The purpose of this qualitative case study was to explore how the local cultural context shaped Western student affairs practices in IBCs from the perspectives of student affairs practitioners. The case study methodology was selected due to the nature of the inquiry and the special close-knit community established in Education City in Doha. A case study approach is typically selected when the phenomenon under research is related to an individual unit or a community whose members share strong ties and common bonds (Flyvbjerg, 2011; Merriam, 1998; Stake, 2008). A case study provides tools for indepth analysis, with multiple methods to collect and triangulate data (Flyvbjerg, 2011; Stake, 2008). Case study methodology connects to environmental factors and provides the researcher with the flexibility to establish boundaries to create the context and the case for the study (Flyvbjerg, 2011).

This study also examined whether and how student affairs staff members adjusted their practices to enhance student engagement and to support student needs. Multiple data sources were utilized, including in-depth semistructured interviews and a review of artifacts provided by the participants (Creswell, 2013; Rossman & Rallis, 2012; Stringer, 2014).

Research Questions

Three research questions guided this study:

- 1. How do student activities professionals describe the role of local culture in shaping Western student affairs services implemented on their branch campuses?
- 2. In what ways do student activities professionals alter their Western practices at IBCs in Qatar?

3. What best practices can emerge from the study findings to guide student activities professionals working at IBCs in the Middle East Region?

Theoretical Framework

The intercultural competence framework was developed by Deardorff (2006) by inviting 23 intercultural scholars to participate in her study. The participants in Deardorff's study represented a wide range of experts from "communication, political science, education, international relations, anthropology, political science, psychology, and business" (Deardorff, 2006, p. 244). These scholars contributed significantly to the success of Deardorff's framework and made it a leading and frequently used intercultural model in higher education (Arasaratnam-Smith, 2017; Deardorff, 2012, 2016a, 2016b; Moscaritolo, Osfield, Perozzi, & Shea, 2016).

The model presented by Deardorff (2012) consists of five stages: attitudes, knowledge, skills, internal outcomes, and external outcomes. Participants move through these stages by developing a sense of curiosity and interest in exploring differences by challenging their personal beliefs. In the advanced stages, participants develop intercultural skills to control their behaviors and reactions in diverse intercultural situations. In a sense, they are in control of their behaviors and adjust them as they experience diverse intercultural environments (Deardorff, 2012).

Deardorff's model was selected because it speaks to student affairs professionals' work, especially in global settings (Arasaratnam-Smith, 2017; Deardorff, 2012, 2016a, 2016b). According to Deardorff, student affairs professionals offer many programs and services without adopting an appropriate intercultural framework. As a result, student learning and development may not occur efficiently, especially in diverse intercultural

environments. Student affairs practitioners should consider implementing such a model to enhance student learning and development (Bakken, 2013; Deardorff, 2012, 2016a, 2016b; Jauregui, 2013).

Limitations and Delimitations of the Study

This study was limited by several factors. First, there was a lack of literature or research that directly explained the role of the cultural context in shaping the student affairs practices at IBCs worldwide. If such impact existed, there was no research regarding how student affairs practitioners adjusted their practices to be relevant to their local context. Furthermore, there was no available research on the effects of cross-cultural training in preparing student affairs professionals for global assignments.

This study was delimited to American IBCs in Education City, Qatar. The data were mainly collected via interviews with student affairs staff members who worked in the American IBCs in Qatar. Also, four interviews were conducted in the United States with former employees who had worked in Education City, based on referrals from participants in Qatar. The study was not designed to assess the quality of any of the participant IBCs nor the individual staff members' performance. Rather, the study was focused on identifying methods or alterations that student affairs professionals have made in their professional practices to accommodate cultural differences in formal or informal ways.

Most of the participants in this study were American citizens who were working as student affairs staff members or administrators at an American IBC in Education City, Qatar. The sample consisted of student affairs professionals who had been recruited in the United States to work at IBCs in Qatar; it excluded student affairs professionals who had been recruited locally or regionally. No interviews were conducted with students at the participating IBCs. No interviews were conducted with faculty members, although these additional samples might provide greater depth of findings in future related studies.

Significance of the Research

Despite dramatic growth in the number of IBCs, there is still almost no research on the potential cultural conflicts that may present challenge student affairs practices. More than half of all IBCs worldwide are located in the Middle East (Arabian Peninsula) and Asia (C-BERT, 2017; Lane, 2011; Ziguras & McBurrnie, 2011). However, no comparative data currently exist from IBCs to address these challenges (Lane, 2011; Hou, Hill, Chen, and Tsai, 2018). For example, the legal agreements created by the IBC and the host country have prevented informational sharing and establishment of effective international networking opportunities (Anderson, 2016; Hou et al., 2018). The goal of this study was to fill this gap by exploring staff members' understanding of this phenomenon through student services that are implemented at IBCs in Qatar. Specifically, this study provided an opportunity for student affairs staff in Education City to voice their concerns and to describe their struggles and accomplishments in their work in an American IBC located in a traditional Eastern cultural context.

Preparing and training student affairs professionals before they arrive at their international assignments may have a lasting effect on the sustainability of IBCs. The findings of this study can help in understand the role of cross-cultural or intercultural training in the transition and adjustment process for these student affairs practitioners, especially those who were recruited in the United States or from Western developing countries that have significant cultural differences from those in the Middle East. The study findings can support efforts by U.S. institutions to establish successful internationalization initiatives and plans. The study findings may inform and guide institutional decisions related to establishing IBCs. Institutional leaders can use the information generated by the findings as a resource for recruiting and training student affairs staff for IBCs, demonstrating the institution's commitment to its internationalization plans and to host country needs.

Definition of Key Terms

The following key terms are defined as used throughout the study.

Globalization: "The broad economic, technological, and scientific trends that directly affect higher education and are largely inevitable in the contemporary world" (Altbach, 2006, p. 123). Globalization forces have pushed higher education institutions toward greater global involvement (Altbach & Knight, 2007).

Home country: The country that initiated the international collaboration through introduction of educational programs and degrees to a foreign country; it is considered to be the educational provider (Ziguras & McBurnie, 2011).

Host country: The country that hosts the foreign educational provider and permits that provider to establish a physical presence in the form of an IBC or other educational services, such as those that apply distant learning methods (Ziguras & McBurnie, 2011).

Intercultural/cross-cultural understanding: "The ability to interact effectively with people from cultures that we recognise as being different from our own" (Guilherme, 2000, p. 297).

International branch campus (IBC): An educational facility that is partially or fully owned and established by a home institution in a foreign country. IBCs have similar

infrastructures to those of the home campus, offer the same services, and engage in faceto-face teaching. IBCs offer academic programs that lead to awarding degrees recognized by the home countries of the institutions (Altbach, 2004; Farrugia & Lane, 2013; Lane, 2011; Marijk, 2003; Naidoo, 2008; Wilkins & Huisman, 2012).

Internationalization: Specific policies and programs that governments and higher education institutions and systems implement to deal with the challenges of globalization (Altbach, 2006).

Student affairs/student services: Divisions or departments designated to provide services and support to students in higher education to ensure their growth and development as they complete their academic coursework in college (Ciobanu, 2013; UNESCO, 2002).

Student mobility: The movement of postsecondary students across national and geographic borders to seek educational opportunities that might not be available to them at home due to cultural, socioeconomic, or linguistic reasons (Knight, 2006).

Western student affairs professionals: Staff members who work in higher education, especially in the United States, and have not lived in the Middle East. These staff members have minimal knowledge of Arab and Islamic culture and traditions, particularly those of the State of Qatar.

Organization of the Dissertation

The dissertation reports the study in five chapters. Chapter 1 provides a general background about internationalization efforts through IBCs' initiatives and identifies challenges facing IBCs and student affairs practitioners. Chapter 2 expands the discussion through the literature review addressing the following main issues: internationalization of

higher education, IBCs, the role of student affairs professionals, the concept of culture, the Arab world, cultural challenges facing IBCs, and the framework utilized in this study. Chapter 3 describes the methodology for the study. Chapter 4 presents the study's findings. Chapter 5 presents a discussion of the findings, implications of the results, and suggestions for future studies.

Chapter 2

Review of Literature

The purpose of this study was to explore how the local cultural context shapes Western student affairs services from the perspectives of student affairs practitioners working in IBCs in the Gulf Region. The study also examined how student affairs staff members adjusted their practices to meet the needs of their local students and to enhance student engagement. This chapter contains background information on (a) internationalization of higher education, (b) Middle Eastern trends in higher education, (c) IBCs, (d) student affairs services, (e) developments in student affairs and services, (f) the concept and role of culture, and (g) the theoretical frameworks utilized in the study.

Internationalization of Higher Education

As the internationalization movement expands in higher education, many scholars use certain key words to describe these efforts: international, global, or intercultural (Knight, 2003; Olson, Green, & Hill, 2006). *Internationalization* can be defined as any attempt to integrate international, global, or intercultural efforts into the mission, structure, or strategies of a higher education institution (Knight, 2003; Olson et al., 2006). *Global learning*, another key concept, enforces and supports efforts at internationalization by promoting

the knowledge, skills, and attitudes that students acquire through a variety of experiences that enable them to understand world cultures and events; analyze global systems, appreciate cultural differences; and apply this knowledge and appreciation to their lives as citizens and workers. (Olson et al., 2006, p. v)

Higher education institutions started their global efforts by (a) increasing enrollment of international students on their home campuses, and (b) moving their operations to other countries (McBurnie, 2000). Globalization can affect higher education in various aspects: financial, labor market, education, information technology, and networking (Carnoy & Rhoten, 2002).

Internationalization Versus Globalization

Many critics have indicated that internationalization has emerged as a strategy to respond to the strong forces of globalization (Olson et al., 2006). The terms *globalization* and *internationalization* have been used interchangeably in higher education but they refer to two distinct phenomena. Globalization can be viewed as a larger phenomenon guided by major governmental, political, and economical forces and leading to more global involvement by higher education institutions (Altbach & Knight, 2007). On the other hand, internationalization refers to specific policies and programs that higher education institutions implement to deal with the challenges of globalization (Altbach, 2006; Knight, 2003).

Another key difference between the two concepts is related to the perception of control. The globalization movement and its impact go beyond the control of any individual, organization, or single government. Thus, globalization provides the context for economic and academic trends that shape the world today (Altbach & Knight, 2007). However, internationalization can be an important institutional strategy to meet the increasing demands of the globalization movement, specifically in preparing students to engage in a globalized world (Horn, Hendel, & Fry, 2007; Knight, 2003; UNESCO, 2009).

Globalization has created a new opportunity for higher education institutions,

especially those in the United States, to assist in meeting the increasing global demands for access to postsecondary education (Altbach & Knight, 2007). However, at the same time, the increasing pressures resulting from the globalization movement have increased competition for these institutions (Stryker et al., 2016). Information technology also has played an important role in introducing new methods to educate students, such as online and long-distance programs. Some institutions have explored the option of establishing IBCs in partnership with foreign entities or governments, constituting another major form of internationalization (Carnoy & Rhoten, 2002; Stryker et al., 2016).

History of the Internationalization Movement

Universities have always been involved in international exposure to some degree, such as exploring the concept of universal knowledge, research initiatives, and movement of students, faculty, and scholars (Altbach, 1998). However, internationalization of higher education has quickly developed into a major phenomenon, encompassing curriculum and learning outcomes; enhanced cross-border program delivery; and competition for students, teachers, and scholars. These internationalization initiatives were particularly noticeable between the end of World War II and the end of the Cold War (de Wit, Hunter, Howard, & Egron-Polak, 2015; Smithee, 2012).

Most universities that were established in the 18th and 19th centuries exhibited a clear national focus rather than an international focus. Study abroad was not offered and local languages replaced the universal language, Latin (de Wit, 2002; de Wit & Meyer, 2012). Although there was a national focus for higher education institutions, there was a greater exchange of ideas regarding higher education structure, especially between

Europe and United States. For example, German universities contributed to establishment of modern American universities by establishing scholarship programs, creating necessary resources for research in libraries and scientific laboratories, and offering Doctor of Philosophy academic programs during the period 1880–1914 (Thelin & Gasman, 2011).

The internationalization movement started between World War I and World War II through formal global exchanges and collaboration within the higher education field (de Wit et al., 2015). This was illustrated by establishment of the IIE in the United States in 1919, the Deutscher Akademischer Austauschdienst in Germany in 1925, and the British Council in the United Kingdom in 1934 (de Wit et al., 2015).

After World War II, as Europe was dealing with recovery and focusing on rebuilding, the United States led efforts in the internationalization movement by establishing special programs such as the Fulbright Program (de Wit & Meyer, 2012; Smithee, 2012). The Fulbright Program is an exchange program allowing U.S. nationals to go abroad to develop an understanding of foreign cultures within an education context. The initial goal was to promote peace and mutual understanding. However, this goal shifted to the promotion of national security and foreign policy, especially during the Cold War between the United States and the Soviet Union (de Wit & Meyer, 2012; Smithee, 2012).

In the late 1990s, the internationalization movement shifted emphasis from political goals to economic goals in response to globalization pressures (de Wit et al., 2015; Smithee, 2012). The tragic events of September 11, 2001 led to an additional focus on understanding the Islamic religion, Arab culture, and Arabic languages (de Wit et al., 2015). Those events had a significant impact on the internationalization movement in the United States (Choudaha & de Wit, 2014; Ewers & Lewis, 2008; Johnson, 2018; Knight, 2013; Sirat, 2008).

Prior to September 11, 2001, the United States was viewed as the most popular destination for international students (Johnson, 2018; Smithee, 2012; Zeman & Adrian, 2005). However, many challenges appeared after that date, not only for new international students who had been planning to come to the United States but also for students who were currently studying in the United States. International students shared the perception that the United States was no longer a welcoming and friendly country because of stringent vetting procedures, visa denials, U.S. surveillance and monitoring systems, racial profiling, and hate crimes. These conditions made many students feel unsafe, especially students from the Middle East region (Ahmad, 2004; Johnson, 2018; Smithee, 2012; Zeman & Adrian, 2005).

While the United States was reforming its visa and immigration system, two major shifts took place with the global internationalization movement: (a) International students, especially from the Middle East region, started to seek alternative locations for study abroad, and (b) countries around the world started to develop equivalent educational opportunities, such as IBCs and educational hubs, in Hong Kong, Singapore, Qatar, and United Arab Emirates (Johnson, 2018; Sirat, 2008; Smithee, 2012).

The internationalization movement introduced new methods in student recruitment and mobility, as well as cross-border delivery programs, and focused on the need for capacity building across the globe (Carnoy & Rhoten, 2002: Stryker et al., 2016). Countries and universities became more proactive in expanding international activities and relationship building (de Wit et al., 2015). Internationalization is increasingly seen as an essential part of institutional missions, and universities are opting to develop more strategic approaches to their internationalization plans (de Wit et al., 2015; Stryker et al., 2016).

As the internationalization movement continued to expand globally, many institutions continued to explore its benefits and began to include it in their mission and strategic plans (Altbach & Knight, 2007). Key factors in expanding the internationalization movement were efforts to generate more revenues for the institution; meet the global demand for higher education; enhance cross-cultural education for faculty, students, and staff; and satisfy student desires to take advantage of such international education opportunities (Altbach & Knight, 2007). The International Association of Universities (IAU) highlighted a recent important study that explained the benefits of the internationalization movement (Egron-Polak & Hudson, 2014).

Benefits of Internationalization

In April 2014, IAU published its fourth global survey on internationalization of higher education, reporting data from 1,336 higher education institutions in 131 countries (Egron-Polak & Hudson, 2014). The IAU global survey project was designed and implemented with support from the European Commission, the British Council, the Association of International Educators (NASFA), and the European Association for International Education (EAIE). The report is considered to be one of the most comprehensive collections of primary data on internationalization of higher education (Egron-Polak & Hudson, 2014). Even though there was variety in the respondent institutions' answers, the researchers identified eight common reasons for considering internationalization: (a) to improve the overall quality of education, (b) to prepare students for a global world, (c) to increase recruitment of international students, (d) to improve institutional reputation and ranking, (e) to improve the quality of research development, (f) to establish competition among educational institutions, (g) to meet the labor market's needs within a global context, and (h) to increase financial benefits for the institutions (de Wit et al., 2015; Egron-Polak & Hudson, 2014).

The major consistent theme throughout the findings was that institutions worldwide were focusing on internationalization. For example, 75% of the respondent institutions indicated that they had an established internationalization plan or were in the process of establishing a plan. More than 15% indicated that internationalization was part of their overall institutional strategic plan (de Wit et al., 2015; Egron-Polak & Hudson, 2014).

The report clearly highlighted the importance of internationalization based on responses to the survey (de Wit et al., 2015; Egron-Polak & Hudson, 2014). Key benefits for students included an increase in international awareness and engagement in global matters. Numerous institutions reported that internationalization efforts had assisted in enhancing the quality of instruction and student learning. International collaboration and cultural exchange were highlighted as important key benefits. Financial incentives were ranked lowest by the respondents (de Wit et al., 2015; Egron-Polak & Hudson, 2014).

Although 1,336 worldwide institutions responded to the 2014 IAU survey, the lowest rate of response came from the Middle Eastern region (Egron-Polak & Hudson,

2014). The IAU reported that only 3 of 60 institutions responded to the survey, which made it difficult to generate a perspective on the Middle East region (Egron-Polak & Hudson, 2014). Additional challenges were reported within the Middle East region in the Arab Human Development report published by the United Nations Development Programme (UNDP) in 2003 and in 2011 (UNDP, 2003, 2011).

Middle Eastern Trends in Higher Education

The original Arab Human Development (AHD) report published in 2003 highlighted the poor condition of higher education throughout the Middle East region (UNDP, 2003). The report stated that the number of students enrolling in higher education institutions was generally low in Arab countries, compared to nearby countries that were investing in education, such as Korea. Furthermore, the report noted a lack of scholarly work in the Arab countries, as they had contributed only 1.1% of publications worldwide, especially in the science and technology fields (UNDP, 2003).

The updated AHD report published in 2011 acknowledged some progress in enhancing educational services; however, this progress was inconsistent across the Arab countries (UNDP, 2011). Additional areas of concerns included the absence of parentteacher relationships, lack of funds dedicated for education, physical problems with the schools' facilities and resources, and the overall low quality of the educational services delivered (Romani, 2009; UNDP, 2011).

Concerns focused on the lack of connection between the higher education sector and the local job market (Romani, 2009; UNDP, 2011). This disconnect led to an increase in unemployment rates because of the high number of graduates with no job opportunities. Potentially based on the reported poor quality of education, investment by Arab companies in developing the workforce was considered to be the lowest in the world (UNDP, 2011).

The countries of the Gulf Cooperation Council (Kuwait, Bahrain, Saudi Arabia, Qatar, UAE, and Oman) have responded to these educational challenges by investing billions of dollars to improve their higher education infrastructure (Carnoy & Rhoten, 2002; Romani, 2009). The growth of the youth population in the region has resulted in a high demand for higher education. Emergent countries, such as Qatar and UAE, have played a significant role in advancing access to higher education in the Gulf Region (Romani, 2009; Sawahel, 2017). Qatar and UAE have intentionally and strategically attracted many Western higher education providers to establish IBCs, and as a result, now serve as important educational hubs within the region. However, it is not clear that hosting IBCs alone will resolve all of the educational challenges in the Arab region (Romani, 2009).

Currently, the Middle East region hosts more than 30% of the total number of IBCs in the world. The United States, United Kingdom, Australia, and Canada continue to serve as the main providers of IBCs worldwide (C-BERT, 2017).

Development of IBCs

Internationalization efforts by higher education institutions go back to medieval times, when universities in Europe recruited international students and faculty (Altbach & Knight, 2007). Some successful U.S. internationalization efforts can be traced to the 1930s, when Florida State University provided educational programs overseas to meet the needs of the U.S. military and other personnel in the U.S. Panama Canal Zone (Lane, 2011). Johns Hopkins University established the first and most sustainable IBC in Italy in the 1950s to provide graduate programs in international relations (Lane, 2011; Verbik & Merkley, 2006).

The second wave of international education began in the 1990s and continued in the 2000s, focused on expanded efforts by Australian, United Kingdom, and U.S. universities to establish IBCs in countries such as UAE, Bahrain, Qatar, Singapore, Hong Kong, and Malaysia (Knight, 2011; Redden, 2015a, 2015b). The establishment of these educational hubs was intended to create a knowledge-based society by attracting foreign education providers, recruiting and retaining local and international students, and providing a world-class education and professional training opportunities. These countries, through their educational hubs, accounted for 80% of IBC growth during this period (C-BERT, 2013; Redden, 2015a, 2015b). The State of Qatar and the UAE focused their efforts on establishing themselves as educational hubs to become major educational providers within the region (Knight, 2012b).

The third wave of internationalization focused on continuing growth and development of IBCs and broadening markets in the host and home countries with interest in these joint ventures (Redden, 2015a). From 2012 to 2014, there was a steady increase in the number of countries hosting IBCs, from 53 to 71. In addition, there was a shift in the markets from a focus on the Arabian Peninsula to a focus on Asian countries. The number of foreign educational providers increased from 24 to 30, with new educational institutions emerging in Russia and South Korea, for example (Redden, 2015a). Recent developments include a new higher education provider: the Islamic Republic of Iran (Sawahel, 2017). Iranian universities have opened nine branch campuses, mostly in Arab countries such as UAE, Iraq, Syria, Lebanon, Oman, Qatar,

and Kuwait (C-BERT, 2017; Sawahel, 2017). Eight more Iranian IBCs are under development or scheduled to open soon around the world (C-BERT, 2017; Sawahel, 2017).

Education City in Qatar

One of the main goals for establishing Education City in the State of Qatar was to support creation of a knowledge-based economy by promoting research, innovation, and technical skills through higher education (Rostron, 2009). The Qatari government leaders hope that this change will positively influence the economy and generate financial returns in the near future (Rostron, 2009). A small country in both size and population, Qatar is known for its high per capita income and rapid development, chiefly due to national revenues from gas and oil industries (Donn & Al Manthri, 2011).

The QF was established in 1995 and chaired by Her Highness Sheikha Mozah Al Missned (Rostron, 2009). The foundation played a significant role in establishing Education City. Fathi Saud, the first president of the QF, stated that they originally aimed to recruit one university that would offer several academic programs (Krieger, 2008). The University of Virginia was one of the strongest options, according to Saud. However, due to problems with ranking criteria, it was deemed best to go with the second option: attracting several universities that represented the best in their fields (Krieger, 2008). The selection process was rigorous; decisions were based mainly on the universities' programs, reputation, and ranking in their home countries (Lane & Kinser, 2011). The selection process mandated creation of an internationally focused curriculum in various academic fields, leading to diversification of the academic programs to be offered in Qatar (Knight, 2012b). As part of this project, the QF provided state-of-the-art facilities, including classrooms, laboratories, student centers, and excellent housing facilities. The QF covered the full operational expenses for the branch campuses, including salaries for faculty and staff, facility management, research grants, and student affairs services and programs. The agreement between the QF and the foreign institutions specified that the branch campuses would have full control and autonomy in management and implementation of their academic programs. This agreement is scheduled to be reviewed and renegotiated every 10 years. The QF continues to manage the budget for this project (Knight, 2012b; Lane & Kinser, 2011).

The government of Qatar covers the full tuition cost for Qatari nationals, who constitute more than half of student enrollment in Education City (Krieger, 2008); the remaining students represent more than 80 nationalities (Donn & Al Manthri, 2011). The QF offers a financial aid program that includes various levels and types of financial support through scholarships, loans for students with financial needs, and assistance through a student work program. This financial support for students demonstrates Qatar's commitment to increase access to higher education for students from the country and students from outside the country (Lane & Kinser, 2011).

Student Affairs Services

The concept of student affairs services as a profession is still relatively new in Qatar (Shea, Gormley, Clarke, & Leary, 2016; UNESCO, 2002). In the United States, the term *student affairs* is used to describe a wide scope of campus operations housed in several departments and provided by professionals with various educational backgrounds (Long, 2012). For the purpose of this study, the definition offered by NASPA is used: a

professional who provides "services, programs, and resources that help students learn and grow outside of the classroom" (NASPA, 2018, para. 1). Professionals in student affairs services aim to provide a wide and diverse variety of services to "enhance student learning, guide academic and career decisions, mentor students, promote leadership skills, and counsel students through crises" (NASPA, 2018, para. 1).

Student affairs offices can include the following service units: academic advising, admissions, nontraditional student services, auxiliary services (e.g., bookstore, dining), student activities and events, career services, spiritual and religious services, precollege programs, counseling services, disability services, financial aid and student employment, health services, international student services, housing, multicultural services, athletics, campus safety, alumni services, student registration and records, and student conduct (der Heyde, 2016; Shea et al., 2016; UNESCO, 2002, 2009).

Historically, student affairs in the United States adopted the philosophy of *in loco parentis*, a Latin term meaning "in place of a parent" (Komives, Woodard, & Associates, 2003; Long, 2012). This approach provided colleges and universities with the legal authority to oversee students closely, with strict adult supervision. Student activities were present at many early colleges, including debating societies, fraternities, and secret societies (Dungy & Gordon, 2011). The faculty and clerical support staff coordinated nonteaching responsibilities such as student housing, food services, academic advising, and student conduct (Long, 2012; Manning et al., 2014). In fact, all members of the college community, "professor, instructor, dean, registrar, adviser, coach, proctor, yes even janitor," shared the responsibility of student services in the early stages of the profession (Manning et al., 2014, p. 54). This image is still apparent in many universities

around the world, as the student affairs profession is not as fully developed in many countries as it is in the United States (Perozzi, Osfield, Moscaritolo, & Shea, 2016; Shea et al., 2016). As demands on faculty members increased, expanding their roles and responsibilities, student affairs practitioners took more responsibility for student welfare (Dungy & Gordon, 2011; Manning et al., 2014; UNESCO, 2002).

In the United States, a fundamental change for student affairs professionals took place during the 1960s (Komives et al., 2003; Long, 2012). The relationship established between students and the student affairs professional changed with the dismissal of the concept of *in loco parentis* in the U.S. Court of Appeals decision in *Dixon v. Alabama State Board of Education* in 1961. In that case, the Court ruled that any student age 18 or older was to be considered a legal adult, with all concomitant rights and responsibilities, such as due process concerning violation of university rules and regulations. The Court decision had major implications for student affairs philosophy and practice. Student affairs practitioners moved away from being punitive figures to educating students regarding making appropriate choices and decisions as adults (Komives et al., 2003; Long, 2012). This shift promoted development of the modern doctrine that is now implemented by student affairs practitioners in their programs and services offered to students.

Outside of the United States, worldviews and approaches to the concept of *student affairs* or *student services* are generally based on the local context of each country or region (Perozzi & O'Brien, 2010; Perozzi, Osfield, et al., 2016). In countries that have adopted the concept of *student services*, the focus tends to be on delivering services that students need to complete their education (Perozzi, Osfield, et al., 2016). However,

countries that are more advanced in this profession have generally adopted the concept of *student affairs* because it goes beyond providing basic services to a consideration of the holistic growth and development of students outside the classroom. The student affairs concept has been implemented mainly in the United States, Canada, and Australia (Perozzi, Osfield, et al., 2016).

UNESCO published two reports confirming the significant role of student affairs professionals in educating and developing college students (UNESCO, 2002, 2009). UNESCO acknowledged that student affairs practices have developed only in countries that have (a) the appropriate infrastructure, (b) access to and demand for higher education, and (c) the cultural context to support such practices (UNESCO, 2002, 2009). Those conditions emphasized the importance of a dedicated, educated, and experienced staff that supports development of the student affairs field internationally (Moscaritolo & Roberts, 2016).

Student Affairs Practitioners

Student affairs professionals take the lead in personal development and growth of students and serve as advocates for student success (Blake, 2007; Ciobanu, 2013; Evans et al., 1998; Magolda, 2003; Manning et al., 2014; Schuh et al., 2011; UNESCO, 2002). Many out-of-classroom activities enhance student learning and development, especially those related to practical competencies. For example, managing student government operations and budgeting, publishing articles in the student newspaper, participating in a university band, or being a part of an athletic team can help students to develop and practice skills that employers value, such as teamwork, problem solving, decision making, and time management (National Institute for Learning Outcomes Assessment

[NILOA], 2010). Furthermore, student affairs practitioners challenge students to enhance critical thinking skills, one of the important goals in higher education (Magolda, 2003; Ciobanu, 2013). These lifelong competencies can aid in the holistic development of the student as a human being and aid in the student's becoming an active member of the global community (Ciobanu, 2013; UNESCO, 2002).

Student affairs professionals play an important role in supporting academic achievement, enhancing student learning, and increasing retention (Blake, 2007; Ciobanu, 2013; UNESCO, 2002). Integration of academic programs and co-curricular activities are examples of how academic and student affairs professionals work together to enhance student learning and achievement in college (Blake, 2007; Manning et al., 2014). Student affairs professionals are also increasingly involved in classroom teaching. In addition, they assist faculty members who are interested in incorporating cocurricular activities or student development activities in their classrooms (Magolda, 2003; Manning et al., 2014). Student affairs practitioners who work outside the United States play similar roles in supporting and advancing academic success of their students (Deardorff, 2016a; der Heyde, 2016). IBCs provide these practitioners with the platform to promote development and growth of students and to provide a Western educational experience without the need for the students to travel (Stryker et al., 2016). The growth of IBCs, especially in the Middle East, is impressive, with more than 53 IBCs in the UAE and the State of Qatar alone (Stryker et al., 2016). However, this growth may present challenges due to conflicts between the host country's culture and the Western institutional culture of the educational providers (Stensberg, Sliva, & Medina, 2016).

32

Student Affairs in the Middle East

The student affairs professional who is recruited from Western societies to work in the Middle East may face major challenges in adjusting to strong cultural identities in this region (Altbach & Knight, 2007). The Qatari government designed Education City as an American educational hub and recruited professionals (mainly Americans) to fill key positions as faculty and staff (Vora, 2014). As a result, the project has generated local concerns about potential American Western influence and its impact on the loss of Arabic language, Muslim values, and Qatari traditions (Vora, 2014).

Student affairs professionals' duties and responsibilities are influenced by personal beliefs and values, which can make a significant impact on students' overall experiences in college (Ciobanu, 2013). Culture is a socially constructed phenomenon that must be understood in terms of how it shapes students' lives to achieve multicultural and global understanding (Magolda, 2003; Ciobanu, 2013). In order to be effective in their work, student affairs practitioners must account for local, regional, and national circumstances to establish successful management and leadership models for their services (Moscaritolo et al., 2016; Stryker et al., 2016; UNESCO, 2002).

Currently, there is a lack of research to describe the work of student affairs and services internationally (Kruger et al., 2016; Moscaritolo et al., 2016; Seifert, Perozzi, & Li, 2016; Stryker et al., 2016). The International Association of Student Affairs and Services (IASAS) was one of the first organizations to conduct research on student affairs practices around the world (Seifert, Perozzi, Bodine Al-Sharif, Li, & Wildman, 2014). The IASAS study involved 622 participants from 36 countries who responded online to English-language open-ended items. The world regions covered in the study were North America, Central America, the Caribbean, Australia and New Zealand, and Europe. A small number of responses were received from the Middle East (16 participants) and Africa (12 participants). No responses were recorded from South America or Asia (Seifert et al., 2014).

One of the important findings regarding student affairs practitioners' credentials was that 90% indicated that they held a bachelor's degree or equivalency (Seifert et al., 2014). However, many had undergraduate degrees that did not relate directly to their work in student affairs (Seifert et al., 2016). The researchers identified challenges in development of student affairs as a global profession because of a lack of opportunities to gain essential educational credentials outside the United States. Furthermore, due to variation in credentials and educational backgrounds among student affairs practitioners across the world, finding a common language to develop conversations among colleagues is currently a major challenge and may become even more difficult in the future (Seifert et al., 2016).

A study by Stryker et al. (2016) focused on student affairs services in Asia and the Middle East. One hundred participants from China, Qatar, United Arab Emirates, Kuwait, Oman, Afghanistan, Lebanon, Caribbean-Lesser Antilles, Singapore, and Malaysia completed the online survey and 26 selected participants were interviewed by the researchers (Stryker et al., 2016; C. Stryker, personal communication, February 4, 2018). Similar to the IASAS study, the participants expressed concern regarding professionalization of the student affairs field. They identified deficiencies in appropriate higher education credentials and a lack of professional development opportunities outside the United States. They expressed concern about lack of funding for student affairs programs, cultural challenges, safety and security, and political issues. Several participants expressed concern about how student affairs work is perceived by academic affairs personnel as limited in significance (Stryker et al., 2016).

Even though both studies addressed many important issues facing student affairs practitioners around the world, neither directly addressed the large population of student affairs practitioners who currently work at more than 280 IBCs around the world, particularly Western student affairs practitioners who were educated in the United States. In the IASAS study, participants from the United States reported holding advanced educational degrees in higher education, educational leadership, or student affairs administration (Seifert et al., 2014). However, the study failed to report whether these professionals had encountered challenges in working in IBCs regarding implementation of Western practices in other cultural contexts.

Although Stryker et al. (2016) did not directly address Western student affairs practitioners, some participants reported cultural challenges in their new environment. These participants emphasized the need for more meaningful cultural dialogs to enhance their knowledge and understanding and to assist them in navigating cultural values and norms in the host country (Stryker et al., 2016). Other scholars have highlighted the same cultural challenges and emphasized the need for student affairs practitioners to acquire cultural competence and intercultural communication skills (Berardo & Deardorff, 2012; Deardorff, 2016a). Further research is needed to address the impact of local cultural communications can assist in advancing student affairs work globally.

Western Student Affairs Practice Models

Manning et al. (2014) identified various student affairs practice models that have shaped the work culture of student affairs practitioners in the United States. These traditional models, specifically the out-of-classroom-centered models, represent the early stages of student affairs development in the United States. These traditional models are currently considered "out of date" in the United States because they do not promote collaboration between student affairs and academic affairs offices nor do they support the holistic developmental approach for the students (Manning et al., 2014, p. 70). These models, especially the extracurricular model, are common in the Middle East region in local public and private universities (Kruger et al., 2016; Manning et al., 2014; Perozzi & O'Brien, 2010; Perozzi, Giovannucci, & Shea, 2016). Two models have been selected to illustrate the differences in these approaches: (a) the extracurricular model, and (b) the academic-student affairs collaboration model (Manning et al., 2014).

The extracurricular model. The prefix *extra* indicates that the activities are additional to or not connected to any academic requirements (Manning et al., 2014). In this model, there is a clear separation between efforts conducted by student affairs offices and efforts conducted by academic affairs offices (Kuk, Banning, & Amey, 2010; McClellon, Stringer, & Associates, 2009). In some cases, student affairs practices may even conflict with the academic mission (Manning et al., 2014).

There are many strengths and weakness in the extracurricular model. The model allows student affairs offices to become specialized in their services, allows faculty to focus on teaching and research, and provides students with unique developmental opportunities beyond the academic sphere. On the other hand, the model lacks focus on collaboration between the academic and student affairs offices, relies on an individual team approach, and may cause confusion about the mission and purpose of the institution (Manning et al., 2014).

The academic-student affairs collaboration model. In the academic-student affairs collaboration model, academic affairs offices and student affairs offices have intentional and significant interactions to establish shared programs to enhance student learning (Manning et al., 2014). In this model, student affairs professionals work side by side with faculty members to use each other's strengths to coordinate the educational mission for students. The main goal of this collaboration between student affairs and academic affairs is to "develop a shared language around the student success" (Manning et al., 2014, p. 164).

For example, student affairs staff members can apply their knowledge of leadership development to assist in creation of a senior capstone course. Another popular example is the collaboration between student affairs professionals who work in residence halls and an academic department to establish a living learning community or theme housing on campus. These experiences enhance student learning through special programs that are developed and delivered by both groups: student affairs and academic affairs (Manning et al., 2014).

The strengths of the academic-student affairs collaboration model include a teamoriented environment created by both sides and the opportunity to establish a direct connection between the work of student affairs and the educational mission of the institution. Additional benefits, such as sharing expenses and resources, are supported in this model (Manning et al., 2014). On the other hand, challenges have been reported in establishing such a collaboration. Some student affairs departments have reported a challenge in taking on a larger responsibility for planning and advancing collaborative opportunities or programs. Furthermore, in most cases, student affairs services tend to initiate the discussions for the collaboration model, instead of academic affairs (Manning et al., 2014).

Student affairs models outside the United States. Perozzi, Giovannucci, et al. (2016) argued that adopting Western student affairs models may be unsuccessful because the local context can play a significant role in the process. In some countries, student affairs services are still struggling to receive recognition and validation as a profession. Major challenges in these countries include a lack of professional associations to represent student affairs practitioners, confusion about the role of the profession, and a lack of research supporting the work of student affairs in these regions (Perozzi, Giovannucci, et al., 2016).

Student affairs services can be part of the university's structured system or they can be outsourced to specialized private companies (Ciobanu, 2013; der Heyde, 2016). Kruger et al. (2016) indicated three models that student affairs have adopted across the world: (a) the cooperative model, (b) the public services model, and (c) the university-based model. In the cooperative model (seen mostly in Japan and Korea), services are provided to students in a business-oriented approach. These services could include housing, food services, library management, bookstore, and insurance plans (Kruger et al., 2016). In the public services model (seen mostly in Europe), nonprofit service providers manage and deliver such services as housing, food services, student activities, and counseling services. In the university-based model (seen mostly in North America

and most of Asia), the operations and delivery of student affairs services are directly connected to the institution. In this model, the university or college takes full responsibility for all aspects of student life under the leadership of student affairs practitioners. In this model, student affairs practitioners apply a holistic approach for student development and growth. Due to advanced support from the institution, financial resources and highly qualified staff are selected to provide student affairs services in this model (Kruger et al., 2016).

Developments in Student Affairs and Services

Although student affairs practitioners share common goals for enriching student experiences globally, the philosophy and delivery of student affairs services can differ across the world (Osfield & Associates, 2008; Perozzi, Osfield, et al., 2016; Shea et al., 2016). The philosophies and delivery of services vary based on the country, the institutional type and size, and resources available at the particular institution (Ciobanu, 2013; Manning et al., 2014; Moscaritolo & Roberts, 2016; UNESCO, 2002).

Developments in Europe. In European countries, the mission of student affairs professionals is to promote access to higher education, improve student retention, and assist students to achieve academic success (der Heyde, 2016). Similar to the United States, the European Higher Education Area (EHEA) adopted the holistic approach and philosophy for the student affairs profession. However, implementation of this approach can be difficult because students are encouraged to become more independent in seeking services and programs offered by student affairs practitioners (der Heyde, 2016).

The EHEA was launched in 2010 after the signing of the Bologna Joint Declaration by 29 European countries. The main goal of the EHEA is to move toward a more transparent and cohesive higher education system in Europe. As of 2015, 48 European countries had joined the EHEA agreement. Student support units, as part of student affairs, have been developed and implemented in EHEA countries, including support for students with disabilities. However, there are still some major deficiencies in the services, and improvements across the EHEA are needed (der Heyde, 2016).

Student affairs services in EHEA can be structured into two ways: (a) coordinated directly by the institutions, or (b) outsourced to private companies (der Heyde, 2016). Countries in Eastern Europe tend to follow the first structure (operating within the institution), while countries in Western Europe tend to follow the second structure (operating from outside the institution). Currently, there is a movement to change the structure in Eastern Europe to become similar to that in Western Europe by outsourcing student affairs services. Some outsourced companies take a nonprofit approach (as in Poland), while others take a for-profit approach (as in Ireland, the United Kingdom, and Spain). This division of how student services and programs are offered in EHEA can present challenges in implementing a holistic approach by student affairs practitioners and can increase overall student debt (der Heyde, 2016: Moscaritolo et al., 2016).

Developments in the Middle East. National reports prepared by student affairs practitioners around the world have highlighted major challenges in the structure of student affairs services, especially in the Middle East (Callahan, 2016; UNESCO, 2009). Reports from Kuwait, Lebanon, Morocco, Qatar, Saudi Arabia, and the UAE highlight difficulties in understanding the purpose of student affairs services, especially regarding student development. Student affairs practitioners are viewed as service providers rather than as educators by the university and the local community (Perozzi, Giovannucci, et al.,

2016; UNESCO, 2009). This lack of understanding has caused major challenges in terms of recognition of the student affairs profession and securing financial support for student affairs services and programs (Perozzi, Giovannucci, et al., 2016; Shea et al., 2016; UNESCO, 2009).

Another challenge faced by student affairs services professionals is a lack of standardized structure, appropriate training, or educational programs for student affairs practitioners in the Middle East region and across the world (der Heyde, 2016; Perozzi, Giovannucci, et al., 2016; UNESCO, 2009). Most institutions in the Middle East focus on offering only basic services, such as housing, admissions, student registration, and recreational activities, including athletics (UNESCO, 2009). Failure to offer other essential student services, such as disability services, can have a potentially negative impact on the overall student educational experience and the student's ability to persist to complete a college education.

The student affairs staff at the American IBCs in the Middle East are facing similar challenges in adopting the student affairs models from the United States. IBCs usually follow the same structure and practices as those implemented on their home campuses. However, student affairs practitioners working in IBCs tend to utilize outdated practice models that no longer exist in the United States, such as the extracurricular model (Farrugia & Lane, 2013; Manning et al., 2014; Vora, 2014). As a result, there are inconsistencies between student affairs practice models that are implemented on home campuses and those that are implemented at their IBCs in Qatar (L. Tillman, International Programs Coordinator, personal communication, September 18, 2017; Vora, 2014). It is imperative to explore why student affairs practitioners have selected these outdated

41

models for their practice at IBCs, especially in the Gulf region.

Future Global Challenges for Student Affairs

In 2008, Osfield and Associates published *Internationalization of Student Affairs and Services: An Emerging Global Perspective*. In this edited publication, many authors highlighted the following challenges that face student affairs services globally: (a) the need to establish counseling services, (b) the need to secure more funds for student affairs programs, (c) the rising costs of higher education, (d) the need to establish more collaboration among institutions, and (e) the need to establish an international student affairs and services association (Osfield & Associates, 2008). Moscaritolo et al. (2016) offered a reflection on some of the concerns discussed by Osfield and Associates (2008) and offered recommendations on how to move forward.

One of the main concerns that students are experiencing around the world is the increased cost of higher education, leading to increased student loan debt worldwide (Moscaritolo et al., 2016). These conditions are reported not just in United States but also in countries such as the United Kingdom and Japan. For example, the average loan debt for students graduating in the United Kingdom doubled from £12,360 to £26,000 from 2008 to 2012 (College Stats, 2013). In Japan, the number of students who applied for student loans increased by 70% from 2001 to 2011. Similar increases in tuition and student debt have been reported in Australia, China, and Canada (College Stats, 2013).

Even though the cost of higher education is increasing globally, a few countries still support the idea of free public higher education, including Argentina, Iceland, Norway, and Sweden (Moscaritolo et al., 2016), although some of these countries are reviewing their policies and practices in terms of their ability to continue to offer free public higher education. This concern is likely to grow in the next few years; it is important to find ways to assist students to complete their education and start a career without significant debt (Moscaritolo et al., 2016).

The need for counseling services was originally mentioned by Osfield and Associates in 2008. Today, despite an increase in student mental health issues in many countries, the concept of offering counseling services on campuses is challenged worldwide (Moscaritolo et al., 2016). To manage academic assignments and studies, students must be fit, both physically and mentally. Thus, as part of the holistic approach, many student affairs practitioners support promotion of all aspects of student wellness on campus. The roles of the psychologist, counseling staff, and wellness coordinator become essential in reducing stress that students encounter and assisting students with mental health issues (Moscaritolo et al., 2016).

Moscaritolo et al. (2016) and Osfield and Associates (2008) also highlighted the need for student affairs practitioners to develop global competence in intercultural communication. This need is based on a significant increase in the number of student affairs professionals who work outside their native country (Moscaritolo et al., 2016), particularly in transnational education initiatives, such as the IBCs (Stryker et al., 2016). The most recent estimate is that there are approximately 283 IBCs in 77 countries (Stryker et al., 2016).

Student affairs practitioners must be aware of cultural differences when they decide to work internationally (Moscaritolo et al., 2016). A challenge that faces many Western student affairs practitioners is how to modify practices in consideration of the local cultural context. In order to succeed, student affairs practitioners must be flexible

and adaptable to their new work environment (Moscaritolo et al., 2016; Stryker et al., 2016). Skills such as intercultural competence and intercultural communication are essential to enhance student affairs practitioners in understanding their students' needs. In fact, this understanding constitutes the core value of the work by student affairs professionals. Therefore, it is critical that student affairs services ensure that all of its professionals, especially those who are working internationally, seek actively to become competent in these skills based on the local cultural context (Deardorff, 2016a; Moscaritolo & Roberts, 2016; Moscaritolo et al., 2016; Stryker et al., 2016).

The Concept and Role of Culture

Culture has been viewed and defined differently across the world. Collier, Hegde, Lee, Nakayama, and Yep (2001) considered culture to be the shared language, beliefs, and values that are commonly accepted and enforced in a community. People in a specific community would be expected to speak the same language and share the same norms, values, beliefs, and traditions (Collier et al., 2001; Gannon & Pillai, 2016). Martin and Nakayama (2010) defined culture similarly as the "learned patterns of behavior and attitudes shared by a group of people" (p. 84).

Communities share similar traditions: "myths, legends, ceremonies and ritual" (Ting-Toomey, 1999, p. 11). For example, people in the United States celebrate Thanksgiving. Cultural identity is the product of individual interactions and socialization within certain cultural members' groups. These memberships are developed and influenced by parents and they ultimately influence children's cultural identity formation process. It is important to note that cultural values and traditions influence the development of norms for a given community; while cultural values are difficult to detect, norms can be detected easily through behaviors (Ting-Toomey, 1999). Failure to know or understand cultural norms can lead to serious misunderstanding and confrontation between cultures. Those who violate cultural norms may not even recognize the impact of their actions (Ting-Toomey, 1999).

The role of culture can serve five essential functions (Ting-Toomey, 1999). First, culture assists in developing individual identity. Second, culture encourages the development of group membership; as a result, people can feel safe through a sense of belonging to a group that confirms their behaviors and beliefs. Third, culture establishes rules that govern internal or external interactions (inside and outside established groups) through attitudes. Fourth, culture adapts, changes, and evolves in response to individual and community needs. Fifth, culture and communication are interconnected through reciprocal influences to enhance overall experiences. A culture cannot be understood without a means to communicate it, and communication ensures the continuation and survival of cultural values and traditions from one generation to the next (Ting-Toomey, 1999).

Therefore, culture is recognized as a core concept for intercultural communication (Gudykunst, 2002; Ting-Toomey, 1999). Research in intercultural communication is often focused on studying how cultural groups vary, for example, how Muslims are different from Christians, Americans are different from Japanese, men are different from women, old people are different from young people, and so on (Gudykunst, 2002). Gudykunst (2002) highlighted the differences between cross-cultural and intercultural research. Even though some authors use these terms interchangeably, cross-cultural communication is focused on comparing interactions of certain people from one specific

45

culture to another culture, while intercultural communication is a larger concept that involves the study of interactions across cultures (Gudykunst, 2002; Lustig & Koester, 2012). This study focuses on intercultural communication.

Intercultural Communication

Communication can be defined as the process by which people express information, ideas, and feelings (Martin & Nakayama, 2010). Communication takes place whether or not meaning is attached to it; communication can be intended or unintended. Symbols are part of the communication process, including written words and verbal or nonverbal symbols.

Nonverbal communication can be defined as "nonlinguistic behavior" that can be expressed intentionally or unintentionally through various communication methods (Ting-Toomey, 1999, p. 115). Nonverbal symbols include body language, such as gestures, facial expressions, body location, touching, and eye movement (Martin & Nakayama, 2010). Nonverbal communication includes universal tools for use around the world. However, nonverbal communication can be culture specific, as certain cultures set rules for "when, where, with whom, and how these different emotions should be expressed or suppressed" (Ting-Toomey, 1999, p. 115).

Remarkably, humans can produce about 700,000 physical signs, 25,000 of which are facial expressions (Axtell, 1998). Body language is considered to be culturally specific, indicating that a gesture or expression or posture may mean one thing in one culture but something different in another culture (Axtell, 1998; Martin & Nakayama, 2010; Ting-Toomey, 1999). For example, former U.S. President George W. Bush used to make the "Hook 'em Horns" hand gesture, which is a well-known symbol among fans of the University of Texas at Austin (Martin & Nakayama, 2010; University of Texas-Austin, 2018). However, many people around the world view this sign in a completely different light. Norwegians view the symbol as the "sign of the devil" (Martin & Nakayama, 2010, p. 275). People in Central and South America view the sign as stating that "someone's wife is unfaithful" (Martin & Nakayama, 2010, p. 275). The U.S. Joint Forces Command cited this gesture as an example of how miscommunication can affect the global image of the U.S. government negatively (Martin & Nakayama, 2010).

As the world becomes more diverse, there is a growing need for better understanding among the diverse cultures around the world to avoid incidents of misunderstanding, intercultural conflicts, and friction (Martin & Nakayama, 2010; Ting-Toomey, 1999; Zweifel, 2003, 2013). Intercultural communication involves processing and interpreting messages that move from one culture to another. One of the most common reasons for miscommunication is that different meanings may be assigned to the same symbol by different cultures (Martin & Nakayama, 2010; Ting-Toomey, 1999; Zweifel, 2003, 2013).

Intercultural communication also involves material culture (Crane & Bovone, 2006; Hicks, 2010; Martin & Nakayama, 2010). *Material culture* is typically defined as any physical item that can be seen, touched, or felt. Clothes are an example of material culture, and specific meanings are attached to clothes (Crane & Bovone, 2006; Hicks, 2010; Martin & Nakayama, 2010). For example, many Muslim women wear the *hijab* or head scarf to express modesty. However, this item of material culture was viewed differently by many Americans after the terrorist attacks on the World Trade Center and the Pentagon. Many Americans viewed Muslim women who wore the *hijab* as the enemy and many women were attacked because Americans associated the *hijab* with the women's religion. Today, most Muslim American women who wear the *hijab* do so to symbolize their religion and to show pride in being Muslim (Martin & Nakayama, 2010).

Communication and culture are interconnected (Martin & Nakayama, 2010). In a way, communication reinforces culture and culture influences communication by enacting it in countless ways. Cultural forms (e.g., rituals, myths, social interactions) are performed through a structure of norms, conversations, and interactions. The exchange of these cultural forms through communication affirms the cultural identity of a community, thereby helping to create cultural reality for the community (Martin & Nakayama, 2010; Ting-Toomey, 1999).

Cultural forms take various shapes across the world (Martin & Nakayama, 2010). For example, certain rituals and traditions are carried out differently in the Arab World than they are in the United States. Understanding specific regional cultural expectations can assist in explaining the behaviors of people in specific regions. This dissertation study explores the Arab world regional culture and highlights differences between Eastern and Western cultures.

The Arab World

People from Western countries use the terms *Arab world* and *Middle East* interchangeably, presuming that they have the same meaning (Barakat, 1993; Nejem, 2016). However, the terms have different meanings to Arabic people. The *Middle East* is a smaller geographical area than the *Arab world*; it includes Jordan, Lebanon, Syria, Palestinian territories, and in some definitions Saudi Arabia, Kuwait, Iraq, and Iran. On the other hand, the *Arab world* includes all 22 countries in the Arab League of Nations that predominantly speak the Arabic language and share similar histories, cultures, and religions. As of 2014, the estimated population of the Arab world was approximately 385.3 million, compared to 200 million in 1993 (Barakat, 1993; Nejem, 2016).

The Arab world can be divided into three regions: (a) Levant or Greater Syria (Syria, Lebanon, Palestinian territories, and Jordan), (b) the Arabian Peninsula or Gulf Region (Iraq, Saudi Arabia, Yemen, Oman, Kuwait, UAE, Qatar, and Bahrain), and (c) Al Maghreb Al Arabi or North Africa (Morocco, Libya, Tunisia, Algeria, and Egypt). It is important to note that the Gulf Cooperation Council consists of all countries in the Arabian Peninsula with the exception of Yemen and Iraq (Nejem, 2016). Iran became separated from the Arab world due to religious and political reasons (Barakat, 1993). Iran did not adopt Arabic as its official language; the population is mostly Shia Muslim, a sect that conflicts with most Arab countries' religious structure, the majority of which are Sunni Muslim (Barakat, 1993; Nejem, 2016). Although there are differences between Sunni and Shia, the divide between the two practices grew significantly after the Iranian revolution in 1979 and the Iraq-Iran war, which ended in 1988 (Barakat, 1993).

The Arab Culture

A recent study by Nejem (2016) examined perceptions of expatriates who worked in Arab countries and discussed their experiences and observations of the Arab people. The participants in the Nejem (2016) study identified cultural values that they observed, such as strong family ties, among Arabs. These strong connections influence daily social interactions and expectations set by Arab communities. Other scholars have highlighted this importance and described the family as the core unit for the Arab society in its essential role of setting expectations for family members (Ayubi, 1991; Barakat, 1993; Zaharna, 1995).

Additional social norms, such as hospitability, valuing strong relationships, preservation of harmony, respecting elders, and protecting the family honor have been confirmed as important values in the Arab world (Ayubi, 1991; Barakat, 1993; Zaharna, 1995). First, Arabs are described as welcoming, hospitable, and valuing strong relationships. Second, Arabs prefer balance and harmony and avoid conflict; direct communication may not be an appropriate means of communicating. Third, Arabs respect their elders, valuing their experience and wisdom. Fourth, the majority of Arab countries, especially those in the Arabian Peninsula, are very conservative regarding dating outside of marriage and they limit social interactions between males and females. Fifth, same-sex relationships are prohibited, considered a sin in Islamic religion, and illegal in most Arab countries. Sixth, the terms *honor, respect*, and *reputation* are essential concepts that are highly valued in the Arab world. Many Arabs, both men and women, shape their behaviors to protect family honor and reputation (Ayubi, 1991; Barakat, 1993; Nejem, 2016; Zaharna, 1995).

Eastern Versus Western Cultures

As defined by many scholars, Eastern cultures are found in Africa, the Middle East, Asia, and Eastern Europe, while Western cultures are found in North America, Australia, and Western Europe (Minkov, 2011; Nejem, 2016; Nisbett, 2003; Storti, 1999). There are major cultural differences between the Eastern and Western worlds based on values, communication preferences, rituals and traditions, and worldviews (Hofstede, 2001; Minkov, 2011; Nejem, 2016; Neuliep, 2015; Nisbett, 2003; Storti, 1999; Ting-Toomey, 1999). Several important cultural concepts, such as individuality versus collectivism, the concept of time, the degree of directness, the importance of face, and locus of control represent major differences between Eastern and Western cultures (Hofstede, 2001; Minkov, 2011; Nejem, 2016; Neuliep, 2015; Nisbett, 2003; Storti, 1999; Ting-Toomey, 1999). People in Eastern cultures tend to focus on collectivism rather than individuality, are flexible with time, are not direct in their communications, place value in preserving harmony and saving face, and perceive that they have limited control over their lives. In contrast, people in Western cultures espouse individuality, value time and consider it limited, are direct in communications, value telling the truth, are not afraid to face conflict, and perceive that they have full control over their lives (Hofstede, 2001; Minkov, 2011; Nejem, 2016; Neuliep, 2015; Nisbett, 2003; Storti, 1999; Ting-Toomey, 1999).

Eastern cultures tend to have more structured norms, with strict rules and norms that are not easily changed (Minkov, 2011; Ting-Toomey, 1999). Western cultures can be described as more loose and flexible regarding social norms. Individuals can deviate from typical norms to experiment, thereby actively influencing change in their own cultural norms. Further, Eastern cultures tend not to encourage risk taking in career change, while Western cultures promote risk

taking in order to advance professionally (Minkov, 2011; Ting-Toomey, 1999).

Nonverbal communications are different between Eastern and Western cultures, as well (Neuliep, 2015; Storti, 1999). For example, Western cultures promote establishment of personal space, display of affection, direct eye contact, and firm hand shaking as a sign of confidence. Eastern cultures promote close personal proximity, touch as a sign of friendship and care (among those of the same sex), discourage direct eye contact, and discourage public display of affection (Nejem, 2016; Neuliep, 2015; Storti, 1999).

Student affairs practitioners who work in the Arab world can benefit from understanding the Arab culture and the social norms that are espoused by their students and enforced in their local community. The role of religion and culture can assist in explaining certain type of behaviors (Nejem, 2016). The importance of family and the role it plays in the student life can explain how family members may affect student involvement on campus. Understanding the differences between the Eastern and Western cultures can assist student affairs practitioners regarding how to alter their behaviors to be appropriate in their new environment.

Communication Challenges

Many barriers to the communication process can cause cultural clashes (Martin & Nakayama, 2010; Zweifel, 2003, 2013). These barriers include cultural biases, lack of awareness of cultural and language differences, maintenance of the same practices, failure to invest in establishing relationships, and ethnocentrism. One of the important barriers arises from cultural biases in the form of assumptions that many make about others based on accent or a different appearance; these can cause many problems in communications across cultures (Martin & Nakayama, 2010; Zweifel, 2003, 2013).

Cultural and language differences constitute another important barrier to cultural understanding (Martin & Nakayama, 2010; Zweifel, 2003, 2013). Assuming that others follow one's own behaviors or speak the same language can lead to cultural misunderstanding and possibly to cultural clashes. For example, silence is viewed differently in different cultures. In the United States, most people are uncomfortable with long periods of silence, while Native Americans regard people who remain silent positively. Similarly, in Asian cultures, silence is viewed as a sign of respect as it allows for reflection on what has been communicated, verbally or nonverbally (Martin & Nakayama, 2010).

It can be a cultural challenge to translate from one language to another (Martin & Nakayama, 2010; Zweifel, 2003, 2013). Language is more than a simple tool of communication. Language conveys unique cultural meanings associated with specific words or phrases. For example, phrases that are intended to express a particular concept often do not convey the intended meaning when translated to another language (Martin & Nakayama, 2010; Zweifel, 2003, 2013).

Several recommendations have been offered to avoid cultural misunderstanding (Martin & Nakayama, 2010; Zweifel, 2003, 2013). One of the key strategies is to listen actively for verbal and nonverbal clues. Contrary to popular belief, the majority of communication during a conversation is based on nonverbal actions, such as body language (Martin & Nakayama, 2010). This can present a challenge to someone who is not familiar with these nonverbal clues while communicating actively with someone from another culture (Martin & Nakayama, 2010; Zweifel, 2003, 2013).

Cultural Challenges for IBCs

Some Western IBCs have been accused of lack of respect for the cultural values of their non-Western host countries and instead have been perceived as trying to impose their own cultural values and beliefs on host country students through the policies and practices of their educational system (Shams & Huisman, 2012). This concept has been described as a form of cultural imperialism or a new form of colonization (Shams & Huisman, 2012). The majority of the Arab countries have suffered from colonizer actions and are afraid that IBCs can serve as a new form of colonization and exploitation of the developing countries that host IBCs (Altbach et al., 2009; Memmi, 1965; Shams & Huisman, 2012; Stryker et al., 2016).

Cultural clashes can place leaders of IBCs in an ethical dilemma (Shams & Huisman, 2012). One choice is for the IBC to standardize curriculum, staffing, and research initiatives to ensure that all students, at home and abroad, receive the same experience. The alternative is to adapt to the host country's environment and respond to local student and community needs. Researchers have concluded that IBCs must balance the two factors and address them simultaneously (Altbach et al., 2009; Farrugia & Lane, 2013; Shams & Huisman, 2012; Stryker et al., 2016). Any IBC that fails to achieve this balance could be accused of cultural imperialism, or the image and legitimacy of the institution could be jeopardized due to a possible claim that the IBC is not enforcing equal standards for all students (Farrugia & Lane, 2013; Shams & Huisman, 2012).

A similar situation was observed in the corporate world upon the opening of EuroDisney (Zweifel, 2003, 2013) in Paris, France, in 1992. The reaction by French consumers was extremely negative, and many boycotted the park. Zweifel (2003, 2013) noted that Disney failed to recognize cultural differences when they created the overall experience, modeling it exactly after the parks in the United States. For example, EuroDisney did not serve wine or local food, an absence that many French people did not appreciate. In addition, the hiring practices of EuroDisney did not encourage local French people to apply to work in the new theme park. Disney was able to fix this problem by

54

hiring a French president for EuroDisney. The new president made changes to adapt to the local culture in park services and structure that restored a positive image of EuroDisney and met the company's profitability goals (Zweifel, 2003, 2013).

This example can serve as a helpful case study for the IBCs to consider when creating new structure and philosophy in the host country. The example also explains that, even within Western societies, such as Europe, cultural differences cannot be ignored; local customs and traditions must be considered for an implementation process to be successful.

Cultural Challenges in Education City

The State of Qatar established Education City in 1995 to become the host of six American universities: Texas A&M, Northwestern, Georgetown, Carnegie Mellon, Weill Cornell, and Virginia Commonwealth (Clark, 2013; Vora, 2014). Reviewing the relevant literature, two studies were found addressing the impact of local culture on faculty and students working in and attending IBCs in Education City, Qatar (Bakken, 2013; Jauregui, 2013).

Jauregui (2013) conducted a study in Education City to explore the impact of cross-cultural training on teaching conducted by expatriate faculty who had been recruited from Western countries. After interviewing Western faculty, Jauregui (2013) concluded that formal and informal cross-cultural training taking place in the State of Qatar aided in development of the faculty's intercultural competencies and enhanced their cultural understanding of their new environment. After training, the faculty adapted the classroom environment to tailor the educational experience to meet the needs of students (Jauregui, 2013). Jauregui (2013) acknowledged cultural differences in Education City between the Western faculty, who were mainly recruited from the United States, and the students, who were mostly from Qatar or the Arab world. Jauregui made recommendations, based on his findings, to develop faculty intercultural competencies. One of his recommendations was to establish a mentoring program in which veteran faculty members would assist new faculty members to adapt to their new environment. In addition, he recommended that the universities in Education City re-examine their formal and informal training and provide more in-depth training that did not focus on cultural stereotypes or superficial facts. Formal or informal interactions and training should be structured and offered frequently to make a meaningful impact (Jauregui, 2013). This position was confirmed by Deardorff's (2006, 2012, 2016a, 2016b) research and framework that addressed development of intercultural competencies.

Bakken (2013) studied native Qatari students' relationships with the Western IBCs in Education City, Qatar. He examined the motivation behind student decisions to join Western IBCs. He concluded that the family played a significant role in influencing the student's decision to join the IBCs and not leave his/her home country because the parents wanted to have their children learning in an environment that respects the local Arab culture and the Islamic religion (Bakken, 2013).

Bakken (2013) also explored student participation in the classroom and activities outside the classroom. Students were seen to be struggling to adapt to their new learning environment, especially regarding communications and working in mixed-gender groups. In addition, students expressed hesitation to participate in extracurricular activities because they viewed those activities as conflicting with or opposed to their culture and/or religious beliefs. For example, the IBC staff who organized social activities promoted mixed-gender interactions even though the practice did not conform to their students' cultural and religious expectations (Bakken, 2013). These findings suggest an absence of sufficient intercultural training of student affairs practitioners who work at IBCs, especially in the Gulf region (Berardo & Deardorff, 2012; Deardorff, 2016a; Jauregui, 2013; Stryker et al., 2016).

The studies reported by Jauregui (2013) and Bakken (2013) confirm the cultural challenges that face faculty and students in IBCs. Faculty were able to enhance their classroom environment after receiving formal and informal intercultural training; student affairs practitioners can benefit similarly from such training to enhance their role in student growth and development through activities and programming offered in and out of the classroom (Deardorff, 2016a; der Heyde, 2016; Jauregui, 2013; Magolda, 2003; Manning et al., 2014). However, research is needed to measure the impact of such interventions on Western student affairs practitioners who work at IBCs across the world.

Theoretical Framework

Deardorff's model of intercultural competence is considered to be one of most influential research-based frameworks in the intercultural training field and is frequently used in higher education settings (Arasaratnam-Smith, 2017; Deardorff, 2012, 2016a, 2016b; Moscaritolo et al., 2016; Stryker et al., 2016).

Before discussing the model, it is important to highlight some of the myths mentioned by Deardorff (2016b) when discussing intercultural competence. Unfortunately, many student affairs practitioners believe in these myths, which can challenge the creation of an impactful intercultural experience for students. Deardorff (2016b) countered the myths with the following positions. First, students do not become culturally competent just because they study abroad. Second, students do not learn from each other just because they stay in the same room to attend a social and a cultural event. Third, intercultural competence does not happen simply by attending a course or a training session. Fourth, understanding cultural differences does not necessarily lead to developing competence. Fifth, speaking a foreign language does not necessarily lead to competence in that culture. The model of intercultural competence addresses how intercultural competence is developed and assists in exposing myths about intercultural competence (Deardorff, 2016b). Deardorff's (2012, 2016a, 2016b) framework can present student affairs practitioners with a new challenge to develop their own intercultural skills and to create a meaningful and impactful intercultural learning experiences for students.

Deardorff's (2012) model consists of five stages: attitudes, knowledge, skills, internal outcomes, and external outcomes. The critical attitudes stage promotes curiosity and openness to explore differences, with the understanding that people may have to go beyond their comfort zone to start the process of developing intercultural competence. In the second stage, knowledge, people develop self-awareness of their own culture and connect it with their worldview so that they can understand others' worldviews from others' perspectives. In the skills stage, people develop needed skills to conduct observations, processing and reflecting on what they have observed. Deardorff (2006) indicated that it is possible to move to the final stage (external outcomes) once the knowledge and skills stages have been navigated successfully. However, they may not develop sufficient competence to complete the full cycle (Deardorff, 2006).

58

The next two stages illustrate the move from interpersonal development (individually focused) to group interactions (Deardorff, 2006, 2012). In the internal outcomes stage, people become flexible and adaptable to the "ethnorelative perspective." In other words, they become comfortable with other environments and traditions by adapting their behaviors and judgments to diverse intercultural settings (Deardorff, 2006, 2012). The final stage, external outcomes, happens when people are in full control of their behaviors and communication skills and use them in an appropriate and effective manner in intercultural situations (Deardorff, 2012).

Deardorff (2012) highlighted key points regarding her model, which was developed in 2006. First, the framework indicates that intercultural competence is a continuous, lifelong process that must be sought intentionally to achieve positive outcomes. Second, the attitudes stage is considered as the catalyst to the intercultural competence cycle. People will not seek the intercultural competence process until they have developed an openness or curiosity to understand other cultures. Third, it should be noted that this is a U.S.-based framework because participants in the study were mostly from the United States. Deardorff (2012) questioned the adaptability and relevance of such framework when applied outside the United States.

Deardorff's framework presents many implications for student affairs practitioners, especially in an international setting (Arasaratnam-Smith, 2017; Deardorff, 2012, 2016a, 2016b). Many of the programs and services offered by student affairs practitioners do not apply any frameworks, particularly regarding intercultural competence (Deardorff, 2012, 2016a, 2016b). Deardorff (2012, 2016a, 2016b) challenged student affairs practitioners to integrate the model of intercultural competence in their daily work to assist students in developing intercultural competence. For example, Deardorff encouraged student affairs practitioners to develop cultural programs that promote students' curiosity and openness as a catalyst to start the intercultural competence development process. Deardorff also urged student affairs practitioners to become intentional in their services and programs to go beyond the superficial level of cultural holidays, international food festivals, and other cultural traditions. Alternatively, Deardorff (2012, 2016a, 2016b) proposed intentional infusion of the theoretical framework into programming and services that are planned by student affairs practitioners and offered to students.

Deardorff (2012, 2016a, 2016b) cautioned student affairs practitioners about approaching the solution to developing intercultural competence through offering certificate programs to students. Student affairs practitioners should be aware of the negative implications of such programs: false assumptions that students will develop intercultural competence, not utilizing a solid framework in the learning process and not implementing an assessment process to measure student learning and advancement through the process (Deardorff, 2012, 2016a, 2016b).

Deardorff's (2006) framework guided this study and provided an understanding of how student affairs practitioners implement services and programs in an international setting. As Jauregui (2013) concluded, Western faculty can enhance the teaching environment in the classroom by encouraging intercultural competence. This framework also assisted in understanding why student affairs practitioners working in IBCs in the Middle East made decisions to tailor their services and programs based on their intercultural development (Deardorff, 2012, 2016a, 2016b; Jauregui, 2013). Although this study was not designed to measure the intercultural competence level of the participants, Deardorff's (2006) framework suggested that enhancing student affairs practitioners' competence may lead to improvement in delivering relevant leadership programs for students (Deardorff, 2012, 2016a, 2016b).

Chapter 3

Research Design and Methodology

The American IBCs were widely recognized in the Gulf region as many U.S.based campuses opened in the region (Lane, 2011). Most notably, six American IBCs were opened between 1998 and 2008 in Education City with collaboration and support from Qatar Foundation, the host entity in the State of Qatar. Since these Western-based higher education institutions were established in Qatar (a non-Western country), it was important to understand the role, if any, of local culture in shaping practices of the student affairs practitioners.

This study explored how student affairs practitioners view their implementation of Western student affairs practices within the local context in the State of Qatar. The study also explored how Western student affairs practices were altered to meet the needs of local students. The examination of student affairs practices addresses (a) how these practices were implemented in IBCs, (b) the role of local culture in IBCs, and (c) best practices that were recommended by the participating student affairs practitioners.

This chapter describes the methods and processes that were followed in the study. The chapter includes the following sections: the research site, the research design, the research questions, the research participants, the data collection and analysis procedures, research credibility and validity threats, and ethical considerations.

The Research Site

The State of Qatar is a small country, approximately half the size of New Jersey, with a total population of 2,450,285 (Qatar Ministry of Development Planning and Statistics [QMDPS], 2018). The local Qatari population size is 306,187, which makes

expatriates the largest group in Qatar (Gulf Research Center, 2017). The Qatari government launched its Qatar National Vision 2030 program (called QNV 2030) in 2008; one of the important pillars of this plan is human development (QMDPS, 2008). The human development pillar is designed to promote progress by providing a worldclass education system to meet the needs of its citizens (QMDPS, 2008). The IBCs operating in the State of Qatar assist in reaching the goals established in QNV 2030 (QMDPS, 2008).

The State of Qatar was selected as the research site for this study based on many important factors: (a) Qatar is considered one of the major educational hubs in the Gulf Region, (b) Qatar Foundation provides generous support to IBCs to sustain their operations in the country, (c) Qatar Foundation hosts a unique cluster of six American IBCs in one location (Education City), and (d) I have access to these six IBCs as I work in the State of Qatar.

The student affairs staff members at the six IBCs in Education City were invited to participate in the study per the criterion established for participation. At the time of the study, approximately 15 student affairs staff members working in Education City met these criteria. Since Education City is a small close-knit community, the names of the participating university sites and staff members were kept confidential and pseudonyms were used. This encouraged participants to feel comfortable to participate and to share their knowledge and experiences.

The Research Design

Researchers often select their approach by considering procedures of inquiry, data collection, data analysis, and data interpretation (Creswell, 2014). However, the most

critical component in the decision process relates to the nature of the phenomenon under study. For example, qualitative research is designed to explore difficult and complex social problems encountered by individuals or groups (Creswell, 2014; Merriam, 1998). In contrast, quantitative research is intended to test a specific hypothesis or theory through examination of defined variables that can be measured using certain statistical tools (Creswell, 2014; Hammersley, 2013; Merriam, 1998). This study utilized a qualitative approach because the goal was to understand student affairs practitioners' perspectives on the topic of study. Furthermore, this study employed the social constructivism worldview to explore the meaning created by the student affairs practitioners through implementation of their Western practices in Qatar.

The Constructivist Worldview

Researchers consider their study approach by examining three important areas: their philosophical worldview, research design, and research methods (Creswell, 2014). The terms *worldview* or *paradigm* can be defined as "a basic set of beliefs that guide action" (Guba, 1990, p. 17). Although the researcher's personal philosophical beliefs and assumptions may not appear to be important, they can in most cases influence their decision in selecting a research methodology (Creswell, 2014; Merriam, 1998). Therefore, I discuss my philosophical worldview, the plan to adopting it for this study, and how my worldview guided my approach to the study (Creswell, 2014; Merriam, 1998).

In the social constructivism approach, people are interested in learning about their environment (Creswell, 2014). People generate meaning for their interactions and behavior within their specific communities that may be quite different from meaning ascribed by others who live in other cultures or nations. Thus, I focused on participants' responses and perceptions to understand the phenomenon under study. The participants played an essential role in communicating the meaning of working in student affairs in the local community in Qatar (Creswell, 2014).

In this study, understanding social and cultural norms were very important, especially as they guide people in establishing expectations about their way of life (Creswell, 2014). Often, these norms and values are passed from one generation to the next through history and education. However, these values and norms are also subject to change and development as people interact and negotiate the cultural meanings for these behaviors over time. Thus, understanding the local context was essential in this study because that context has certain values and cultural norms that differ from American values and norms on which student affairs practices are based. In addition, my own understanding of the differences assisted in my role as the researcher to comprehend the meaning of the experiences taking place within the local context (Creswell, 2014).

As a constructivist researcher, I acknowledge two important assumptions that guided this study (Creswell, 2014; Crotty, 1998). First, people create meaning by interacting in their environments and by reflecting on those experiences. For example, I posed general unstructured questions to encourage participants to share their perspectives and to feel comfortable to reflect on those experiences. Second, people utilize meanings that are already established and recognized within their communities. I invested time to study and understand the local context of the study participants in their work (Creswell, 2014; Crotty, 1998).

This study followed a social constructivist approach due to the nature of the study. One of the important goals for the study was to understand how Western student affairs practices were implemented in the American IBCs in the State of Qatar. I examined the perceptions and insights from participants to identify their understanding of the services that they deliver to local students in the State of Qatar. Furthermore, the social constructivist approach aided in understanding how the local context shapes student affairs practices, which may assist in determining the most successful practices to complement the local cultural and social conditions in the State of Qatar. Social constructivism is typically associated with qualitative methodology; this study utilized a qualitative methods approach (Creswell, 2014; Merriam, 1998).

Positionality Statement

Researchers can bring their own worldviews, experience, and work ethic to a research project (Creswell, 2014; Flyvbjerg, 2006, 2011; Merriam, 1998). Therefore, qualitative researchers should identify and acknowledge any biases or personal views that they bring to their research (Creswell, 2014; Flyvbjerg, 2006, 2011; Merriam, 1998). Since I have been working professionally in student affairs for more than 12 years, I bring a considerable amount of experience to the project. However, in this research, I sought not to allow my personal experiences to interfere with or influence my role as a researcher. As a qualitative researcher, I did not hold preconceived opinions about the phenomenon that I was exploring. Rather, I used my work experiences to reflect on the participants' feedback as it was relevant and applicable in the situation.

I believe in pursuing my own subjectivity by acknowledging and reflecting on my emotions and any biases that may arise while conducting research (Peshkin, 1988). One of the important steps to accomplish this task was to keep a personal journal to account for my biases. I can identify these biases not only by acknowledging the participants who shared my perspectives but also by recognizing others who presented a different or new perspective (Peshkin, 1988). The personal journal helped me to reflect on the interviews and identify any personal feelings or connections that could limit my ability to be neutral and subjective when reviewing and processing the collected data (Peshkin, 1988; Stringer, 2014; Yin, 2016).

My main motivation to conduct this research came from observations during my transition to work in Education City, Qatar. I moved to Qatar in 2012 after Texas A&M University recruited me from New York. I faced a few challenges during my transition; however, I felt that my transition went much more smoothly than that of my U.S. colleagues because of my ability to speak the local language (Arabic) and my knowledge and understanding of the local culture, since I grew up in the Middle East region. As a result, I saw the need to conduct research on this phenomenon because of the rapid growth and expansion of American International Branch campuses across the world. Although there are strong established standards in the United States, there are currently no recognized international standards and accepted practices for student affairs practitioners to follow when working overseas. This is mainly due to a lack of research in the student affairs and student development areas in general outside the United States. My commitment to my profession and advancing it internationally has made me passionate about conducting this research study.

As a researcher, I sought to conduct a study that would deliver the facts, be transparent, and be neutral to all parties involved. I did not anticipate or support any specific outcome and hoped that the study results would fill the gap and produce a wellrounded study to benefit all student affairs practitioners working overseas and ultimately benefit local students who are participating in this unique U.S. educational experience overseas.

Qualitative Methodology

The world is becoming more complex and diverse every day (Martin & Nakayama, 2010; Zweifel, 2013). Rapid growth and development can present a challenge in understanding the cultural changes in both small and large communities (Covarrubias, 2002; Drew, 2001; Flick, 2009). As a result, there is a growing need for using qualitative methodology to study social problems (Flick, 2009; Merriam, 1998; Yin, 2016). Qualitative research contributes to creating new knowledge that enhances understanding of the social world and how it is constructed (Creswell, 2014; Crotty, 1998; Merriam, 1998; Yin, 2016). For these reasons, qualitative methodology was selected for this study to understand student affairs practitioners' perspectives regarding implementation of Western student affairs practices in the local context in the State of Qatar.

Writers and historians have for centuries utilized observations and interviews in their work; however, these methods were not formally adopted as part of the qualitative methodology until the end of the 19th century (Clifford, 1983; Taylor, Bogdan, & DeVault, 2016; Wax, 1971). Researchers typically select their methodology by defining the approach that they wish to pursue to explore an existing problem (Taylor et al., 2016). Qualitative methodology naturally produces descriptive data rather than numerical data, as are used in quantitative methodology (Creswell, 2014; Merriam, 1998). The data generated in this study came directly from the participants' spoken or written words through the interviews and the review of the material culture items provided by the participants (Bagnoli, 2009; Creswell, 2014; Hammersley, 2013; Merriam, 1998; Taylor et al., 2016). This qualitative research study is designed to understand and interpret the meaning of the student affairs practitioners' experiences and how they construct the meaning of those experiences within their social reality (Hammersley, 2013; Merriam, 1998; Yin, 2016).

One of the core values in qualitative research is its ability to recognize the importance of the meaning that people apply to their experiences, pay attention to the way that people think or act in real-life situations, promote the study of people in their local cultural context and natural settings, encourage researchers to explore a wide range of perspectives, and enable researchers to keep an open mind to gain information and insight about the participants and their local context (Hammersley, 2013; Taylor et al., 2016; Tracy, 2013; Merriam, 1998; Yin, 2016). In addition, qualitative methodology provides the researcher with flexibility in determining how data collection and analysis processes are conducted (Merriam, 1998). Thus, I have chosen qualitative methodology because of its adaptability to develop the needed structure for the study (Creswell, 2014; Hammersley, 2013; Merriam, 1998; Taylor et al., 2016; Yin, 2016).

There are additional advantages and strengths in selecting qualitative research methods (Creswell, 2014; Merriam, 1998; Yin, 2016). First, qualitative methodology promotes field work and encourages researchers to conduct inquiries in real-life situations (Creswell, 2014; Merriam, 1998). Second, as the researcher, I will play an essential role in directly collecting data from the participants rather than relying on tools or instruments (Creswell, 2014; Merriam, 1998). Third, I will utilize multiple sources of data, including interviews and a review process of material culture, as recommended by proponents of qualitative methodology (Creswell, 2014; Merriam, 1998; Yin, 2016).

On the other hand, qualitative research is often criticized for lacking scientific rigor, strong justification for its methods, or the potential negative impact of the researcher's bias (Merriam, 1998; Rolfe, 2006; Sandelowski, 1993). For example, some qualitative researchers have been criticized for relying solely on interviews as the main source of study data (Hammersley, 2013). There is a high risk in depending on only interviews alone because the researcher's objectivity in conducting interviews may come into question (Flick, 2009; Hammersley, 2013; Merriam, 1998). The qualitative researcher must have good communication skills in order to connect well with participants, listening actively and asking appropriate questions (Merriam, 1998). The researcher's ability to listen openly without prejudice or preconceived ideas can be a critical challenge in qualitative inquiry (Flick, 2009; Hammersley, 2013; Merriam, 1998; Yin, 2016).

Some of the concerns with qualitative research can be addressed by applying various strategies such as (a) acknowledging personal biases, (b) establishing and following ethical standards to avoid those biases, (c) ensuring accurate and clear record keeping to reflect objectivity in the decision-making process, (d) seeking participants' feedback and endorsement of the interview transcripts and initial findings, and (e) using multiple data sources to conduct triangulation for clear and comprehensive findings (Creswell, 2014; Flick, 2009; Long & Johnson, 2000; Merriam, 1998; Morse, Barrett, Mayan, Olson, & Spiers, 2002; Sandelowski, 1993; Slevin & Sines, 2002; Yin, 2016). I implemented many of these strategies to enhance this qualitative study. In addition, I

utilized a qualitative case study approach to support the social constructivist worldview that was described earlier.

Qualitative Case Study Approach

The case study approach, employed in this study, is one of the most popular designs used in social sciences and in qualitative methodology (Merriam, 1998; Yin, 2014). There are several definitions for the case study approach, which has caused confusion and misunderstanding among researchers (Merriam, 1998). For example, some scholars (e.g., Yin, 2014) have made a clear distinction between the case under study and the end product of the study; others (e.g., Merriam, 1998; Stake, 1995, 2005) have selected to focus on the case instead. The three cited scholars, Merriam, Stake, and Yin, are the primary contributors to the development of the case study approach. Each has a particular perspective on how to view, define, and structure the case study approach.

The major debate between these schools of thoughts is whether the case study should be classified as a qualitative research *design* or as an independent research *method* (Baxter & Jack, 2008; Flyvbjerg, 2006, 2011; George & Bennett, 2005; Gerring, 2004; Merriam, 1998; Simons, 2009; Stake, 1995, 2005; Verschuren, 2003; Yin, 2014). For example, some scholars consider the case study to be a research design within the qualitative methodology and define it as a *qualitative case study* (Merriam, 1998; Stake, 1995, 2005). In this perspective, researchers are expected to follow the qualitative methodology guidelines in collecting, analyzing, and validating data (Merriam, 1998; Stake, 1995, 2005). The second school of thought presents the case study as an *independent research method* with its own guidelines to be applied in either qualitative or quantitative studies (Yin, 2014). Yin (2014) presented a very structured approach to case study that could present challenges to researchers who like to adopt flexibility in approaching their case study design.

Another important difference in the approaches of these three scholars is their *epistemological beliefs* (Merriam, 1998; Stake, 1995, 2005; Yin, 2014). The term refers to the basic beliefs on how knowledge is created and how research should proceed to reach its findings (Yin, 2016). Epistemological beliefs have influenced how these three scholars developed the case study design and its guidelines for use in research (Merriam, 1998; Stake, 1995, 2005; Yin, 2014). While Merriam (1998) and Stake (1995, 2005) share a belief in the constructivist worldview, Yin (2014) follows a positivist view. The positivist worldview restricts the generation of knowledge to scientific methods, such as statistical procedures (Creswell, 2014; Yin, 2016). However, constructivism recognizes the important role of people in creating the meaning and social realities that cannot be measured simply by following specific scientific

procedures and methods (Creswell, 2014; Crotty, 1998; Yin, 2016).

For the purpose of this study, I adopted the perspective of Merriam (1998) and Stake (1995, 2005) in defining the case study approach as a qualitative research design intended to explore a social phenomenon within a single or small number of occurrences. This approach allowed flexibility and led to a holistic understanding of the various perspectives presented by the participants (Merriam, 1998; Stake, 1995, 2005). In addition, it was important not to be limited in design structure (Stake, 1995, 2005). It was difficult to predict the challenges that would be encountered during the study; thus, it was

important to maintain flexibility and adoptability to address these obstacles (Stake, 1995, 2005).

The qualitative case study approach was developed to explore a phenomenon in its real-life setting (Creswell, 2014; Merriam, 1998; Stake, 1995, 2005). The case can be considered as an individual (e.g., student, faculty, staff member) or a class, institution, or community (Merriam, 1998; Stake, 1995, 2005). The qualitative case study design is influenced by its environmental factors and provides the researcher tools to determine limits and context for the case (Flyvbjerg, 2011; Merriam, 1998; Stake, 1995, 2005). For example, Baxter and Rideout (2006) conducted a qualitative case study to explore decisions made by nursing students and how the environment shaped those decisions. The local context in which the decision process was taking place was an important factor in the study. As a result, the researchers studied the conditions in clinics and classrooms because students were developing skills to make decisions in those settings (Baxter & Rideout, 2006).

The qualitative case study approach can be classified in two types: a single case or a multiple-case study (Merriam, 1998). In the single-case type, the focus is on an individual phenomenon, defined as a *case*. In the multiple-case study, the focus is on investigating several qualitative case studies, each studied individually, followed by a comparison across the case studies to determine common themes or differences (Merriam, 1998). In this study, I will utilize the single-case type by focusing on student affairs practices as the phenomenon that forms the case (Merriam, 1998).

The single qualitative case study can be further categorized in terms of the concept of time to include the following types: "*retrospective, snapshot, and diachronic*"

(Thomas, 2011, p. 517). In the retrospective case study, the goal is to study a historical phenomenon that occurred in the past. The data collection process involves reviewing essential historical documents (Merriam, 1998; Thomas, 2011). In the snapshot case study, the case under study is explored during a specific period of time span (Thomas, 2011). For example, the study can involve a specific event taking place during a semester or a full academic year. In the diachronic case study, the events of focus occur over time, highlighting the growth and development of the case; this type shares characteristics with longitudinal studies (Thomas, 2011). For the purpose of this study, a snapshot case study approach was adopted to examine current student affairs practices, which seem to change over time (Thomas, 2011). In addition, cultural norms and values can change over time; thus, studying the past may not be essential in identifying current key practices of student affairs practitioner in response to their current students' needs (Creswell, 2014; Crotty, 1998).

The qualitative case study design provides an in-depth approach to inquiry by promoting collection of data from several sources, leading to triangulation (Merriam, 1998; Stake, 1995, 2005). The typical data collection methods include interviews, observations, review of artifacts and documents, and other techniques such as audio and video recordings (Creswell, 2014; Merriam, 1998; Rossman & Rallis, 2012; Stake, 1995, 2005; Stringer, 2014). Field notes and journals can be included in the data collection process (Stringer, 2014; Yin, 2016). Field notes and journals can include the researcher's own words to be used for further exploration or to discuss questions with members of a research team; they can aid in reflection and in generating new ideas (Stringer, 2014; Yin, 2016).

The qualitative case study approach has many advantages and benefits, such as providing tools to examine complex social phenomena, leading to a holistic and in-depth understanding of the phenomenon, and guiding reform of practices and policies (Merriam, 1998). In addition, case studies can help in exploring difficult cultural situations that involve complex human relationships and unclear context (Bloor & Wood, 2006; George & Bennett, 2005). On the other hand, case study has been criticized. For example, the approach can be expensive and extensive in time (Merriam, 1998). Also, in some cases it is difficult to generalize the findings (Flyvbjerg, 2006, 2011; Merriam, 1998). Researcher bias can influence findings based on preconceived beliefs or opinions, since the researcher plays an essential role in data collection and analysis (Flyvbjerg, 2006, 2011; Merriam, 1998).

Applying the Case Study Approach

The six American IBCs in Education City form a special community of schools that share many goals. In fact, many of the branch campuses work on shared projects, and students generally interact and identify as part of the Education City community (Vora, 2014). The contract between the Qatar Foundation and the six American IBCs contains many similar expectations, which makes it easy to form a bond (Anderson, 2016). For example, the student activities coordinators at the six American IBCs have established the Student Life Committee, which meets biweekly to discuss shared events and collaboration in Education City (personal communication, L. Tillman, International Programs Coordinator, September 18, 2017).

A qualitative case study approach was selected to explore this phenomenon within the real-life context of Education City, Qatar. The student affairs practitioners working in Education City were treated as a single case (Merriam, 1998; Stake, 1995, 2005). The qualitative case study approach was suitable for this study because the study was conducted in a tight-knit community where the student affair practitioners may operate independently but within the same unified and integrated system in Education City, Qatar (Merriam, 1998; Stake, 1995, 2005; Yin, 2016). Consideration of the cultural context was important because local cultural norms, values, and customs played an essential role in defining expectations for activities that were conducted by student activities professionals (Creswell, 2014; Crotty, 1998).

Multiple data sources were utilized, including in-depth semistructured interviews and a review of artifacts created by student affairs professionals who work or had worked in Education City (Creswell, 2013; Rossman & Rallis, 2012; Stringer, 2014; Yin, 2016). Artifacts included publicity materials for events, instructions for student clubs and organizations, and promotional materials created to engage students in activities that were provided by some participants during the interview process (Creswell, 2013; Rossman & Rallis, 2012; Stringer, 2014; Yin, 2016).

Research Questions

This study addressed the following key questions:

1. How do student activities professionals describe the role of local culture in shaping Western student affairs services implemented on their branch campuses?

2. In what ways do student activities professionals alter their Western practices at IBCs in Qatar?

3. What best practices can emerge from the study findings to guide student activities professionals who work at IBCs in the Middle East Region?

Research Participants

The concept of data saturation was introduced by Lincoln and Guba (1985); it refers to researchers encountering repetition of the data, common and consistent patterns, and no new ideas emerging (Creswell, 2014; Lincoln & Guba, 1985; Merriam, 1998). The goal of any qualitative study, especially in the case study approach, is to provide indepth understanding of the phenomenon, rather than having a large number of participants to represent populations or to generalize findings, as is done in quantitative research. A single case study can generate great insight, regardless of the size of the sample (Boddy, 2016; Creswell, 2002, 2012, 2014; Gentles, Charles, Ploeg, & McKibbon, 2015). Creswell (2012, 2014) recommended having a range of 2 to 20 participants in qualitative studies, based on the specific type of study and keeping in mind the point of data saturation. Therefore, my goal was to recruit a sufficient number of participants (10 to 15), based on availability, to reach data saturation. Furthermore, my goal was to recruit a diverse sample by inviting participants from all six American branch campuses located in Education City.

Based on qualitative research principles and the case study approach, participants were selected using two purposeful sampling strategies: criterion sampling and snowballchain sampling (Creswell, 2014; Kuper, Lingard, & Levinson, 2008; Maxwell, 2013; Patton, 2002; Rossman & Rallis, 2012; Stringer, 2014). The following criteria were established for participation: (a) actively engaged with students by organizing student activities, (b) having worked at one of the American IBCs or in Education City for at least a year, (c) having worked previously in the higher education profession in the United States, and (d) having completed education in the United States.

These criteria assisted in narrowing the selected participants in different functional areas of student affairs. First, I started by selecting student activities coordinators because they typically have the most interactions with students outside the classroom environment and they were potentially the most affected in considering the impact of culture on their activities. Second, I selected staff members who had worked for at least one year in Qatar because the goal of the study was to examine the impact of the local culture on student activities, and new staff would not be able to understand the local culture immediately. To do so requires time to adjust, observe, and learn about the new environment. Third, participants had prior work experience in the United States to be able to identify differences in student activities practices between the United States and the Middle East, especially in Qatar. Local student affairs staff members who had never worked in the United States might not be able to recognize those differences. Similarly, only student affairs staff members who had completed their education in the United States were selected because they had experienced the U.S. educational system and were familiar with the services that were offered through student affairs in the United States.

Participants were invited via an email. The specific criteria were listed in the email to help potential participants to determine whether they were eligible to participate in the study. A snowball-chain sampling technique was implemented by asking initial participants to recommend colleagues who could provide rich data for the study (Patton, 2002; Stringer, 2014). This method was especially helpful in locating and inviting former employees who had recently left and returned to the United States.

Data Collection Procedures

Interviews served as the primary method for collecting the data needed for the study. Interviews were conducted in the United States and Qatar after approval by the Rowan University Institutional Review Board (IRB) and a local IRB (provided by Weill Cornell Medicine-Qatar). The semistructured interview structure was selected to provide flexibility during the interviewing process (Creswell, 2012; Rubin & Rubin, 2012; Stringer, 2014). Semistructured interviews are carefully designed to elicit interviewees' perspectives and opinions about a particular phenomenon, instead of limiting their ability to express their thoughts through a more structured approach. In a semistructured interview, the interviewer first asks the primary question, then utilizes probes and followup questions to gain in-depth information from participants (Creswell, 2012; Rubin & Rubin, 2012; Stringer, 2014). Probes can be classified into two types: verbal and nonverbal (Rubin & Rubin, 2012). Verbal probes are direct, such as, "Can you tell me more about this? Can you give me an example?" Nonverbal probes can take the form of the researcher giving the participant more time to answer a question or showing physical expressions to encourage the participant to continue to speak (Rubin & Rubin, 2012). I used both verbal and nonverbal probing strategies in the interviews.

Interviews ranged from 60 to 90 minutes in length, depending on the information discussed by the participant. The interviews started with a short introduction and broad questions (Appendix A), followed by appropriate probing to ensure in-depth understanding of the participant's perspective. All interviews were recorded and transcribed (Rubin & Rubin, 2012).

In addition to interviews, available artifacts from the branch campuses provided by the participants (such as events posters, newsletters, policies and regulations) and related to student activities were reviewed (Creswell, 2014; Rubin & Rubin, 2012). These artifacts aided in the analysis process and served as an important example of how student affairs practitioners alter practices to enhance student engagement in their programs and activities.

Data Analysis

A deductive analysis approach was applied to guide the process of organizing the collected data (Yin, 2014). An interpretive analysis technique was implemented through three stages: deconstruction, interpretation, and reconstruction (Miles, Huberman, & Saldaña, 2014). In the deconstruction stage, the raw data were assigned codes and categories to describe the content. In the interpretation stage, the focus was on understanding and making sense of the coded data by comparing codes and categories to identify similarities and differences. In the reconstruction stage, central themes emerged based on the interpretation of the coded and categorized data (Creswell, 2014; Miles et al., 2014; Rubin & Rubin, 2012; Yin, 2014).

The data analysis process began with review of the transcriptions of the interviews in order to identify actual words exchanged between participants and researcher (Creswell, 2014; Rubin & Rubin, 2012). Once the transcription process was complete, I organized all data by placing them into their original sources (interviews, personal notes, memos, and material culture). Then, I reviewed the data to examine the general meanings. This included identifying common and different ideas, coding the data, and organizing the coded data into small manageable categories for further analysis. This

process allowed categories to emerge naturally from the collected data (Creswell, 2014; Hsieh & Shannon, 2005; Patton, 2002 Rubin & Rubin, 2012; Yin, 2014).

I then wrote analytical memos to record the preliminary findings and to develop a codebook to be utilized during the analysis stage (Saldaña, 2009). The InVivo method was implemented during the coding process to allow use of the participants' own words and expertise to understand the phenomenon under study (Creswell, 2014; Saldaña, 2009). The raw data, organized by identifying codes and categories, generated overarching themes (Anfara, Brown, & Mangione, 2002; Creswell, 2014; Hsieh & Shannon, 2005; Mills, 2003; Rubin & Rubin, 2012; Saldaña, 2009). These themes addressed the main research questions, including how the student affairs practitioners viewed their work within the local cultural context in Qatar (Merriam, 1998).

Trustworthiness and Validity Threats

Trustworthiness for any research study is considered one of the essential conditions (Lincoln & Guba, 1985). Trustworthiness can be achieved through several strategies, such as credibility, transferability, dependability, and confirmability (Creswell, 2013; Lincoln & Guba, 1985; Stringer, 2014). Some of these strategies were employed to ensure the trustworthiness and validity of the study. First, a clear interview protocol was established, including the process for reviewing the material culture, in a clear connection to the research questions (Anfara et al., 2002; Creswell, 2012, 2013; Toma, 2006). This protocol and questions were pilot tested with non-student affairs staff to ensure their effectiveness and to make modifications as needed (Creswell, 2012, 2013). Second, member checking was used after transcribing the interviews by sharing the interview transcripts with the participants to confirm the data and to allow them to suggest

corrections or additional insights (Creswell, 2014; Stringer, 2014; Toma, 2006). Finally, the data from all sources, including the material culture review, were triangulated (Creswell, 2013; Toma, 2006).

Ethical Considerations

I work for Weill Cornell Medicine-Qatar (WCM-Q), one of the American branch campuses selected for this study. However, I do not currently serve in any supervisory role in that institution's student affairs department. To avoid any conflict of interest, I asked the assistant dean for student affairs to send the invitation to my colleagues at WCM-Q to ensure that their participation was done on a voluntary basis. In addition, I emphasized to all participants that identifying information would be kept confidential. The collected data were stored in a secure server and will be destroyed after the study is completed. Pseudonyms for the selected institutions and participants were used in reporting the study in order to protect the identifies of all participants.

Chapter 4

Results and Findings

This chapter presents the findings of the qualitative case study that was designed to explore the perceptions and understanding of student affairs professionals who are implementing Western student affairs practices within the local context in the State of Qatar. The study also explored possible best practices developed by these student affairs practitioners through their work in the IBCs in Qatar.

The chapter begins with a brief description of the participants, the themes and sub-themes that emerged from the study in connection with each research question, and examples of the participants' own words in support of each theme. Three research questions guided the study:

1. How do student affairs professionals alter their Western practices at IBCs in Qatar?

2. How do student affairs professionals describe the role of local culture in shaping the Western student affairs services implemented at IBCs in Qatar?

3. What best practices can emerge from the study findings to guide student affairs professionals working at IBCs in the Gulf Region?

Research Participants

I interviewed 14 individuals from September 19 through November 17, 2019 in Qatar and in the United States. The interviews ranged in length from 30 to 70 minutes. The same protocol, consisting of nine questions, was asked in all interviews, as described in Chapter 3. In addition, three participants provided the following material culture items: posters for events, policies for clubs and organizations, and promotional pamphlets for leadership programs offered to their students.

All participants met the criteria: (a) were actively engaged with students by organizing student activities, (b) had worked at one of the American branch campuses for at least a year, (c) previously worked in the United States, and (d) completed their higher education in the United States. Each participant signed a consent form. All participants agreed to have the interview audio recorded to improve the data collection process. Member checking was conducted to verify the collected data (Creswell, 2014; Stringer, 2014; Toma, 2006).

Background of the participants. Many student affairs practitioners who work in Education City are from the United States; others have been hired locally. Some, but not all, have educational degrees and credentials from the United States. For the purpose of this study, only those student affairs practitioners who had been directly recruited by the institutions in Education City were interviewed, per the criteria established for participation.

In terms of current employment status, 4 participants were former employees in Education City and 10 were currently employed in Education City. There were six males and eight females. The participants worked at five American IBCs in Education City, Qatar. They served in various positions in students affairs, ranging from entry-level positions, such as program and events coordinators, to mid-level positions, such as assistant directors, and senior-level directors. Their length of employment ranged from 2 years to 12 years.

Lucas. Lucas worked in Education City for 9 years but had left recently to pursue new opportunities. He had been recruited to work in Education City from the United

States. In his role, he worked with advising the student government and student clubs and organizations, and he coordinated major student events, including orientation and the service-learning initiatives. Lucas cited major differences between his previous large campus in the United States and the IBC. He described the differences as "night and day," especially because of the small number of students and the quality of their interactions in the small IBC community. Lucas recalled his experience, highlighting changes that he had had to make and the importance of being respectful of the local culture and host country. He stated, "We are just kind of trying to be culturally mindful in general." He emphasized the importance of establishing strong relationships and encouraged student affairs practitioners to adjust their practices to meet their students' needs.

William. William served for 9 years in Education City but then left the position and currently works in the United States. His responsibilities in Education City included coordinating major events for the department, athletic programs, and the student conduct hearing process. He was recruited directly from the United States. William started his interview by discussing the high school preparations for students in the United States and students at his IBC. He concluded that his students at the IBC were less prepared to be involved in extracurricular activities and programs than were their counterparts in the United States. "Basically, they are much less experienced than what it had been my experience in working with students in U.S. schools." William highlighted the essential role of the family in his students' lives. For example, he stated, "There were a lot of expectations for students . . . to contribute to their families." William concluded by offering recommendations to student affairs professionals who are considering joining IBCs in Education City.

Sophia. Sophia has been employed in Education City for 12 years. She started to work in student affairs as the lead person, coordinating the design of their student affairs program. She is currently working in Education City in a different role. Sophia recalled early lessons that they learned together and noted that it was difficult to establish programs and events for her students. "Figuring out how to set up the students' activities was very challenging . . . students did not have a background in college activities and we did not have a background in activities in this culture." She highlighted cultural differences and explained how they influenced student behavior. "They [Education City students] are far more group oriented, less individualistic than the kids in the U.S." Similar to the other participants, Sophia confirmed the important role of the family in influencing student experiences. "Family constraints are huge . . . like sports and girls, you can really see the conflict." She mentioned this as an example of how families limited their daughters' participation in the college athletics programs.

Olivia. Olivia worked in Education City for 2 years, then left and is currently working in the United States. In her role in Education City, she oversaw the international service-learning opportunities and assisted international students with the immigration process in Qatar. Similar to Lucas, Olivia discussed the transition from a very large campus to a small campus. She expressed appreciation for the opportunity to work with a different student population. "There were 30 different nationalities represented at the institution. . . . I appreciated the opportunity to learn from students, a population that I haven't worked with previously." Olivia cited cultural differences and emphasized the important role of the family in students' lives. As an example, she recalled that parents

had accompanied their daughters on international travel, something than Olivia had never experienced before. Throughout the interview, Olivia focused on the need to understand the local culture and context in order to be efficient as a student affairs practitioner working in an IBC. She stressed that student affairs practitioners cannot simply apply the practices that they used in the United States. "It takes a lot of time to learn the local culture and context to understand our job, to build those relationships."

James. James worked in Education City for 3 years, then left and is currently working in the United States. In his role in Education City, he managed the leadership development programs and coordinated major events across Education City. He recalled that he had noticed differences in the cultural context immediately after he started. As did other participants, he mentioned the family role in restricting student participation in some of the leadership opportunities, noting, "Maybe that wasn't something accepted by their families or in their cultures." James mentioned that many students did not fully understand or value the involvement opportunities outside the classroom. He attributed this to a lack of exposure to similar opportunities during high school. At the end of his interview, James mentioned the importance of intercultural training to improve student affairs competence skills. He recommended that IBCs provide such training, concluding, "Yes, 100%, I think that it should be a part of onboarding."

Logan. Logan has worked for 6 years in Education City and is currently serving as the advisor and coordinator for international programs. He also coordinates the athletic programs for his institution. He was recruited directly from the United States to work in Qatar. Logan started by comparing student populations between his IBC and the United States, highlighting cultural differences. For example, he cited the concept of time, which was a challenge for him in adjusting to his new setting. "The most difficult to navigate and to adjust to is the concept of time." Logan also discussed the role of the family in the students' lives, the buy-in that is needed from students, and the importance of preparing student affairs staff members before they start their work at an IBC. He recommended, "Prepare people before they come out here, so they know what they are getting into and they know what is appropriate [within the local context]."

Jack. Jack has worked in Education City for 8 years. In his leadership role, he coordinates educational programs and manages student events conducted in Education City. He was recruited from the United States. Jack expressed early in the interview the challenge of unifying students who come from across the globe. However, he indicated that most of the students share some common things, such as religion and cultural background. "I think one unifying factor is that many of the students are Muslim . . . even from the Gulf region, so culturally there are some shared elements there." Similar to other participants, Jack highlighted the challenge of finding appropriate times to offer events and activities. He cited specific cultural differences and made suggestions on how to navigate them. For example, he discussed gender intermixing at events and stated, "What's the educational benefits of having a mixed-gender as opposed to single-gender? There are times where a single-gender [event] makes more sense." James concluded by noting that the local context in Qatar is completely different from the context in the United States. Based on those differences, he stated that he did not recommend applying U.S. practices to the local context in Education City or vice versa.

David. David has worked in Education City for 2 years. In his leadership role, he oversees the department of student affairs and all of its services to students. He also

supervises full-time professionals. He was recruited from the United States. Similar to other participants, David started his interview by discussing the high school preparation level for his IBCs students. He noted that their limited exposure to leadership opportunities in high school had an impact on their maturity and readiness for college involvement. "At times, it feels like a prep school or high school." David recalled a couple of incidents that he had experienced in dealing with cultural conflicts and commented that he had struggled with the situations because of his lack of knowledge of the cultural expectations. During these situations, he was approached by students or colleagues either to stop the music or to stop the program during prayer times. He concluded that it is very important to keep a balance between honoring the main campus culture and respecting the local culture and host country. "It's a delicate balance. . . . We try to finds ways as much as possible to blend and honor both."

Oliver. Oliver has worked in Education City for 2 years. He works in student affairs as an academic advisor. He also manages some of the academic programs and events for students. Oliver started by describing his students. Although they belong to various cultures, the majority are Muslim. Similar to other participants, Oliver emphasized the need to adjust practices to meet student needs. "We have to demonstrate sensitivity to their needs." Oliver also mentioned the impact of the local context on students' lives, which he had not experienced in the United States. For example, he mentioned the holy month of Ramadan and described how students must adjust to balance meeting their religious obligations and fulfilling their academic requirements. He is happy to assist his students in meeting this challenge. Oliver also focused on describing his transition and the cultural differences between Eastern and Western cultures. For

example, he described himself as a direct communicator ("Even among Westerners, I am considered direct"). He recounted his struggle to use that skill in an Eastern culture that support indirect communication.

Amelia. Amelia was the only participant to have been recruited from a country other than the United States. However, she had been educated and had worked in the United States before joining her IBC in Qatar. Amelia is currently working in the student events area and is developing leadership initiatives for students. She has worked in Education City for 2 years. Amelia started her interview by comparing Eastern and Western cultures. "Our students come from . . . collectivist cultures; they tend to bring their cultures with them." She highlighted the impact of the local cultural on setting the context for student activities and programs. She emphasized need to respect the local cultural and host country, saying that the focus should be on "Finding ways to show consideration but also respect for our local culture and customs and religion and traditions."

Samantha. Samantha has worked in Education City for 6 years. She was recruited from the United States, and she is currently coordinating student leadership programs at her institution. Similar to Amelia, Samantha started by identifying the local culture as a collective society, in contrast to the United States. She cited some of the unwritten rules that are common in any society and stated that student affairs practitioners need to learn and acknowledge these rules to be successful. "There's a lot of unwritten culture rules that people can explain when they explain the how-to-do something here." At the end of the interview, Samantha recommended caution in following U.S. practices based on a presumption that that they are superior; rather, she

suggested keeping an open mind to learn how to adapt successfully to the new cultural context.

Kathy. Kathy has worked in Education City for 10 years. In her first role, she was responsible for student activities and events on her campus. In her current role, she is responsible for international students and study abroad initiatives. She also coordinates the sports programs. Kathy began her interview by highlighting cultural differences and describing her students within the local context in Education City. One of the major differences that Kathy noted was how she views her students as very dependent on their families. "Students here are more sheltered. Those apron strings are still connected." Furthermore, Kathy talked about the importance of investing in building relationships. This was very important to Kathy, as she felt that her students started to treat her differently and trust her more. Kathy then focused on cultural impact on her programs and events. For example, she mentioned that they do not have events on Friday because of how this day is viewed in the local culture. "We don't program on a Friday. Friday is a family day. Friday, people go to pray and they spend time with their family."

Lina. Lina has worked in Education City for 4 years. She is responsible for supporting international students and coordinating international service trips for her institution. She was recruited directly from the United States. Lina started the interview by reflecting on her interactions with students and how cultural differences played a role in these interactions. For example, she mentioned that she prefers to communicate via email but her students prefer face-to-face interactions in order to get to know her personally. "They very much value that face-to-face interaction." Lina shared her fascination with the concept of time. She noted that her students always plan things at the last minute because they view time as a flexible entity. "It may not be as early as we want it, but it's all going to work out." She elaborated on this example, noting that it is acceptable to have different cultural values and that it is important that student affairs practitioners respect such values.

Emma. Emma has worked in Education City for 6 years. She was recruited from the United States and currently serves in a leadership position in her student affairs department. Emma started her interview by recalling that she had not been very aware of IBCs or the local cultural context in Education City when she arrived. "I did not quite know what to expect. . . . I did not understand the fact that [people in] Qatar . . . speak English widely." Emma cited differences in cultural aspects and described how she had modified some of her practices to meet the needs of her students. She noted that her adaptations "provide an inroad to engaging and to being more culturally relevant or religiously relevant." Emma concluded by offering recommendations for new student affairs staff members who are planning to join IBCs in the future. For example, she suggested, "Ask more questions, listen . . . don't impose your perspective or experiences."

Summary. The participants provided valuable information to address the research questions. Three main themes emerged related to the three research questions. Each main theme contained subthemes. The main themes and subthemes were supported directly by the participants' words.

Theme 1: The Role of the Local Culture

The participating student affairs practitioners who are working or had worked in the local context in Qatar confirmed that they had modified their Western practices to become more suitable to the local cultural context. The three sub-themes related to the role of the local culture were (a) conservative cultural context, (b) respecting the local culture and the host country, and (c) involvement and engagement outside the classroom environment.

Conservative cultural context. The participating student affairs practitioners described the local context and noted how it influenced and shaped their work and relationships with students. Jack described the existing local culture in Education City and stated, "There's a home culture here, a home culture in terms of Qatar and Qatari culture, which is related to Islamic and Arabic culture." William mentioned that his local students are more strongly committed to their faith than are his U.S. students. "Muslim and Islam is so much more integrated in the student lives, than even students that I know in the US who consider themselves to be very religious." Oliver connected the same idea: "Certainly, there is majority Muslim population here with a variety of cultures that are connected in various ways to their faith." Jack similarly stated, "I think one unifying factor is many of the students are Muslim, or come from a Muslim family; maybe even from the Gulf region, so culturally there are some shared elements there."

In terms of how the students' faith influences their daily lives, Kathy described the influence as "strong." "It is vital. It holds a huge weight. These students pray before they make decisions. They pray before they go into a test. That spiritual component is really big for them. . . . It's a big component for students." Sophia noted that, in the United States, religious matters are usually kept private and not discussed in public, while in the Middle East religious matters are more public and are clearly expressed by Muslims. "In a deeper way, there's a stronger religious and cultural identity that is

manifested in behavior more here. In the United States, people can keep their religion private . . . here, the linkages are more public and more clear."

Several participants explained the importance of prayer times and described how they experienced this during their events. For example, David described a situation that he had encountered during an event. They were coordinating a student event and playing music in the background.

One of my co-workers who had been here a longer time mentioned to me, "Hey, will you just pause it for a second? I will explain here in a second." So, I did. . . . there was a prayer time that was scheduled to happen. . . . And so during prayer time, we avoid playing any music.

As the participant further explained, he learned from his co-worker that music should not interrupt the call for prayers or the prayers themselves. A similar situation happened with Kathy.

I was having a sports banquet and we're playing music and the students were like, "Turn off the music. Turn off the music." I'm like, "Why?" They were like, "It's prayer." I was like, "Oh okay." So then I heard they call the prayer and I was like, "Okay, shut it down. Shut down the music."

Several participants noted the limitations of having male-female interaction during student events or generally in public. Emma explained the local cultural context and how male-female interactions are viewed. "Culturally, being seen as a female with a male publicly could have cultural implications about the perceptions that others have about the interaction." She explained that relationships between genders are forbidden before marriage. David discussed the official greeting in the local culture, indicating that it would not be the same as the handshake in the United States. "I was certainly told that, unless a female extends their hand for a hello, then I should not extend mine and that it's perfectly acceptable just to put your hand on your heart or your chest."

Lina described a situation that took place during an activity that she planned with students.

There was an ice breaker where you were meant to touch someone and instead of touching someone or holding hands you would hold pens, the ends of pens, so you didn't need to physically touch somebody else, especially in a co-gendered event.

Over-all, a majority of the participants described the environment as conservative, strongly influenced by the Arab culture and the Islamic religion. These examples illustrated how the local culture and religion shaped the environment and the work of the student affairs practitioners in their IBC.

Respecting the local culture and host country. During the interviews, the majority of the student affairs practitioners expressed a clear commitment to respecting and honoring their local cultural context. Amelia discussed her process of planning events and how she considers the local context when she reviews the details of events. "I really have to consider my context. So, what kind of event is this going to be? Is this appropriate for the context, first of all?" Lucas highlighted challenges of explaining some laws and regulations form the United States, such as sexual assault and Title IX, without offending the students, parents, or the local community.

How do we have a conversation with students around sexual assault without offending them? . . . So we have to figure out ways that we could have a

meaningful education conversation. We will also try to be as respectful as we possibly could to our host country.

David highlighted the main goal and purpose of his work in Qatar. "I think that's our biggest responsibility to being employees here in the state of Qatar, and being able to bring the education and be part of the culture here respecting local laws, religions, cultures that already existed here."

The participants clearly acknowledged the importance of respecting local cultural norms, values, and beliefs. In addition, they were appreciative of the host country and how mindful they should be to not offend the host country by introducing or implementing programs that would conflict with the local cultural context.

Involvement and engagement outside the classroom. Several participants indicated that many of their students were experiencing leadership opportunities outside the classroom for the first time. William stated, "A lot of our students, when they got to college, that was their first exposure to any kind of leadership opportunities . . . basically, they're much less experienced than what it had been my experience working with students in U.S. schools." Olivia mentioned that the concept of becoming involved outside the classroom is a foreign idea. "It seems as though this idea in involvement is more specific to the U.S. than it is international." Similarly, James indicated that the concept of engagement outside the classroom is not known or acknowledged by the local culture. "It'd be like, goes back to like, the cultural part…there wasn't as much of a value [on]… outside of the classroom programming, but that's such a core part of like, US higher education." The participant further explained that this made it more difficult for students when they needed to explain their involvement to their parents or to ask

permission to participate in a leadership opportunity.

The second issue that many of the participants indicated was related to how their students did not have basic exposure to leadership opportunities because these opportunities had not been available in their high schools, in contrast to their counterparts in the United States.

You work at a school with a lot of students [who] will not get involved with organizations or athletics, or some kind of leadership position in high school so that ... basically, they're much less experienced than what it had been my experience working with students in U.S. schools.

David described the maturity level of his local student. "At times, it feels like a prep school or high school when it comes to maybe some of the students' maturity level." Similarly, Sophia stated that he had to adjust his practices to meet his students' leadership level. A participant stated, "Figuring out how to set up student activities was very, very challenging. . . . The students did not have a background in college activities." She elaborated that she had to explain to the students the purposes of these activities and seek their interest in order to establish involvement programs for students.

As a result of the students not fully understanding the value of engagement outside the classroom (as emphasized in the United States) and not being prepared to become involved, many participants indicated that it was very challenging to engage their students in activities. Several participants described their environment as a "commuter college," where students attend classes and then go home. Olivia stated,

It feels almost like a commuter campus where folks they come, and they're there to be students; they go to class and then they go home and it's a challenge to get student buy-in to that engagement outside of the classroom. James stated that his students tend to focus on academic work and ignore leadership involvement opportunities because they do not see these opportunities as important to their overall development as students do in the United States. "Outside of the U.S., the student is mostly focused on academics first and then maybe on the involvement second."

To summarize, the student affairs practitioners who participated in this study described their environment as very conservative, with culture and religion playing a significant role in defining accepted norms. The participants were consistent in indicating the importance of respecting local cultural beliefs and host country customs, traditions, laws, and regulations. The participants highlighted the challenges of engaging students in activities, as many were new to such a concept and did not see the value of such involvement. This may suggest reasons for adjustments that the participants made in their practices to ensure that programs and activities were relevant to their students.

Theme 2: Student Affairs Professional Practices

This theme highlighted differences between Eastern and Western cultures, according to the participants' perceptions. In general, the theme was that Western practices are generally not very effective in Eastern cultures. The three subthemes that emerged were (a) replicating Western practices is not effective, (b) differences between Eastern and Western cultures, and (c) gender interactions and mixing.

Replicating Western practices is not effective. Several participants mentioned push from their management to replicate experiences and traditions from their main campus in the United States in their local IBCs in the State of Qatar. The participants clearly agreed that this was not feasible or effective; they had to alter experiences and traditions to fit the local context. Jack expressed his commitment to his students: "You can't just take what's in the U.S. and then implement it. It's not just a cut-and-paste kind of thing. . . . I still want to be culturally sensitive, and define what works in this context with these students." Samantha confirmed the same idea, comparing some routine things that she usually did in the United States with those in her new international environment. "You can't push the U.S. way of doing things out here . . . some things will just take 2 weeks where in the U.S. they may take 36 hours. And you can't push this. You can't change it." James highlighted the need for creating a local campus identity for the branch campus, instead of using the identity or culture from the main campus in the United States.

I think it's a really flawed model for branch campuses to try to recreate the main campus because it's not possible. So, yes things will get adapted... [we should] push for more local practices rather than just adopting things from the main campus.

David discussed the delicate balance that he needed to establish between the two cultures (the home campus and the local community). "It's a delicate balance between … what I would call our campus culture and the country culture. And so we try and find ways as much as possible to blend and honor both." Logan mentioned the same concern and addressed the issue of preparing students to face the current and future challenges as they join the workforce and their local community. "That's always the balance of what is progress, what is engaging students, but not at the expense of losing the local cultural identity."

Overall, the participants agreed concerning the necessity to adopt or alter their practices to be more efficient and to meet the needs of their students. The participants also highlighted the necessary balance to honor and respect both cultures (home campus and host country).

Differences between Eastern and Western cultures. The participants elaborated on cultural differences between the United States and the Middle East, citing many examples in the interviews. William stressed the difference in how family expectations and obligations are viewed between Eastern and Western cultures. "[In the United States], most college students are trying to distance themselves from their families and gain that independence away from their families." However, he noted that, locally,

Our students are very connected to their families even as college students . . . just culturally, the students had a lot more obligations with their family. . . . Parents still have much more say in terms of what their students did or did not do.

Emma highlighted the same fact in terms of the family role. "Parents still had control over that decision making . . . students having to ask permission from their families to do things like come back to campus in the evening or to do something on a weekend." James discussed how these family commitments have affected student participation in events.

Family commitments seem to trump any of those extracurricular opportunities for students . . . their family just wasn't as supportive of it...I thought that was kind of an interesting difference ... from like the traditional campus-based in the US.

The important role that family plays in students' lives was further explained by Amelia. "We have more of a collectivist culture. And our students come from these kind of collectivist cultures, they also tend to bring their cultures with them." Similarly, Kathy added that, in a collectivist culture, the family unit is the main focus, not the individual person. "Students in the U.S. tend to be more individualistic in their approach or how they live their lives or make decisions. . . . This is one of the biggest differences between students in the U.S. and students in Qatar." Samantha confirmed the idea, stating that parents are well respected and play an important role in student lives, based on local cultural expectations.

The students in Qatar are, because it's more of a collectivist culture in this part of the world than in the U.S. . . . I think the respect for the parents that is over here, that is taught, again, because of the collectivist culture.

The second issue that was highlighted in terms of cultural differences was the concept of time. The participants described how their students define time as "fluid" and "flexible," compared to the United States, where time is considered a valuable commodity that should be respected and not wasted. Sophia stated,

We really struggled initially with concept of time and planning efficiency [for student events] . . . coming from the U.S., you do things far ahead, you have to do things in advance... Here, things work at far more, kind of a last-minute kind of way.

Lina described how her students responded when they were planning their events. "Things are more fluid. 'Oh, it will work out, like it's going to be fine. . . . It may not be as early as we want it, but it's all going to work out.""

Logan highlighted the same challenge:

The concept of time is [that] time is very fluid. So, time is merely a suggestion rather than a contract or a commitment. . . . When I say that something starts at

6:30 pm, you can probably show up at 6:45 and 7:00 and it would probably still be okay.

Logan explained that he had experienced multiple challenges in having to adjust the time for his local students and extending the program time because many of them arrived late to participate in these events.

The third cultural issue described by participants was the importance of establishing personal relationships with students. This entails spending the time to get to know them before discussing business-related items. Lucas indicated the importance of developing relationships in order to be successful. "In order to have any kind of relationship with the students, it was incumbent on me to develop a personal relationship ... like the students require you to have a strong relationship with them first and do the work later."

Kathy described a disconnect with her students when she started working and how this changed over time as she built strong personal relationships with her students. "I found ... that, to get them more engaged and involved, it's about relationships.... I learned that you really have to cultivate a relationship to be successful."

Emma explained how the relationship differs between the U.S. perspective and the Middle Eastern perspective. "Culturally, the relationship is the foundation of so much, whereas if you are like in the U.S., there's a lot of really transactional interactions where it's like a, 'Okay thanks. Bye.'"

The participants addressed many cultural differences, including the family role, the time concept, and the importance of establishing a relationship. This set the stage for many of the adaptations or changes that these professionals have made to enhance their services to meet the needs of their students.

Gender interactions and mixing. One of the areas that all participant addressed was that the co-gendered environment is different at IBCs in Education City from that in the United States. For example, many indicated that they offered both single-sex and cogendered programming. Emma stated, "We also wanted to make sure there are spaces where women only and men only can also engage co-curricularly. . . so we have a student organization that is specifically for women."

Sophia highlighted the same concern and mentioned an event that involved traditional dancing. She said that they had to create a separate space, a private space, for the females so they could dance and participate fully in the event. "Obviously, the boys and girls couldn't be together. . . . The guys participated by doing the sound system, but they had to be in a separate room."

William described a question that the parents always asked him during new student orientation: whether their son or daughter would be studying in a co-gendered environment. "A question that still came up every year in terms of concerns about even boys and girls being in the classroom together, let alone them being in a sort of clubs and organizations together." He explained the challenge of engaging students in extracurricular activities, but agreed that making accommodations to have certain events single gendered was helpful, as well.

The majority of the participants cited these examples to illustrate that they have to adapt their practices to fit the local context. They also mentioned ways to advise staff members who are relocating to work in the Middle East. This information assisted in addressing the third research question.

Theme 3: Best Practices for IBCs in the Middle East

This theme highlighted many of the successful changes that the student affairs practitioner had adopted to make their services more relevant to their students. The theme included important recommendations by the participants for those who are considering work in IBCs in the Middle East region. Amelia best summarized this main theme in the following statement: "Everything is driven by the local context or the cultural context we are in." Two subthemes emerged related to this research question: (a) adopted successful student affairs practices, and (b) best practices for incoming staff.

Adopted successful student affairs practices. All 14 participants expressed deep commitment to their students' development and growth. They also indicated their flexibility to modify their practices to ensure that the practices meet their students' needs. This flexibility and adaptability seemed very important to these student affairs practitioners who work in IBCs. They agreed that it is a key to their success. Lucas stated, "If it's important to the students, you adapt to the culture. . . . We say, 'Meet the students where they're at." Samantha mentioned a similar perspective: "We learn that you have to meet students where they are, and kind of adjust to them."

Emma mentioned the importance of getting to know students in order to be effective in student affairs approaches and practices. "We need to be open and cognizant of that as well because this is the population that we are serving so we need to know them. We need to know what their needs are in order to meet them." Similar, William highlighted the same issue by mentioning the need to be flexible with the students' initial level of leadership skills, as an example. "Just in terms of meeting the students where they were with their leadership experience that they had coming in." All 14 participants confirmed that they had modified their Western practices to be efficient, to be respectful of the local culture and host country, and to meet the needs of their students. The participants cited multiple examples of these modifications.

First example: Events and programs timing. The participants discussed in depth the struggle to determine the best time for their events, programs, and activities. For example, all participants agreed that they do not plan events for Friday, which is the Muslim holy day and family day. William stated, "For a Muslim student, that is something that they are very immersed in, the prayer times, we never did anything on Fridays to be accommodating to students."

The family commitment and obligation, as explained earlier, also limits the students' ability to participate, especially during the evening. Most participants indicated that they plan events to occur during the day, especially during class break time, to allow students to participate. For example, Olivia stated,

Recognizing that it is difficult sometimes for students to be able to commit to participating in involvement opportunities outside their typical kind of classroom schedule. . . . There were a lot of programs during the day, especially over lunchtime. . . . Having it during the day it was better than at nights.

Oliver confirmed the idea and indicated that students do not attend any events that conflict with their scheduled classes.

It should not interfere with the regularly scheduled courses of the day. Having it immediately after their final schedule session seems to be away to get the most participants. Asking them to come back several hours after classes are finished, is a recipe for disaster. Many participants mentioned that students' families expect them to come home and to be with the family at dinner, which introduces difficulty in arranging evening and weekend events. Kathy noted, "We also don't program in the evenings; all of our programs are during the lunch break time from 12:00 to 2:00, because that's our break or block that we set aside for no classes."

Other participants offered ideas about student events. They mentioned restrictions in including music and public dance in their events. Sophia stated, "In terms of campus programming, we try to be mindful of music, we try to be mindful of dancing. . . . We try to pull back and make sure that we were not offending anybody, we aren't offending our host country." Similarly, Emma described a situation during one of her events. The students danced during the event, which had not been planned. Some students who were attending the event approached the practitioner and stated, "Oh, no, dancing is *haram* (forbidden according to Islamic teaching). You can't do it."

Second example: International travel opportunities. The second example mentioned by the participants was related to student international travel. Several participants mentioned the concept of having a family member travel with a student, especially a female student. According to one participant, an unmarried female is expected to travel with family members in accordance with Arabic cultural expectations. Olivia stated, "If they were able to participate, sometimes a family member would join on that trip, which is something that I've not experienced in the U.S. or New Zealand, where I worked before." This participant explained the difficulties of having female students participate in international service-learning or exchange opportunities. Many students are unable to convince their families or to have an available family member to travel with them.

Emma highlighted the same issue and suggested flexibility to allow this accommodation in order to have the student travel on these opportunities. "Some of the international trips that some students would only be able to participate if their family attended as well. . . . We try to accommodate as long as the student is able to fully participate in the experience."

William argued that the same expectation (for having an escort accompany students) applied to evening and weekend events.

We made an allowance for students to be able to bring a guest because again it was an evening activity.... The concern was for our female students, if we didn't allow them to bring a guest with them, then they wouldn't be allowed to attend by their family.

Several participants cited the need to ensure that certain things happened during travel. For example, ensuring that Halal food (for Muslim students) is available and that students are allowed to pray at scheduled times. Samantha stated,

We would need to build in time while we were traveling for students to be able to pray in our daily itineraries. We had to be very careful about food and the available choices for students...making sure that the food they were getting was Halal.

Third example: The holy month of Ramadan. The third example that the participants highlighted was the holy month of Ramadan. During this month, Muslims fast during the day, only to break the fast at sunset. The local governmental laws do not allow food vendors to be open during the day during Ramadan, and it is prohibited to eat

in public during the day. The participants confirmed that most of their events could not take place during the month of Ramadan; they utilized other opportunities such as organizing Iftar to bring all students together. Kathy stated, "With orientation, if it's during Ramadan, we had no food and then have Iftar. . . . So we have to accommodate for that." Oliver discussed the impact on academics and arrangements that are made to accommodate Muslim students who are fasting. "During Ramadan, the class schedule changes. Days are often shortened. . . . I am not aware of many U.S. institutions that adjust their schedule during the holy month of Ramadan." All participants described the same general impact and noted that many events were planned without food or done after the Iftar time.

As I worked professionally in two separate IBCs in Education City, I encountered the same challenges as those faced by a student affairs practitioner. For example, the majority of the student events that I coordinated usually took place between 11:30 and 12:30 pm during the business day. Last year, I made a similar accommodation for a female student leader who was planning to attend a conference in the United States. We allowed her father to serve as a chaperone. In both IBCs where I worked, they made special arrangements during the month of Ramadan to accommodate students who were fasting. In general, all IBCs in Qatar follow the same practices and procedures.

Best practices for new incoming staff. Several participants reflected on their earlier experiences and indicated that it took some time to adjust, to learn their local cultural context, and to build strong relationships with their students. They agreed that much time is required to learn the local culture and context to build relationships. The participants offered other recommendations for new staff. First, it was recommended that new staff be patient, observe the environment, and not make any assumptions; instead, ask questions. Logan suggested, "Just try to observe everything. Like with any culture, try to not assign good or bad meaning to anything; just kind of observe everything."

Second, it was recommended that new staff be open minded, not automatically applying previous experiences and perspectives to the new local context. David suggested "being open-minded. Realizing that, just because you are used to doing things in a certain way may not be the best way to do it here." Emma suggested, "Ask more questions. Listen, don't tell. Don't impose your perspectives or experiences on this place and be willing to listen and learn."

Third, it was recommended that new staff not make any immediate changes. Rather, it was suggested to experience all of the events for a full 1-year cycle and ask questions during this process. Jack suggested, "Try not to change anything. . . . Go through a cycle with the traditional events, with all the things that happen, just to understand the full cycle." William noted, "Unless you understand the full context of why it is the way it is, it's hard to make changes."

Fourth, it was recommended that new staff learn about the culture and local context and adjust practices accordingly. For example, the new staff member should be aware of family obligations and how to accommodate those expectations so the student can participate fully in events and programs. Amelia suggested "adjusting the theory and practice, applying it to a different cultural and context."

Fifth, it was recommended that new staff focus on building relationships with students from the first day of service. Emma stated, "the relationship is the foundation of

so much."

Sixth, it was recommended that new staff look for opportunities to develop cultural competency, if it is offered by the institution. James stated, "Do I think training on cultural competence and contacts would be valuable? Yes, 100%. I think that should be part of onboarding."

Samantha recommended the following for incoming staff:

I would say is if you chose to come here and you're a U.S. citizen, then just be really careful about thinking that the U.S. is superior, or you're better than these local people or those types of things. I think sometimes people come here, and they don't want to change their way of life at all. You can't really do that if you move to another country. So I think you need to be open minded and be willing to grow and adjust and change even if it is the same institution that you worked for in the U.S., knowing that it is going to be a really different experience. And you have to be open minded and be willing to change and adjust yourself to how the culture is so that you can thrive there.

Summary of the Results

This chapter presented findings related to each of the three research questions that guided this study. The qualitative analysis resulted in eight major themes, all of which addressed the services offered by student affairs practitioners in IBCs in Qatar.

Related to first major theme, the participants described how their local environment and cultural context are shaped. The local context is influenced by the Arabic culture and Islamic religion. Students were described as having had only limited experience with leadership involvement opportunities outside the classroom when they enrolled at the IBCs. The participants clearly agreed that they are committed to respect the local cultural context and their host country's traditions, culture, laws, and regulations.

Related to the second major theme, participants discussed that replicating Western student affairs practices within their local context was not efficient or effective in meeting student needs. Participants discussed in detail the differences between Western and Eastern cultures. They cited multiple examples, such as the impact of family expectations, to illustrate these differences. They described gender interactions and mixing during student events as limited and restricted by local cultural norms.

Related to the third major theme, participants described successful practices that they had implemented after modifying their Western approaches to fit the local context. All participants agreed that they had to make modifications to meet local needs of their students. They cited three successful approaches for revising their practices: events and programs timing, international travel opportunities, and the holy month of Ramadan. They presented six specific recommendations for approaches that should be adopted by incoming student affairs practitioners who are relocating to work in the Middle East region.

All participants in this study had been exposed to a new environment that challenged their established Western practices and opened the door to developing new approaches to assist their students. The commitment by the participant to serve their students led them to think creatively and establish or modify practices to meet the needs of their students.

Chapter 5

Discussion

The focus of this study was to examine and understand Western student affairs practices implemented in the six American IBCs within the local cultural context in the State of Qatar. The study also addressed modifications that student affairs professionals made to adopt their practices to meet local students' needs. Additionally, the study explored possible best practices that could be adopted by new student affairs professionals who are in transition to join IBCs in the Middle East region. Some participants contended that these practices could also be adopted at other IBCs, as well.

The chapter begins by reporting results to address the research questions, followed by analysis and discussion of findings, the role of the theoretical framework in explaining the findings, implications for practice, and recommendations for future research.

Results for Research Questions

This study was focused on Western student affairs practices as they are implemented or adapted in Education City, Qatar. The findings from the qualitative interviews suggest that student affairs practitioners have altered Western practices in consideration of the local cultural context and the needs of their local students. Many of the participants reflected on their practices and experiences and shared lessons that they had learned that could assist new student affairs practitioners with their transition to IBCs in the Middle Eastern region.

Research question 1: The role of the local culture. The first research question was, *How do student activities professionals describe the role of local culture in shaping*

Western student affairs services implemented on their branch campuses? All of the participants highlighted the strong influence of the local culture and religion on their students' interests and participation in extracurricular activities offered by the student affairs offices. For example, one of the major topics that participants discussed and highlighted during the interviews was the significant role of the family in students' lives. William described the strong role of the family in values and cultural differences between Eastern and Western cultures. "[In the United States], most college students are trying to distance themselves from their families and gain that independence away from their families. . . . [In Qatar], our students are very connected to their families even as college students." Amelia and Kathy confirmed the idea but directly characterized the local society as collective in nature versus the United States being an individualistic society. Amelia stated, "We have more of a collectivist culture . . . they [students in Qatar] also tend to bring their cultures with them." Kathy said, "Students in the U.S. tend to be more individualistic in their approach or how they live their lives or make decisions... This is one of the biggest differences between students in the U.S. and students in Qatar."

Participants indicated that parents exerted a strong influence in the decisionmaking process, especially regarding being involved outside the classroom. William stated, "Parents still have much more say in terms of what their students did or did not do." Emma confirmed the idea: "Parents still had control over that decision making . . . students having to ask permission from their families to do things like come back to campus in the evening or to do something on a weekend." Similarly, James discussed how family obligations influence the student's ability to participate in extracurricular activities, especially in cases where conflicts existed. "Family commitments seem to

trump any of those extracurricular opportunities for students . . . their family just wasn't as supportive of it."

These student affairs professionals were concerned about the possibility of some of their Western practices offending the local culture, students and colleagues, and the host country. Thus, they were considerate in planning and implementing programs and activities and were careful to make appropriate modifications. For example, Amelia discussed how she considered local cultural values and traditions, especially when planning events, to avoid any signs of disrespect. She stated, "I really have to consider my context. So, what kind of event is this going to be? Is this appropriate for the context, first of all?" Similarly, Lucas discussed some of the legal language used openly in the United States that might be considered as offensive in the local culture, such as *sexual assault*. He stated that he needed to be indirect in stating such language.

How do we have a conversation with students around sexual assault without offending them? [Students]. . . So we have to figure out ways that we could have a meaningful education conversation. We will also try to be as respectful as we possibly could to our host country.

David highlighted similar concerns and stated that student affairs practitioners have an important responsibility to deliver educational programs that fit the local context and meet student needs. "I think that's our biggest responsibility to being employees here in the State of Qatar, and being able to bring the education and be part of the culture here respecting local laws, religions, cultures that already existed here."

There was a clear consensus by participants that the local cultural context significantly influences their work and practices. As many of them emphasized,

"meeting the students where they are" is an essential philosophy in their approach to working with students in IBCs.

Research question 2: Student affairs professional practices. The second research question was, In what ways do student activities professionals alter their Western practices at IBCs in Qatar? All participants agreed on the importance of altering practices to meet the needs of students. One participant stated, "If it's important to the students, you adapt to the culture." All participants confirmed that they modified events and programs to be suitable and relevant to their students. For example, many participants mentioned that they accommodated cultural expectations by allowing students to have a family escort during participation in international opportunities, such as service-learning trips. Emma was extremely supportive, as were other participants, and allowed such a practice to ensure that her students were not disadvantaged by not being able to participate. "Some of the international trips that some students would only be able to participate if their family attended as well.... We try to accommodate as long as the student is able to fully participate in the experience." Similarly, William was very supportive of allowing students to have a family escort, especially for events taking place in evenings and on weekends. He stated,

We made an allowance for students to be able to bring a guest because again it was an evening activity. . . . The concern was for our female students, if we didn't allow them to bring a guest with them, then they wouldn't be allowed to attend by their family.

Another example was related to the limitations for gender interactions within the local cultural context, which shaped student events. For example, many participants

indicated that they offered a mix of single-sex and co-gendered programming. Emma explained that sometimes it makes more sense to have all-female events versus all-male events or have mixed events. "We also wanted to make sure there are spaces where women only and men only can also engage co-curricularly. . . so we have a student organization that is specifically for women."

Sophia described one of her events that involved a traditional dance; she related that they had to create a private space for the female students separate from male students so they could participate fully in the event. "Obviously, the boys and girls couldn't be together. . . . The guys participated by doing the sound system, but they had to be in a separate room." Sophia provided an event poster demonstrating the separation between male and female students by having two separate locations for the event.

The participating student affairs practitioners discussed the importance of creating a balance between practices and philosophies that were derived from their main campuses in the United States and local values, customs, and practices. As David indicated, referring to U.S. and Arab cultures, it is important to find a balance in implementing U.S. practices within the local cultural context. "It's a delicate balance. . . . We try to find ways as much as possible to blend and honor both." Furthermore, some participants highlighted the importance of initiating local traditions to create a unique identity for their branch campus and its students. The material culture artifacts collected from participants illustrated modifications made by the participants to make their events and programs relevant to the local cultural context. For example, William showed a parent's agreement form needed for international travel, regardless of the student's legal age, to demonstrate the parental influence during these processes. **Research question 3: Best practices for IBCs in the Middle East**. The third research question was, *What best practices can emerge from the study findings to guide student activities professionals working at IBCs in the Middle East Region?* As the participants reflected on their experiences and changes that they had adopted, they cited successes in some of their approaches. For example, the majority agreed that it was important to change the timing of events offered to their students. They reported that they scheduled most of their events during the day, rather than in the evening, due to family obligations that students were expected to fulfill.

Participants noted major challenges in replicating culture from their main U.S. campuses on the IBC campuses. Many agreed that such a practice is neither efficient nor practical in terms of the goals of student affairs professionals. For example, Jack indicated that it is not effective to implement practices or campus traditions from the home campus in the United States to the branch campus located overseas. "You can't just take what's in the U.S. and then implement it. It's not just a cut-and-paste kind of thing." Similarly, James stated that IBCs should not be focused solely on reinventing experiences from their home campuses. "I think it's a really flawed model for branch campuses to try to recreate the main campus because it's not possible." Other participants were more direct and stated that student affairs practitioners should not force any Western practices on the local cultural context because it might lead to negative outcomes. For example, Samantha stated, "You can't push the U.S. they may take 36 hours. And you can't push this. You can't change it."

As a result, the majority of the participants recommended flexibility and balance between adopting some home campus practices and implementing some modified events to be relevant to the local culture. For example, David indicated the importance of achieving a balance for honoring both cultures. "It's a delicate balance between . . . what I would call our campus culture and the country culture." Other participants mentioned the importance of adjusting Western practices to meet the needs of students. For example, Lucas stated, "If it's important to the students, you adapt to the culture. . . . We say, 'Meet the students where they're at.'"

The participants reflected on how they had revised their practices to implement successful programs. They highlighted three examples to illustrate the importance of adjusting these practices: evening specific timing, accommodating certain requests during international travel opportunities, and holding events during the holy month of Ramadan.

Participants offered six recommendations for new staff members who are joining IBCs in the Middle East. The majority of these recommendations were intended to assist new staff members with the transition in understanding the new cultural context and establishing meaningful connections with students. Some of these recommendations focused on observing the cultural context, not making immediate changes without fully understanding why local practices had been established, and searching for opportunities to develop their own cultural competencies. For example, David stated, "Be openminded. Realizing that, just because you are used to doing things in a certain way may not be the best way to do it here." William added, "Unless you understand the full context of why it is the way it is, it's hard to make any changes."

Discussion and Analysis

In the previous chapters, I highlighted the literature from Education City and from around the world addressing IBC operations, cultural differences between the IBC and the host country and related challenges, and the important work by student affairs professionals around the world. In this section, I review significant themes that were introduced in Chapters 1 and 2 and discuss how they connect to the study findings.

The role of the local culture. Some of the participants highlighted how some of their students did not fully understand the purpose of student affairs or the activities implemented outside the classroom. As a result, they were not fully engaged in these activities and programs. The participants indicated that one of the main reasons was the lack of student exposure to these activities during their high school which is limited within the local cultural context. The reports prepared by UNESCO (2009) have explained the same concerns, especially within the Gulf Region. In these reports, the students and their parents did not fully understand the purpose of student affairs nor the services and programs offered by the student affairs practitioners (Callahan, 2016; UNESCO, 2009).

Another important area that all participants highlighted was the strong family structure in which the parents play a significant role in setting expectations for their children (Ayubi, 1991; Barakat, 1993; Nejem, 2016; Zaharna, 1995). For example, in a study by Bakken (2013), the parents played an important role in influencing their students' decision to apply to the IBCs in Qatar. The participant in this study reported having encountered this cultural characteristic regarding family expectations and obligations. For example, several participants mentioned that their students "disappear" immediately after classes because they are expected to be at home to interact with siblings and to be at the dinner table. The participants described how family expectations have limited students' ability to participate in student affairs programs and activities. For example, many students have difficulty in explaining to their parents the available international leadership opportunities, particularly those that involve overnight participation. In some cases, a family member travels with the student, especially with a female student. The study findings confirmed the clear impact of the family on student affairs practicipation in leadership experiences outside the classroom and the need for student affairs practitioners to adjust their approaches to accommodate and support their students. Furthermore, these opportunities were considered by the parent s to be extracurricular activities than curricular activities.

Two other important Arab cultural values were highlighted by the participants: (a) the value of establishing a strong relationship with the students, and (b) gender interactions. The majority of the participants agreed that establishing personal and trusting relationships was extremely important in connecting with and working with their students. One participant emphasized that it was important to spend time in getting to know his students before discussing the business on hand. Another participant described a complete change in attitude toward her by her students as they stopped viewing her as an "outsider" and began to trust her. The value of the personal relationship can be explained by the strong collective values that Eastern cultures exhibit, especially in the Middle East region (Hofstede, 2001; Minkov, 2011; Nejem, 2016; Neuliep, 2015; Nisbett, 2003; Storti, 1999; Ting-Toomey, 1999).

With regard to gender interactions, the Gulf region was identified as very conservative, limiting interactions between males and females outside of specific family approval and supervision (Ayubi, 1991; Barakat, 1993; Nejem, 2016; Zaharna, 1995). The participants highlighted this concern and stated the need to create a balance between opportunities for single-sex programming and co-gendered programming. Some participants reported that they designate some spaces in their academic building for women-only events and allocate other areas for gender interaction. These participants viewed this practice as an extension of the cultural values that are imbedded in the local community and a recognition of the importance of respecting such values.

Eastern cultures have many general values that were discussed by the participants in this study. Eastern cultures value flexibility with time, indirect communications, and harmony and saving face; there is a general belief that people have limited control over their lives (Hofstede, 2001; Minkov, 2011; Nejem, 2016; Neuliep, 2015; Nisbett, 2003; Storti, 1999; Ting-Toomey, 1999). The participants in this study reported that they have observed many of these values in their work environment and in their local community. For example, several participants expressed fascination or frustration about how the local culture addressed the concept of time. The participants explained that they did not expect logistical arrangements for their events to be successful at the last minute, without advance preparation. On the other hand, some were frustrated by this concept, reporting that they struggle in coordinating events for students who do not arrive on time or who complete tasks at the last minute. In general, they stressed the need to be flexible in dealing with all cultural differences, such as the concept of time.

Student affairs practitioners. *Culture* was defined as the common values, beliefs, and shared language recognized and upheld in a specific community (Nakayama & Yep, 2001). Misunderstanding and conflict usually occur when cultural norms are not clearly understood or followed in the community (Ting-Toomey, 1999). Student affairs professionals can be challenged if they are not familiar with their students' values and beliefs and the local cultural context (Moscaritolo et al., 2016; NASPA, 2016c; Otten, 2003; Stryker et al., 2016; UNESCO, 2002). The participants share several stories that demonstrated these challenges. For example, a couple of participants said that they had not fully understood requests from students or colleagues to stop playing music or to take a break during prayer times. The cultural norms in this local context recognized the importance of such requests. Although some participants clearly lacked knowledge of some local customs and the need to conform to them, the findings of this study showed a clear desire on the part of these student affairs practitioners to respect the local culture, religion, and host country traditions. They were learning the details and significance of such norms and they were learning how to acknowledge and respect them.

A few participants cited a need for training to improve their cultural competence. Several scholars have highlighted this important issue, especially for the student affairs staff who are relocating to work internationally (Ciobanu, 2013; Deardorff, 2016a; Moscaritolo et al., 2016; Osfield & Associates, 2008; UNESCO, 2002). It is imperative that student affairs practitioners actively engage in opportunities to enhance their skills and competencies in order to understand their local cultural context (Ciobanu, 2013; Deardorff, 2016a; UNESCO, 2002). Although the participants agreed on the need for such cultural training opportunities, they stated that such opportunities were not directly provided by their IBCs. They were forced seek such opportunities to advance their skills and abilities on their own.

Student affairs practices at IBCs. The study findings clearly showed agreement by all participant practitioners regarding the need to be flexible in their practices and approaches. Several researchers have confirmed the importance of this issue, describing how IBCs in general should establish a balance between replicating practices from home campuses and adapting them to the host country's cultural context (Altbach et al., 2009; Farrugia & Lane, 2013; Shams & Huisman, 2012; Stryker et al., 2016). Failure to achieve such a balance can lead to accusations of disrespect for host country values and traditions by the IBC (Altbach et al., 2009; Farrugia & Lane, 2013; Shams & Huisman, 2012). Participants discussed this balance in detail and agreed that that they are providing "Western" education within the local cultural context in Qatar. They also emphasized that focusing solely on "American" approaches or supporting an assumption that "US is superior or better" will not make student affairs programs successful in international work at IBCs. Shams and Huisman (2012) argued that forcing such approaches can be viewed as a type of cultural imperialism or a new form of colonization.

Deardorff Intercultural Competence Theoretical Framework

Although the purpose of this study was not to examine directly the student affairs practitioner's intercultural competence, this framework guided the study and provided insight into student affairs practitioners' decisions to alter their practices. The stages characteristics of the Deardorff (2012) model partially explain the developmental level of the participants and why they behave in certain ways when they encounter cultural

challenges.

For example, a couple of participants in their second year of service reported that they were still experiencing challenges in dealing with and accepting cultural differences in their work environment. David cited challenges that he had encountered when implementing his programs and events; he said that he was still struggling to understand some of the cultural differences. He wondered, "Why is this happening?" He also stated that he had to rely on colleagues and students to teach him about some important cultural norms, such as stopping the music during prayer time. Logan mentioned his struggle in adjusting to several cultural differences. He highlighted the concept of time as the greatest challenge. "I think the element of local culture that is the most difficult to navigate and adjust to is the concept of time." In both instances, David and Logan were still in the "attitudes" stage of the Deardorff model, as they were going through the initial adjustment period and working on developing a sense of curiosity and openness to examine and explore cultural differences.

On the other hand, participants who had worked for a long time in Education City clearly indicated that they had developed intercultural competence and had moved in the model to advanced stages. For example, participants who had worked in Education City for more than 6 years had advanced to the "skills" stage, as their answers and self-reflections demonstrated their ability to process and reflect on the cultural differences that they observed. Amelia was not only able to reflect on the student behaviors but was also able to connect them to Eastern cultural values, such as collectivism. She stated, "We have more of a collectivist culture. And if our students come from these kind of collectivist cultures, they also tend to bring their cultures with them." James reflected

critically on the larger picture of how the IBCs were focusing on replicating cultural experiences from their home campus instead of developing their own cultural identity. "I think it's a really flawed model for branch campuses to try to recreate the main campus. . . . It's not possible and that takes away the opportunity of the local campus to create their own culture."

The study findings were consistent with the stages of the Deardorff (2006, 2012) model and explained why and how participants handled their practices to make them relevant to the local cultural context. The findings also suggested the importance of developing such skills or abilities to establish meaningful connections with local students.

Implications for Practice

This study was designed to examine Western student affairs practices implemented or adapted at IBCs in the State of Qatar. The study also examined modifications made by student affairs practitioners to make such practices relevant to the local cultural context. The study findings confirmed that participants did not fully follow the practices that they had learned and utilized previously in the United States.

The participants argued that cultural differences were the main reason for adapting and forming new practices. They mentioned the importance of connecting and meeting the needs of local students. As this study was focused on determining whether local culture exerted an influence on student affairs practices, more research is needed to determine how to prepare practitioners to work in IBCs. This process can start at the recruitment stage and continue through the onboarding and settlement stages.

Several participants indicated the need for more targeted and intentional training opportunities to enhance their intercultural competence. The faculty members who

participated in Jauregui's (2013) study confirmed that developing intercultural competencies skills helped them to understand their students and the local context. As a result, they enhanced student learning experiences in their classrooms (Jauregui, 2013). Student affairs practitioners should invest in enhancing their intercultural skills and competence to improve programs and events. Further research is needed in this area to determine the best methods to train student affairs practitioners who are recruited to work in IBCs worldwide.

The recommendations provided by participants could be very effective in preparing newly recruited staff in their transition. The recommendations could also encourage establishment of local "community of practice" forums to serve as support groups during the transition. It is important to acknowledge that transitions between Western and Eastern cultures can cause difficulties and present challenges because of significant differences.

Implications for Research

Research on student affairs practitioners' experiences in IBCs is very limited. This study focused on determining whether student affairs professionals encountered challenges as they moved their Western practices outside the United States. This opened the door for many possible future studies that can be implemented to expand the study of this phenomenon.

First, further research is needed to assess how to prepare student affairs practitioners to work in IBCs outside the United States. Many participants reported that they had not been well prepared and had not anticipated the new environment in which they would serve. Some, such as David and Logan, were still struggling with their transition. Emma mentioned that she had been surprised when she moved to Qatar to discover that many in the community spoke English very well. As noted in this study by Jauregui (2013), it can be useful to determine whether student affairs practitioners would be benefit from informal and formal cross-cultural training opportunities to improve their cultural competence, which could lead to enhancing programs and activities offered to local students.

Second, further research is needed to identify students' perceptions of the best practices that student affairs practitioners should adopt and implement. For example, students in Bakken's (2013) study in Education City, Qatar, expressed hesitation to engage in extracurricular activities that they felt violated their cultural or religious beliefs. The participants in the current study mentioned that, in some cases, their students approached them to make these requests. This may suggest the need to make further adjustments or modifications to activities and programs to make them suitable for local students and the local cultural context. Another study could be implemented to seek student feedback that could be very useful in assisting student affairs practitioners in their future approaches.

Third, many of the participants highlighted external factors that made it challenging to engage their students in their activities and programs. One of these concerns was student preparation and engagement in high schools before joining these IBCs. Further research may be helpful in determining whether such preparation, similar to U.S. high school experiences, could be helpful in the student transition and engagement in extracurricular activities in higher education.

Implications for Leadership and Leadership Theories

Leadership can serve as an important process to guide people to achieve their goals, mission, and vision. Leadership involves a relationship between leaders and followers in which both can equally influence the outcomes of the process. Leadership theories inform ways to work together as leaders and followers to achieve desired goals and outcomes (Hughes et al., 1995; Kelly, 1995; Northouse, 2012; Rost, 1995). Higher education can play an important role in modeling how the leadership process is implemented, especially for generations who will lead society in the future. Hence, these institutions must invest time and effort to develop students, faculty, and staff by recognizing how the leadership process is carried out and accomplished.

Implications for leadership. This results of this study suggest that preparing student affairs practitioners who are deciding to work in IBCs may be beneficial in terms of the cultural transition that they must undergo. For example, some participants mentioned that they had not known what to expect before their arrival and/or had not been aware of the important cultural norms that impacted their events.

Transformational leadership (TL) can play an important role in establishing a shared process with goals that are shared by leaders and followers. TL acknowledges followers' needs and the shared plan to address these needs. In addition, TL promotes creation of collaborative approaches that focus on mutual goals. Leaders aim to influence followers through ethical, intellectual and inspirational practices (Burns, 1995; Couto, 1995; Megerian & Sosik, 1996; Shields, 2010). Exercising TL in IBCs can create a supportive environment that encourages both the senior management and the student affairs practitioners to generate common solutions to address the changes that are needed in practices and approaches. A successful example mentioned by several participants is that their senior leaders have established an open period during the business day to engage students in feedback on activities and programs.

Another important area that senior leaders can explore with their current staff is how to prepare incoming staff. Utilizing TL, senior leaders can work together with their current staff to establish training programs to educate the new staff about the local cultural context before and/or soon after their arrival. According to Deardorff's (2006, 2012, 2016a, 2016b) intercultural framework, only intentional structured programs utilizing a developmental framework can be successful in enhancing participants' intercultural skills and competencies. Such programs can be implemented in advance (during the recruitment process) to prepare new staff before their arrival. Additional development opportunities should be considered to continue the growth and development process for the staff, based on recommendations by participants in this study.

Implications for leadership theories. Leadership scholars in the United States have focused mostly on creating theories that were socially constructed to address leadership development in the Western cultural context (Chemers, 1995; Ciobanu, 2013; Couto, 1995; Goleman, 1998; Magolda, 2003; Shields, 2010). For example, the majority of Western leadership theories promote the individual development approach to encourage student independence and self-reliance. This fits well in Western culture, which promotes individuality; however, this would not work effectively in Eastern cultures, which promote collectivism. This may be problematic for student affairs practitioners who leave the United States to work in IBCs worldwide. The main concern is how Western practices based on Western leadership theory may be applicable outside the borders of the United States. Even though this study was not designed to explore application of Western leadership theory, the participants clearly implied that application of Western theories in their practices would not be effective or efficient in international settings.

As the IBCs movement continues to expand worldwide, there will be a need to develop leadership theories that will guide student affairs practices to become more applicable and transformable to a wide range of cultures and countries around the world. These theories shape student affairs practitioners' work and the leadership programs that they offer to students. More research is needed to develop such applicable theories to enhance student affairs practices around the globe.

Conclusion

As the IBC phenomenon has continued to grow globally, there is a need for research to identify best practices for IBC operations. This study's findings showed that Western student affairs practitioners who were recruited in the United States to work in IBCs in Qatar experienced many challenges in implementing and adapting their programs and activities. They have modified many practices to fit the local cultural context. The participant student affairs practitioners highlighted the importance of being respectful of local cultural and host country values and traditions.

The findings also confirmed cultural differences between the IBC's home country and the host country in the Middle East. The student affairs practitioners had to learn these cultural values differently, and all emphasized the need to be adaptable and flexible in such environment. They made recommendations that can be utilized by new student affairs practitioners who are moving to work in an IBC in the Middle East region. More research is needed to determine how to prepare and train student affairs practitioners to work in environment that are different from the ones that they experienced in the United States. Student affairs practitioners must seek opportunities to develop and enhance their intercultural competence so that their practices are relevant and meet the needs of their local students.

References

- Ahmad, M. I. (2004). A rage shared by law: Post-September 11 racial violence as crimes of passion. *California Law Review*, 92, 1259-1330.
- Altbach, P. G. (1998). The complex roles of universities in the period of globalization. *Higher Education in the World*, *3*(1), 5-14.
- Altbach, P. G. (2001). Higher education and the WTO: Globalization run amok. *International Higher Education*, 23, 2-4.
- Altbach, P. G. (2004). Higher education crosses borders. *Change*, 36(2), 18-24.
- Altbach, P. G. (2006). Globalization and the university: Realities in an unequal world. In J. J. F. Forest & P. G. Altbach (Eds.), *International handbook of higher education* (pp. 121-140). Dordrecht, The Netherlands: Springer.
- Altbach, P. G. (2011). Is there a future for branch campuses? *International Higher Education*, 65, 7-10.
- Altbach, P. G., & Knight, J. (2007). The internationalization of higher education: Motivations and realities. *Journal of International Education*, 11, 290-305.
- Altbach, P. G., Reisberg, L., & Rumbley, L. E. (2009). Trends in global higher education: Tracking an academic revolution. Retrieved from http://www.uis.unesco.org/Library/ Documents/trends-global-higher-education-2009-world-conference-en.pdf
- American Council on Education (ACE). (2011). *Internationalization in action*. Retrieved from https://www.acenet.edu/news-room/Pages/Internationalization-in-Action.aspx
- Anderson, N. (2015). In Qatar's education city, U.S. colleges are building an academic oasis. Retrieved from https://www.washingtonpost.com/local/education/in-qatarseducation-city-us-colleges-are-building-an-academic-oasis/2015/12/06/6b538702-8e01-11e5-ae1f-af46b7df8483_story.html
- Anderson, N. (2016). Texas university gets \$76 million each year to operate in Qatar, contract says. Retrieved from https://www.washingtonpost.com/news/gradepoint/wp/2016/ 03/08/texas-university-gets-76-million-each-year-to-operate-in-qatar-contractsays/
- Anfara, V. A., Brown, K. M., & Mangione, T. L. (2002). Qualitative analysis on stage: Making the research process more public. *Educational Researcher*, *31*(7), 28-36.

- Arasaratnam-Smith, L. A. (2017). Intercultural competence: An overview. In D. K. Deardorff & L. A. Arasaratnam-Smith (Eds.), *Intercultural competence in higher education: International approaches and application* (pp. 7-17). New York, NY: Routledge.
- Axtell, R. E. (1998). *Gestures: The do's and taboos of body language around the world*. New York, NY: Wiley.
- Ayubi, N. N. (1991). *Political Islam: Religion and politics in the Arab world*. New York, NY: Routledge.
- Bagnoli, A. (2009). Beyond the standard interview: The use of graphic elicitation and arts-based methods. *Qualitative Research*, *9*, 547-570.
- Bakken, R. (2013). *Exploring the impact on students of western universities on foreign soil: A case study of Qatar* (Doctoral dissertation). University of San Diego, San Diego, CA.
- Barakat, H. (1993). *The Arab world: Society, culture and state*. Berkeley, CA: University of California Press.
- Baxter, P., & Jack, S. (2008). Qualitative case study methodology: Study design and implementation for novice researchers. *The Qualitative Report, 13*, 544-559.
- Baxter, P., & Rideout, L. (2006). Decision making of 2nd-year baccalaureate nursing students. *Journal of Nursing Education*, 45(4), 121-128.
- Becker, R. F. J. (2009). International branch campuses: Markets and strategies—United Kingdom. Retrieved from http://www.obhe.ac.uk/documents/view_details?id=770
- Berardo, K., & Deardorff, D. K. (2012). *Building cultural competence: Innovative activities and models*. Sterling, VA: Stylus.
- Blake, H. J. (2007). The crucial role of student affairs professionals in the learning process. *New Directions for Student Services*, 117, 65-72.
- Bloor, M., & Wood, F. (2006). *Keywords in qualitative methods: A vocabulary of research concepts.* Thousand Oaks, CA: Sage.
- Boddy, C. R. (2016). Sample size for qualitative research. *Qualitative Market Research Journal*, *19*, 426-432.
- Burns, J. M. (1995). Transactional and transforming leadership. In J. T. Wren (Ed.), *The leader's companion: Insights on leadership through the ages* (pp. 100-101). New York: NY: The Free Press.
- Calderon, A. (2015). *What will higher education be like in 2040?* Retrieved from http://www.universityworldnews.com/article.php?story=20150908145912643

- Callahan, K. (2016). Higher education trends and issues around the world. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 277-289). Washington, DC: Student affairs Administrators in Higher Education (NASPA).
- Carnegie Mellon University in Qatar. (2015). *Life in Education City*. Retrieved from https://www.qatar.cmu.edu/life-in-education-city
- Carnoy, M., & Rhoten, D. (2002). What does globalization mean for educational change? A comparative approach. *Comparative Education Review*, 46(1), 1-9.
- Chalmers, I. (2011). International branch campuses and unique risk considerations. Retrieved from http://irc.nacubo.org/Documents/URMIA_ InternationalBranchCampuses_FINAL_20110718.pdf
- Choudaha, R., & de Wit, H. (2014). Challenges and opportunities for global student mobility in the future: A comparative and critical analysis. In B. Streitwieser (Ed.), *Internationalisation of higher education and global mobility* (pp. 19-34). Oxford, UK: Symposium Books.
- Ciobanu, A. (2013). The role of student services in the improving of student experience in higher education. *Procedia—Social and Behavioral Sciences*, 92(1), 169-173.
- Chemers, M. M. (1995). Contemporary Leadership Theory. In The Leader's companion: Insights on leadership through the ages (pp. 83-99). New York: NY: Free Press.
- Clark, N. (2013). *Qatar, building a national and regional knowledge economy*. Retrieved from http://wenr.wes.org/2013/04/wenr-april-2013-qatar-building-a-national-and-regional-knowledge-economy/
- Clifford, J. (1983). On ethnographic authority. Representations, 2, 118-146.
- College Stats. (2013). *Dollars and sense: A global look at student debt* [blog post]. Retrieved from https://collegestats.org/2013/04/dollars-and-sense-a-global-lookat-student-debt/#us
- College Student Educators International (ACPA). (2016). *Social justice (formerly equity, diversion, and inclusion)*. Retrieved from http://www.myacpa.org/docs/social-justice-formerly-equity-diversity-and-inclusionpdf
- Collier, M. J., Hegde, R. S., Lee, W. S., Nakayama, T. K., & Yep, G. A. (2001). Dialogue on the edges: Ferment in communication and culture. In M. J. Collier (Ed.), *International and Intercultural Communication Annual 24: Transforming communication about culture*— *Critical new directions* (pp. 219-280). Thousand Oaks, CA: Sage.

- Connelly, S., & Olsen, A. (2013). Education as an export for Australia: Green shoots, first swallows, but not quite out of the woods yet. Paper presented at the Australian International Education Conference, Canberra, Australia.
- Couto, R. A. (1995). Defining a citizen leader. In J. T. Wren (Ed.), *The leader's companion: Insights on leadership through the ages* (pp. 11-17). New York, NY: The Free Press.
- Covarrubias, P. O. (2002). *Culture, communication, and cooperation: Interpersonal relations and pronominal address in a Mexican organization*. Lanham, MD: Rowman & Littlefield.
- Crane, D., & Bovone, L. (2006). Approaches to material culture: The sociology of fashion and clothing. *Journal of Science Direct, 34*, 319-333.
- Creswell, J. W. (2002). *Educational research: Planning, conducting, and evaluating quantitative and qualitative research*. Upper Saddle River, NJ: Merrill Prentice Hall.
- Creswell, J. W. (2012). *Educational research: Planning, conducting, and evaluating quantitative and qualitative research*. Boston, MA: Pearson Education.
- Creswell, J. W. (2013). *Qualitative inquiry and research design: Choosing among five approaches.* Thousand Oaks, CA: Sage.
- Creswell, J. W. (2014). *Research design: Qualitative, quantitative, and mixed methods approaches.* Thousand Oaks, CA: Sage.
- Croom, P. W. (2012). Internationalization and institutional strategy. *AUDEM: The International Journal of Higher Education and Democracy*, 3(1), 99-119.
- Cross-Border Education Research Team (C-BERT). (2013). *Educational hubs*. Retrieved from http://www.globalhighered.org/?page_id=32
- Cross-Border Education Research Team (C-BERT). (2017). *Branch campus listing*. Retrieved from http://cbert.org/?page_id=34
- Crotty, M. (1998). *The foundation of social research: Meaning and perspective in the research process*. Thousand Oaks, CA: Sage.
- Couto, R. A. (1995). Defining a citizen leader. In J. T. Wren (Ed.) *The leader's companion: Insights on leadership through the ages (pp. 11-17).* New York, NY: The Free Press.
- de Wit, H. (2002). *Internationalization of higher education in the United States of America and Europe: A historical, comparative, and conceptual analysis.* Westport, CT: Greenwood.

- de Wit, H., Hunter, F., Howard, L., & Egron-Polak, E. (2015). *Internationalization of higher education: Study*. Brussels, Belgium: European Parliament Committee on Culture and Education.
- de Wit, H., & Meyer, G. (2012). The history of internationalization of higher education. In D. K. Deardorff, H. de Wit, J. D. Heyl, & T. Adams (Eds.), *The Sage handbook* of international higher education (pp. 43-60). Thousand Oaks, CA: Sage.
- Deardorff, D. K. (2006). Identification and assessment of intercultural competence as a student outcome of internationalization. *Journal of Studies in International Education*, *10*, 241-266.
- Deardorff, D. K. (2012). Framework: Intercultural competence model. In K. Berardo & D. K. Deardorff (Eds.), *Building cultural competence: Innovative activities and models* (pp. 45-49). Sterling, VA: Stylus.
- Deardorff, D. K. (2016a). Developing intercultural competence. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 127-145). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Deardorff, D. K. (2016b). Framework: Intercultural competence model. In K. Berardo & D. Deardorff (Eds.), *Building cultural competence: Innovative activities and models* (pp. 45-52). Sterling, VA: Stylus.
- der Heyde, A. M. (2016). Europe's impact on global higher education and student affairs services. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 43-63). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Diconsiglio, J. (2009). A new landscape: Opportunities and pitfalls for universities expanding in the Persian Gulf. Retrieved from http://eric.ed.gov/?id=EJ927358
- Dixon v. Alabama State Board of Education, 294 F.2d 150 (5th Cir. 1961).
- Donn, G., & Al Manthri, Y. (2011). Globalisation and higher education in the Arab Gulf states. University of Chicago Press Journals, 55, 488-490.
- Drew, R. (2001). *Karaoke nights: An ethnographic rhapsody*. Walnut Creek, CA: Altamira.
- Dungy, G., & Gordon, S. A. (2011). The development of student affairs. In J. H. Schuh, S. R. Jones, & S. R. Harper & Associates (Eds.), *Student services: A handbook for the profession* (pp. 61-79). San Francisco, CA: Jossey-Bass.

- Egron-Polak, E., & Hudson, R. (2014). *Internationalization of higher education: Growing expectations, fundamental values.* Paris, France: International Association of Universities.
- Evans, N. J., Forney, D. S., & Guido-DiBrito, F. (1998). *Student development in college: Theory, research and practice.* San Francisco, CA: Jossey-Bass.
- Ewers, M. C., & Lewis, J. M. (2008). Risk and the securitisation of student migration to the United States. *Journal of Economic and Social Geography*, 99, 470-482.
- Farrugia, C. A., & Lane, J. E. (2013). Legitimacy in cross-border higher education: Identifying stakeholders of international branch campuses. *Journal of Studies in International Education*, 17, 414-432.
- Flick, U. (2009). An introduction to qualitative research. Thousand Oaks, CA: Sage.
- Flyvbjerg, B. (2006). Five misunderstandings about case-study research. *Qualitative Inquiry*, *12*, 219-245.
- Flyvbjerg, B. (2011). Case study. In Norman K. Denzin & Yvonna S. Lincoln (Eds.), *The Sage handbook of qualitative research* (pp. 301-316). Thousand Oaks, CA: Sage.
- Gannon, M. J., & Pillai, R. (2016). Understanding global cultures: Metaphorical journeys through 34 nations, clusters of nations, continents, & diversity. Thousand Oaks, CA: Sage.
- Gentles, S. J., Charles, C., Ploeg, J., & McKibbon, K. (2015). Sampling in qualitative research: Insights from an overview of the methods literature. *The Qualitative Report*, 20, 1772-1789.
- George, A. L., & Bennett, A. (2005). *Case studies and theory development in the social sciences*. Cambridge, UK: MIT Press.
- Gerring, J. (2004). What is a case study and what is it good for? *American Political Science*
- Goleman, D. (1998). The emotional intelligence of leaders. *The Leader to Leader*, *98*, 20-26.
- Green, M. F., Kinser, K., & Eckel, P. D. (2008). *On the ground overseas: U.S. degree* programs and branch campuses abroad. Washington, DC: American Council on Education.
- Guba, E. G. (1990). The alternative paradigm dialog. In E.G. Guba (Ed.), *The paradigm dialog* (pp. 17-30). Newbury Park, CA: Sage.

- Gudykunst, W. B. (2002). Cross-cultural communication. In W. B. Gudykunst & B. Mody (Eds.), *Handbook of international and intercultural communication* (pp. 19-25). Thousand Oaks, CA: Sage.
- Guilherme, M. (2000) Intercultural competence. In M. Byram (Ed.), *Routledge* encyclopedia of language teaching and learning (pp. 297-300). London, UK: Routledge.
- Gulf Research Center. (2017). *Demography, migration, and the labour market in Qatar*. Retrieved from http://gulfmigration.eu/media/pubs/exno/GLMM_EN_2017_03.pdf
- Hammersley, M. (2013). What is qualitative research? New York, NY: Bloomsbury.
- Hicks, D. (2010). The material-cultural turn. In D. Hicks & M. C. Beaudry (Eds.), *The Oxford handbook of material culture studies* (pp. 25-98). Oxford, UK: Oxford University Press.
- Hofstede, G. (2001). Culture's consequences: Comparing values, behaviors, institutions, and organizations across nations. Beverly Hills, CA: Sage.
- Horn, A. S., Hendel, D. D., & Fry, G. W. (2007). Ranking the international dimension of top research universities in the United States. *Journal of Studies in International Education*, 11, 330-358.
- Hsieh, H.-F., & Shannon, S. (2005). Three approaches to qualitative content analysis. *Qualitative Health Research*, 15, 1277-1288.
- Hou, A. Y., Hill, C., Chen, K. H. J., & Tsai, S. (2018). A comparative study of international branch campuses in Malaysia, Singapore, China, and South Korea: regulation, governance, and quality assurance. *Asia Pacific Education Review*, 19(4), 543–555.
- Hughes, R. (2011). Strategies for managing and leading an academic staff in multiple countries. *New Directions for Higher Education*, 155, 19-28. doi:10.1002/he.441
- Hughes, R. L., Ginnett, R. C., Curphy, G. J. (1995). Power, influence, and influence tactics. In J. T. Wren (Ed.), *The leader's companion: Insights on leadership through the ages* (pp. 339-351). New York: NY: Free Press.
- Institute of International Education (IIE). (2014). *Open doors: Fast facts*. Retrieved from Fast-Facts-2014.pdf
- Jauregui, M. (2013). Cross-cultural training of expatriate faculty teaching in international branch campuses (Doctoral dissertation). Los Angeles, CA: University of Southern California.

- Johnson, K. A. C. (2018). 9/11 and international student visa issuance. *Journal of Studies in International Education*, 22, 1-21.
- Kelly, R. E. (1995). In praise of followers. In J. T. Wren (Ed.), *The leader's companion: Insights on leadership through the ages* (pp. 193-204). New York: NY: Free Press.
- Knight, J. (2003). Updating the definition of internationalization. *International Higher Education*, *33*(1), 2-3.
- Knight, J. (2004). Internationalization remodeled: Definitions, rationales, and approaches. *Journal for Studies in International Education*, 8(1), 5-31.
- Knight, J. (2006). Higher education crossing borders: A guide to the implications of the General Agreement on Trade and Services (GATS) for cross-border education. Retrieved from http://unesdoc.unesco.org/images/0014/001473/147363e.pdf
- Knight, J. (2008). Higher education in turmoil: The changing world of internationalization. Retrieved from https://www.sensepublishers.com/media/ 475-higher-education-in-turmoil.pdf
- Knight, J. (2011). Education hubs: A fad, a brand, an innovation? *Journal of Studies in International Education*, *15*, 221-240. doi:10.1177/1028315311398046
- Knight, J. (2012a). Concepts, rationales, and interpretative frameworks in the internationalization of higher education. In D. Deardorff, H. De Wit, J. D. Heyl, & T. Adams (Eds.), *The SAGE handbook of international higher education* (pp. 27-42). Thousand Oaks, CA: Sage.
- Knight, J. (2012b). *Internationalization: Three generations of crossborder higher education*. New Delhi, India: India International Centre.
- Knight, J. (2013). Education hubs: International, regional and local dimensions of scale and scope. *Comparative Education*, *49*, 374-387.
- Knight, J. (2014). *Internationalization of education*. Retrieved from http://www.aqu.cat/ elButlleti/butlleti75/articles1_en.html#.WBYla03VzX6
- Komives, S. R., Woodard, D. B., & Associates. (2003). *Student services: A handbook for the profession*. San Francisco, CA: Jossey-Bass.
- Krieger, Z. (2008). Academic building boom transforms the Persian Gulf. *Education Digest*, 74(1), 4-10.

- Kruger, K., der Heyde, A. M., El Housseini, W. Y., Zereik, R., Seeto, E., & Schreiber, B. (2016). Global student affairs and services associations and internationalization. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 257-275). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Kuk, L., Banning, J. H., & Amey, M. J. (2010). Positioning student affairs for sustainable change: Achieving organizational effectiveness through multiple perspectives. Sterling, VA: Stylus.
- Kuper, A., Lingard, L., & Levinson, W. (2008). Critically appraising qualitative research. British Medical Journal, 337, 1756-1833.
- Lane, J. E. (2011). Global expansion of international branch campuses: Managerial and leadership challenges. *New Directions for Higher Education*, 155, 5-17.
- Lane, J. E., & Kinser. K. (2011). The cross-border education policy context: Educational hubs, trade liberalization, and national sovereignty. *New Directions for Higher Education*, 155, 79-85.
- Lawton, W., & Katsomitros, A. (2012). *International branch campuses: Data and developments*. London, UK: The Observatory on Borderless Higher Education.
- Lewin, T. (2009). *George Mason University, among first with an Emirates branch, is pulling out.* Retrieved from http://www.nytimes.com/2009/03/01/education/01campus.html?r=4
- Lincoln, Y. S., & Guba, E. G. (1985). Naturalistic inquiry. Newbury Park, CA: Sage.
- Long, D. (2012). The foundations of student affairs: A guide to the profession. In L. J. Hinchliffe & M. A. Wong (Eds.), *Environments for student growth and development: Librarians and student affairs in collaboration* (pp. 1-39). Chicago, IL: Association of College & Research Libraries.
- Long, T., & Johnson, M. (2000). Rigour, reliability and validity in qualitative research. *Clinical Effectiveness in Nursing*, 4(1), 30-37.
- Lustig, M. W., & Koester, J. (2012). *Intercultural competence: Interpersonal communication across cultures*. London, UK: Pearson.
- Magolda, M. B. (2003). Identity and learning: Student affairs' role in transforming higher education. *Journal of College Student Development*, 44, 231-247.
- Manning, K., Kinzie, J., & Schuh, J. H. (2014). One size does not fit all: Traditional and innovative models of student affairs practice. London, UK: Routledge.

- Marijk, W. C. (2003). Globalisation and access to higher education. *Journal of Studies in International Education*, 7(2), 193-206.
- Martin, J. N., & Nakayama, T. K. (2010). *Intercultural communication in contexts*. New York, NY: McGraw-Hill.
- Maxwell, J. A. (2013). *Qualitative research design: An interactive approach*. Thousand Oaks, CA: Sage.
- McBurnie, G. (2000). Pursuing internationalization as a means to advance the academic mission of the university: An Australian case study. *Higher Education in Europe*, 25(1), 63-73.
- McClellon, G. S., Stringer, J., & Associates. (2009). *The handbook of student affairs administration* (3rd ed.). San Francisco, CA: Jossey-Bass.
- Megerian, L. E., & Sosik, J. J. (1996). An affair of the heart: Emotional intelligence and transformational leadership. Retrieved from http://Wk3_Megerian.pdf
- Memmi, A. (1965). The colonizer and the colonized. Boston, MA: Beacon.
- Merriam, S. B. (1998). *Qualitative research and case study applications in education*. San Francisco, CA: Jossey-Bass.
- Megerian, L. E., & Sosik, J. J. (1996). An affair of the heart: Emotional intelligence and transformational leadership. Retrieved from http://Wk3_Megerian.pdf
- Meyer, C. F., & Rhoades, E. K. (2006). *Multiculturalism: Beyond food, festival, folklore, and fashion*. Retrieved from http://www.redorbit.com/news/education/348629/ multiculturalism_beyond_food_festival_folklore_and_fashion/
- Miles, M. B., Huberman, A. M., & Saldaña, J. (2014). *Qualitative data analysis: A methods sourcebook*. Thousand Oaks, CA: Sage.
- Mills, G. (2003). *Action research: A guide for the teacher researcher* (2nd ed.). Columbus, OH: Merrill-Prentice Hall.
- Minkov, M. (2011). *Cultural differences in a globalizing world*. Bingley, UK: Emerald Group.
- Morse, J. M., Barrett, M., Mayan, M., Olson, K., & Spiers, J. (2002). Verification strategies for establishing reliability and validity in qualitative research. *International Journal of Qualitative Methods*, *1*(2), 13-22.

- Moscaritolo, L. B., Osfield, K. J., Perozzi, B., & Shea, R. (2016). The future of international student affairs and services. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 291-306). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Moscaritolo, L. B., & Roberts, D. (2016). Global competencies for student affairs and services professionals. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 109-126).
 Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Naidoo, V. (2008). Transnational higher education: A stock take of current activity. Journal of Studies in International Education, 13, 310-330. doi:10.1177/1028315308317938
- National Institute for Learning Outcomes Assessment (NILOA). (2010). *Role of student affairs*. Retrieved from http://www.learningoutcomesassessment.org/occasionalpaperseven.htm
- Nejem, R. F. (2016). *When in the Arab world: An insider's guide to living and working with Arab culture.* Amman, Jordan: Yarnu.
- Neuliep, J. W. (2015). *Intercultural communication: A contextual approach*. Thousand Oaks, CA: Sage.
- Nisbett, R. E. (2003). *The geography of thought: How Asians and Westerners think differently and why.* New York, NY: Free Press.
- Northouse, P. G. (2012). *Introduction to leadership concepts and practice*. Thousand Oaks, CA: Sage.
- Olson, C. L., Green, M. F., & Hill, B. A. (2006). A handbook for advancing comprehensive internationalization: What institutions can do and what students should learn. Washington, DC: American Council on Education.
- Osfield, K. J., & Associates. (2008). *Internationalization of student affairs and services: An emerging global perspective*. Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Otten, M. (2003). Intercultural learning and diversity in higher education. *Journal of Studies in International Education*, 7(1), 12-26.
- Patton, M. Q. (2002). *Qualitative research and evaluation methods*. Thousand Oaks, CA: Sage.

- Perozzi, B., Giovannucci, G. L., & Shea, R. (2016). The global dialogue. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 21-41). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Perozzi, B., & O'Brien, A. (2010). The global practice of student affairs and services: A United States and Australian case study. *Journal of the Australian and New Zealand Student Services Association*, 36(1), 1-25.
- Perozzi, B., Osfield, K. J., Moscaritolo, L. B., & Shea, R. (2016). Collaborative internationalization in global student affairs and services. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 3-19). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Peshkin, A. (1988). In search of subjectivity—One's own. *Educational Researcher*, *17*(7), 17-21.
- Qatar Foundation (QF). (2016). *Mission*. Retrieved from http://www.qf.org.qa/about/about
- Qatar Foundation (QF). (2017). *QF entities*. Retrieved from http://www.qf.org.qa/qfentities/qf-entities
- Qatar Ministry of Development Planning and Statistics (QMDPS). (2008). *Qatar National Vision 2030*. Retrieved from https://www.mdps.gov.qa/en/qnv/ Pages/QNVDocument.aspx
- Qatar Ministry of Development Planning and Statistics (QMDPS). (2018). *Qatar total population*. Retrieved from https://www.mdps.gov.qa/en/Pages/default.aspx
- Redden, E. (2015a). *The branch campus boom(s)*. Retrieved from https:// www.insidehighered.com/news/2015/03/16/international-branch-campusphenomenon-just-fad
- Redden, E. (2015b). *Persona non grata*. Retrieved from https://www.insidehighered .com/news/2015/03/18/nyu-professor-denied-entry-uae-where-university-hascampus
- Rolfe, G. (2006). Validity, trustworthiness and rigour: Quality and the idea of qualitative research. *Journal of Advanced Nursing*, *53*, 304-310.
- Romani, V. (2009). The politics of higher education in the Middle East: Problems and prospects. *Brandeis University Middle East Brief, 36*, 1-8.
- Rossman, G. B., & Rallis, S. F. (2012). *Learning in the field: An introduction to qualitative research* (3rd ed.). Thousand Oaks, CA: Sage.

- Rost, J. C. (1995). Leaders and followers are the people in this relationship. In J. T.
 Wren (Ed.), *The leader's companion: Insights on leadership through the ages* (pp. 189-192). New York: NY: Free Press.
- Rostron, M. (2009). Liberal arts education in Qatar: Intercultural perspectives. *Intercultural Education*, 20, 219-229.
- Rubin, H. J., & Rubin, I. S. (2012). *Qualitative interviewing: The art of hearing data* (3rd ed.). Thousand Oaks, CA: Sage.
- Ruby, A. (2010). *Thinking about a branch campus? Think twice*. Retrieved from https://www.chronicle.com/article/Thinking-About-a-Branch/64741/
- Saldaña, J. (2009). *The coding manual for qualitative researchers*. Thousand Oaks, CA: Sage.
- Sandelowski, M. (1993). Rigor or rigor mortis: The problem of rigor in qualitative research revisited. *Advances in Nursing Science*, *16*(2), 1-8.
- Sawahel, W. (2017). *Ambitious but secretive Arab education provider: Iran*. Retrieved from https://www.al-fanarmedia.org/2017/03/ambitious-secretive-arab-education-provider-iran/
- Schuh, J. H., Jones S. R., & Harper, S. R. & Associates (Eds.). (2011). *Student services: A handbook for the profession* (5th ed.). San Francisco, CA: Jossey-Bass.
- Seifert, T., Perozzi, B., Bodine Al-Sharif, M., Li, W., & Wildman, K. (2014). Student affairs and services in global perspective: A preliminary exploration of practitioners' background, roles and professional development. Retrieved from https://www.academia.edu/35561798/Student_Affairs_and_Services_in_Global_ Perspective_A_Preliminary_Exploration_of_Practitioners_Background_Roles_ and_Professional_Development
- Seifert, T., Perozzi, B., & Li, W. (2016). Two sides of the same coin: Educational pathways and professional learning. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 65-86). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Shams, F., & Huisman, J. (2012). Managing offshore campuses: An analytical framework for institutional strategies. *Journal of Studies in International Education*, 16(2), 106-127.
- Shea, R., Gormley, B., Clarke, A., & Leary, T. (2016). Professional development for student affairs and services staff. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 87-105). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).

- Shields, C. (2010). Transformative leadership: Working for equity in diverse contexts. *Educational Administration Quarterly*, 46, 558-589.
- Simons, H. (2009). Case study research in practice. London, UK: Sage.
- Sirat, M. (2008). The impact of September 11 on international student flow into Malaysia: Lessons learned. *International Journal of Asia Pacific Studies*, 4(1), 79-95.
- Slevin, E., & Sines, D. (2002). Enhancing the truthfulness, consistency, and transferability of a qualitative study: Using a manifold of two approaches. *Nurse Researcher*, 7(2), 79-98.
- Smithee, M. B. (2012). Finding leadership for internationalization of U.S. higher education. *Journal of International Education and Leadership*, 2(1), 1-29.
- Soilemetzidis, I. (2011). International Association of Universities (IAU) 3rd global survey report: Anglohigher[®]. Retrieved from http://www.anglohigher.com/casestudies/casestudy_detail/45/39
- Stake, R. E. (1995). The art of case study research. Thousand Oaks, CA: Sage.
- Stake, R. E. (2005). Qualitative case studies. In N. K. Denzin & Y. S. Lincoln (Eds.), *The handbook of qualitative research* (3rd ed., pp. 443-466). Thousand Oaks, CA: Sage.
- Stake, R. E. (2008). Qualitative case studies. In N. K. Denzin & Y. S. Lincoln (Eds.), *Strategies of qualitative inquiry* (pp. 119-150). Thousand Oaks, CA: Sage.
- Staley, O. (2013). *Duke to NYU missteps abroad lead colleges to reassess expansion*. Retrieved from http://www.bloomberg.com/news/articles/2013-10-04/duke-to-nyu-missteps-abroad-lead-colleges-to-reassess-expansion
- Stensberg, K., Sliva, C., & Medina, D. (2016). International employment in student affairs and services. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 193-210).
 Washington, DC: Student Affairs Administrators in Higher Education (NASPA).
- Storti, C. (1999). *Figuring foreigners out: A practical guide*. Boston, MA: Intercultural Press.
- Stringer, E. T. (2014). Action research (4th ed.). Thousand Oaks, CA: Sage.
- Stryker, J. C., Witt, E., & Konecny, K. S. (2016). Branches, hubs and hybrids: Trends in international education models. In K. J. Osfield, B. Perozzi, L. B. Moscaritolo, & R. Shea (Eds.), *Supporting students globally in higher education* (pp. 213-235). Washington, DC: Student Affairs Administrators in Higher Education (NASPA).

- Student Affairs Administrators in Higher Education (NASPA). (2016a). *Equity and diversity*. Retrieved from https://www.naspa.org/focus-areas/equity-and-diversity
- Student Affairs Administrators in Higher Education (NASPA). (2016b). NASPA careers. Retrieved from https://www.naspa.org/careers
- Student Affairs Administrators in Higher Education (NASPA). (2016c). NASPA International Student Services Institute. Retrieved from https://www.naspa.org/focus-areas/globalism/NISSI
- Student Affairs Administrators in Higher Education (NASPA). (2018). *What is student affairs?* Retrieved from https://web.archive.org/web/20120321190223/http://www.naspa.org/career/default.cfm
- Taylor, S. J., Bogdan, R., & DeVault, M. L. (2016). *Introduction to qualitative research methods: A guidebook and resource*. Hoboken, NJ: Wiley.
- Thelin, J. R., & Gasman, M. (2011). Historical overview of American higher education. In J. H. Schuh, S. R. Harper, & Associates (Eds.), *Student services: A handbook for the profession* (pp. 3-23). San Francisco, CA: Jossey-Bass.
- Thomas, G. (2011). A typology for the case study in social science following a review of definition, discourse and structure. *Qualitative Inquiry*, *17*, 511-521.
- Ting-Toomey, S. (1999). Communicating across cultures. New York, NY: Guilford.
- Toma, J. D. (2006). Approaching rigor in applied qualitative research. In C. Conrad & R. Serlin (Eds.), *Sage handbook for research in education* (pp. 405-423). Thousand Oaks, CA: Sage.
- Tracy, S. J. (2013). *Qualitative research methods: Collecting evidence, crafting analysis, communicating impact.* Malden, MA: Wiley.
- UNESCO Institute of Statistics. (2013). *Global flow of tertiary-level students*. http://www .uis.unesco.org/Education/Pages/international-student-flow-viz.aspx
- United Nations Development Programme (UNDP). (2003). Arab human development report 2003: Building a knowledge society. Retrieved from http://hdr.undp.org/ sites/default/files/rbas_ahdr2003_en.pdf
- United Nations Development Programme (UNDP). (2011). Arab development challenges report 2011: Towards the developmental state in the Arab region. Retrieved from http://www.undp.org/content/undp/en/home/librarypage/hdr/arab-development-challenges-report-2011.html

- United Nations Educational, Scientific and Cultural Organization (UNESCO). (2002). *The role of student affairs and services in higher education: A practical manual for developing, implementing and assessing student affairs programmes and services.* Retrieved from http://unesdoc.unesco.org/images/0012/001281/128118e.pdf
- United Nations Educational, Scientific and Cultural Organization (UNESCO). (2009). *Trends in global higher education: Tracking an academic revolution*. Retrieved from http://www .uis.unesco.org/Library/Documents/trends-global-higher-education-2009-worldconference-en.pdf
- University of Texas-Austin. (2018). *History & traditions*. Retrieved from https://www.utexas.edu/about/history-and-traditions
- van der Wende, M. C. (2003). Globalization and access to higher education. *Journal for Studies in International Education*, 7(2), 193-206.
- Verbik, L., & Merkley, C. (2006). *The international branch campus: Models and trends*. London, UK: The Observatory on Borderless Higher Education.
- Verschuren, P. J. M. (2003). Case study as a research strategy: Some ambiguities and opportunities. *International Journal of Social Research Methodology*, *6*(2), 121-139.
- Vora, N. (2014). Between global citizenship and Qatarization: Negotiating Qatar's new knowledge economy within American branch campuses. *Ethnic and Racial Studies*, 37, 2243-2260.
- Wax, R. H. (1971). Doing fieldwork: Warnings and advice. Chicago, IL: University of Chicago Press.
- Wilkins, S., & Huisman, J. (2012). The international branch campus as transnational strategy in higher education. *International Center for Higher Education Management*, 64, 627-645.
- World Bank. (2000). Higher education in developing countries: Peril and promise. Retrieved from http://siteresources.worldbank.org/EDUCATION/Resources/278200-1099079877269/547664-1099079956815/peril_promise_en.pdf
- Yin, R. K. (2014). *Case study research: Design and methods* (5th ed.). Thousand Oaks, CA: Sage.
- Yin, R. K. (2016). *Qualitative research from start to finish* (2nd ed.). New York, NY: Guilford.

- Zaharna, R. S. (1995). Understanding cultural preferences of Arab communication patterns. *Public Relations Review*, 21(3), 241-255.
- Zeman, J., & Adrian, M. (2005). The effect of post 9-11 policy on foreign students: A critical analysis. Retrieved from https://www.researchgate.net/publication/ 237434217_The_Effect_of_Post_9-11_Policy_on_Foreign_Students_ A_Critical_Analysis
- Ziguras, C., & McBurnie, G. (2011). Transnational higher education in the Asia-Pacific region: From distance education to the branch campus. *Higher Education Dynamics*, *36*(2), 105-122.
- Zweifel, T. D. (2003). *Culture clash: Managing the global high-performance team*. New York, NY: SelectBooks.
- Zweifel, T. D. (2013). *Culture clash 2: Managing the global high-performance team.* New York, NY: SelectBooks.

Appendix

Interview Questions

- 1. Tell me about your work with the students on your campus.
- 2. In what ways is the student population on your campus different from the student population in the United States?
- Describe a student event that you recently planned or attended on your campus or in Education City.
- 4. In what ways, if any, does the local cultural context shape the planning and implementation of your student events, activities, and programs?
- 5. In what ways, if any, does the Islamic religion influence the planning and implementation processes of the student activities on your campus or in Education City?
- 6. What are the main factors that motivate your students to engage in student activities?
- 7. Tell me about the learning objectives, if any, for student activities on your campus.
- 8. Have you made any special arrangements to accommodate a specific student's participation in an event? If so, how?
- 9. Reflecting on your experiences, what changes have you made in practices or what changes would you like to make in the future for planning your events/programs/services?
- 10. Is there anything that I should have asked you but did not?